AIRGAS INC Form 10-Q February 08, 2010 Table of Contents

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

b QUARTERLY REPORT UNDER SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended: December 31, 2009

Commission file number: 1-9344

AIRGAS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 56-0732648 (I.R.S. Employer Identification No.)

259 North Radnor-Chester Road, Suite 100 Radnor, PA (Address of principal executive offices)

19087-5283 (ZIP code)

(610) 687-5253

(Registrant s telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes b No "

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes "No"

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of large accelerated filer, accelerated filer and smaller reporting company in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer b Accelerated filer

Non-accelerated filer " (Do not check if a smaller reporting company)

Smaller reporting company

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes " No þ

Shares of common stock outstanding at February 3, 2010: 82,729,623 shares

AIRGAS, INC.

FORM 10-Q

December 31, 2009

INDEX

PART I FINANCIAL INFORMATION	
Item 1. Financial Statements	
Consolidated Statements of Earnings for the Three and Nine Months Ended December 31, 2009 and 2008 (Unaudited)	,
Consolidated Balance Sheets as of December 31, 2009 (Unaudited) and March 31, 2009	4
Consolidated Statements of Cash Flows for the Nine Months Ended December 31, 2009 and 2008 (Unaudited)	:
Notes to Consolidated Financial Statements (Unaudited)	
Item 2. Management s Discussion and Analysis of Financial Condition and Results of Operations	2:
Item 3. Quantitative and Qualitative Disclosures About Market Risk	4
Item 4. Controls and Procedures	4:
<u>PART II OTHER INFORMATIO</u> N	
Item 1. Legal Proceedings	4:
Item 1A. Risk Factors	4:
Item 6. Exhibits	4:
SIGNATURE	4

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

AIRGAS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF EARNINGS

(Unaudited)

(In thousands, except per share amounts)

	Three Months Ended December 31, 2009 2008		Nine Months I December 2009				
Net Sales	\$ 942,10	7 \$	1,078,733	\$ 2	2,883,630	\$:	3,357,355
Costs and Expenses:							
Cost of products sold (excluding depreciation)	416,36	4	500,806		1,282,633		1,597,291
Selling, distribution and administrative expenses	366,30	1	392,665		1,109,306		1,186,448
Depreciation	53,64	2	49,739		157,872		146,767
Amortization	5,81	1	5,001		16,104		16,487
Total costs and expenses	842,11	8	948,211	2	2,565,915	2	2,946,993
Operating Income	99,98	9	130,522		317,715		410,362
Interest expense, net	(15,03	4)	(23,267)		(49,744)		(64,393)
Discount on securitization of trade receivables	(1,403) (3,191)		(4,503)			(9,041)	
Loss on debt extinguishment	(6,66	7)			(8,678)		
Other income (expense), net	(5	8)	(604)		890		(471)
Earnings before income taxes	76,82	7	103,460		255,680		336,457
Income taxes	(29,96	1)	(40,557)		(99,458)		(131,850)
Net Earnings	\$ 46,86	6 \$	62,903	\$	156,222	\$	204,607
	,		ĺ		,		,
Net Earnings Per Common Share:							
Basic earnings per share	\$ 0.5	7 \$	0.77	\$	1.91	\$	2.49
Diluted earnings per share	\$ 0.5	6 \$	0.76	\$	1.87	\$	2.43
British currings per share	Ψ 0.5	ο φ	0.70	Ψ	1.07	Ψ	2.13
Weighted Average Shares Outstanding:							
Basic	82,13	4	81,207		81,895		82,121
	,		~-,,		,		,
Diluted	83,91	0	82,706		83,576		84,161
Dilucu	05,91	0	02,700		05,570		07,101

See accompanying notes to consolidated financial statements.

Table of Contents 4

3

AIRGAS, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

(In thousands, except per share amounts)

	Unaudited) ecember 31, 2009	March 31, 2009
ASSETS		
Current Assets		
Cash	\$ 33,559	\$ 47,188
Trade receivables, less allowances for doubtful accounts of \$28,447 and \$27,572 at		
December 31, 2009 and March 31, 2009, respectively	180,890	184,739
Inventories, net	347,167	390,445
Deferred income tax asset, net	38,467	34,760
Prepaid expenses and other current assets	64,475	60,838
Total current assets	664,558	717,970
	2 = 22 < 24	2 7 7 2 7 2 2
Plant and equipment at cost	3,732,691	3,558,730
Less accumulated depreciation	(1,313,209)	(1,192,204)
Plant and equipment, net	2,419,482	2,366,526
Goodwill	1,106,089	1,063,370
Other intangible assets, net	216,251	216,070
Other non-current assets	33,814	35,601
Total assets	\$ 4,440,194	\$ 4,399,537
LIABILITIES AND STOCKHOLDERS EQUITY		
Current Liabilities		
Accounts payable, trade	\$ 120,152	\$ 156,838
Accrued expenses and other current liabilities	278,416	264,564
Current portion of long-term debt	10,103	11,058
Total current liabilities	408,671	432,460
Long-term debt, excluding current portion	1,604,070	1,750,308
Deferred income tax liability, net	614,141	565,783
Other non-current liabilities	74,679	79,231
Commitments and contingencies		
Stockholders Equity		
Preferred stock, 20,030 shares authorized, no shares issued or outstanding at December 31, 2009 and		
March 31, 2009		
Common stock, par value \$0.01 per share, 200,000 shares authorized,		
86,115 and 85,542 shares issued at December 31, 2009 and March 31, 2009, respectively	862	856
Capital in excess of par value	566,652	533,030
Retained earnings	1,310,980	1,198,985
Accumulated other comprehensive income (loss)	2,732	(10,753)
Treasury stock, 3,926 and 4,139 common shares at cost at December 31, 2009 and March 31, 2009, respectively	(142,593)	(150,363)

Total stockholders equity	1,738,633	1,571,755
Total liabilities and stockholders equity	\$ 4,440,194	\$ 4,399,537

See accompanying notes to consolidated financial statements.

4

AIRGAS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

(Unaudited)

(In thousands)	Nine Months Ended December 31, 2009		Nine Months Ended December 31, 2008	
CASH FLOWS FROM OPERATING ACTIVITIES		2009		2000
Net earnings	\$	156,222	\$	204,607
Adjustments to reconcile net earnings to net cash provided by operating activities:	*	100,222	, T	20.,007
Depreciation		157,872		146,767
Amortization		16,104		16,487
Deferred income taxes		39,000		67,804
(Gain) loss on sales of plant and equipment		2,604		(418)
Stock-based compensation expense		19,139		16,347
Loss on debt extinguishment		8,678		10,547
Changes in assets and liabilities, excluding effects of business acquisitions:		0,076		
Securitization of trade receivables		(43,500)		
Trade receivables, net		50,911		50,468
Inventories, net		45,444		(37,773)
Prepaid expenses and other current assets		(3,916)		(12,489)
Accounts payable, trade		(35,891)		(43,546)
Accrued expenses and other current liabilities		1,605		2,201
Other non-current assets		2,320		882
Other non-current liabilities		(4,857)		2,101
Net cash provided by operating activities		411,735		413,438
CASH FLOWS FROM INVESTING ACTIVITIES Capital expenditures		(191,674)		(282,286)
Proceeds from sales of plant and equipment		8,889		11,598
Business acquisitions and holdback settlements		(75,067)		(253,184)
Other, net		(2,622)		(577)
Net cash used in investing activities		(260,474)		(524,449)
CASH FLOWS FROM FINANCING ACTIVITIES				
Proceeds from borrowings		971,914		1,235,326
Repayment of debt		(1,120,976)		(999,559)
Financing costs		(3,168)		(5,746)
Premium paid on redemption of senior subordinated notes		(6,438)		
Purchase of treasury stock				(120,219)
Proceeds from the exercise of stock options		7,175		14,182
Stock issued for the employee stock purchase plan		11,336		12,235
Tax benefit realized from the exercise of stock options		3,824		10,530
Dividends paid to stockholders		(44,227)		(32,892)
Change in cash overdraft and other		15,670		(3,723)
Net cash (used in) provided by financing activities		(164,890)		110,134
Change in cash	\$	(13,629)	\$	(877)
Cash Beginning of period	Φ	47,188	Φ	. ,
Cash Degining of period		47,100		43,048

Cash End of period \$ 33,559 \$ 42,171

See accompanying notes to consolidated financial statements.

5

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(UNAUDITED)

(1) BASIS OF PRESENTATION

The consolidated financial statements include the accounts of Airgas, Inc. and its subsidiaries (Airgas or the Company). Intercompany accounts and transactions are eliminated in consolidation. The accompanying consolidated financial statements have been prepared in accordance with U.S. generally accepted accounting principles (GAAP). These consolidated financial statements do not include all disclosures required for annual financial statements. These consolidated financial statements should be read in conjunction with the more complete disclosures contained in the Company s audited consolidated financial statements for the fiscal year ended March 31, 2009.

The preparation of financial statements in accordance with GAAP requires the use of estimates. The consolidated financial statements reflect, in the opinion of management, reasonable estimates and all adjustments necessary to present fairly the Company s results of operations, financial position and cash flows for the periods presented. The interim operating results are not necessarily indicative of the results to be expected for the entire year.

Business Segment Change

During the fourth quarter of fiscal 2009, the Company changed the operating practices and organization of its air separation production facilities and national specialty gas labs. The new operating practices and organization reflect the evolution of these businesses and their role in supporting the regional distribution companies. The regional distribution companies market to and manage the end customer relationships, coordinating and cross-selling the Company s multiple product and service offerings in a closely coordinated and integrated manner. As a result of these changes, the air separation production facilities and national specialty gas labs are now reflected in the Distribution business segment. Also as a result of an organizational realignment, National Welders is now part of the Distribution business segment. Segment information from fiscal 2009 as disclosed in Note 15 has been recast to reflect these changes. The change in business segments had no effect on the Company s financial position, results of operation or liquidity.

(2) NEW ACCOUNTING PRONOUNCEMENTS

(a) Recently adopted accounting pronouncements

In June 2009, the Financial Accounting Standards Board (FASB) designated the FASB Accounting Standards Codification (the Codification) as the source of authoritative accounting principles to be applied by nongovernmental entities in the preparation of financial statements in conformity with GAAP. Concurrent with this designation, the FASB also issued guidance on the framework for selecting the accounting principles used in the preparation of financial statements. The Codification is a structure which takes accounting pronouncements and organizes them into approximately 90 accounting topics. In addition to the Codification, the rules and interpretive releases of the Securities and Exchange Commission (SEC) under federal securities laws are considered sources of authoritative GAAP for SEC registrants, as the Codification does not replace or affect guidance issued by the SEC or its staff for public entities in their filings with the SEC. Upon the adoption of the Codification, all previously existing non-SEC accounting and reporting standards are superseded, with the exception of certain pronouncements grandfathered by the FASB. Going forward, the FASB will issue Accounting Standards Updates (ASUs) in order to update the Codification, provide background information about new guidance and describe the bases for conclusions on changes to the Codification. The Codification and the guidance surrounding the framework for selecting accounting principles is effective for financial statements issued for interim and annual periods ending after September 15, 2009. The Company adopted the Codification and associated guidance for the interim period ended September 30, 2009. As a result, previous references to former GAAP in the consolidated financial statements have been eliminated (with the exception of certain grandfathered pronouncements), and the financial statement disclosures starting with the interim period ended September 30, 2009 explain the accounting concepts utilized rather than specific GAAP references from the Codification. The adoption of the Codification and associated guidance did not change the application of GAAP to the Company s consolidated financial statements.

In September 2006, the FASB issued accounting guidance on fair value measurements and disclosures, effective for financial statements issued for fiscal years beginning after November 15, 2007. The guidance did not require any new fair value measurements; rather it replaced multiple existing definitions of fair value with a single definition, established a consistent framework for measuring fair value and expanded financial statement disclosures regarding fair value measurements. In February 2008, the FASB delayed the effective date of the new fair value measurement and disclosure guidance until fiscal years beginning after November 15, 2008 for non-financial assets and liabilities that are not

recognized or disclosed at fair value in the financial statements on a recurring basis. In April 2009, the FASB also issued clarifying guidance on determining fair value when the volume and level of activity for the asset or liability have significantly decreased and on identifying transactions that are not orderly, with the guidance effective for interim and annual reporting

6

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

periods ending after June 15, 2009. The Company applied the fair value measurement and disclosure guidance to its non-financial assets and liabilities as of April 1, 2009, as required. Additionally, the Company adopted the guidance regarding the (1) determination of fair value when the level of activity and/or volume has declined, and (2) identification of transactions that are not orderly, for the interim reporting period ended June 30, 2009, as required. The adoption of the new guidance over fair value measurements and disclosures did not have a material impact on the Company s financial position, results of operations or liquidity.

In December 2007, the FASB issued revised guidance on accounting for business combinations, which replaces FASB Statement of Financial Accounting Standards (SFAS) No. 141, Business Combinations (SFAS 141). The revised guidance is effective prospectively for business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008, and prohibits early adoption. The revised guidance significantly changes the way the Company accounts for business combinations. The Company has historically pursued new business opportunities through acquisitions and intends to maintain this strategy for the foreseeable future. Accordingly, the Company expects the revised guidance on accounting for business combinations to impact its operating results as significant acquisitions are completed and during the subsequent acquisition measurement periods when the fair values of the individual assets and liabilities acquired are determined. The principles contained in the revised guidance are, in a number of ways, very different from those previously applied to business combinations under SFAS 141. The impact of the revised guidance on the consolidated financial statements may be driven by, among other things, recognizing the direct costs of acquisitions as period costs when incurred and recasting previously issued consolidated financial statements as the provisional values assigned to the assets and liabilities acquired are trued-up to their acquisition date fair values. The Company adopted the revised guidance on April 1, 2009 and accounted for fiscal 2010 acquisitions based on the new guidance (see Note 3). Transaction and other integration costs related to fiscal 2010 acquisitions, which under the new guidance are required to be recognized in current period income, were not material for the three- and nine-month periods ended December 31, 2009. Additionally, no recasting of previously issued consolidated financial statements was required. However, should the Company enter into a material business combination, or a series of individually immaterial business combinations that are material collectively, the provisions of the new guidance may have a material impact on the Company s consolidated financial statements.

In December 2007, the FASB issued amended guidance on accounting for noncontrolling interests (formerly minority interests). The amended guidance establishes accounting and reporting standards that require (1) noncontrolling interests held by non-parent parties be clearly identified and presented in the consolidated balance sheet within equity, separate from the parent sequity, and (2) the amount of consolidated net income attributable to the parent and to the noncontrolling interests be clearly presented on the face of the consolidated statement of earnings. The amended guidance also requires consistent reporting of any changes to the parent sownership interest while retaining a controlling financial interest, as well as specific guidelines over how to treat the deconsolidation of controlling interests and any applicable gains or losses. The amended guidance is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008 and is to be applied prospectively, except for the presentation and disclosure requirements, which are to be applied retrospectively for all periods presented. The Company adopted the amended guidance for noncontrolling interests on April 1, 2009 with no impact on the consolidated financial statements, as all of the Company subsidiaries are currently 100% owned subsidiaries. Additionally, the retrospective application of the presentation and disclosure requirements was not material. Therefore, the Company did not amend its fiscal 2008 and prior presentation or disclosures regarding the former minority interest in its National Welders joint venture.

In November 2008, the FASB issued new guidance on accounting for defensive intangible assets. The new guidance clarifies how to account for acquired defensive intangible assets subsequent to initial measurement in a business combination. The new guidance is effective for fiscal years beginning after December 15, 2008. The Company adopted the new guidance on April 1, 2009, with no material impact on the Company s financial position, results of operations or liquidity.

In April 2008, the FASB issued amended guidance on the determination of the useful life of intangible assets, which modifies the factors that should be considered in developing renewal or extension assumptions used to determine the useful life of a recognized intangible asset. The amended guidance is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008 with early adoption prohibited. The amended guidance shall be applied prospectively to intangible assets acquired after the effective date; however, the disclosure requirements of the amended guidance shall be applied prospectively to all intangible assets recognized as of, and subsequent to, the effective date. The Company adopted the amended guidance on April 1, 2009. The adoption of the amended guidance did not have an impact on the Company s financial position, results of operations or liquidity, as the useful lives of the Company s intangible assets were not impacted by

renewal or extension provisions.

7

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

In April 2009, the FASB issued amended guidance regarding the initial recognition and measurement, subsequent measurement and accounting, and disclosure of assets and liabilities arising from contingencies in a business combination. The amended guidance carries forward the general requirements of SFAS 141 for acquired contingencies in a business combination, yet encourages greater use of fair value when determinable. The amended guidance is effective for fiscal years beginning on or after December 15, 2008 with early adoption prohibited. The Company adopted the amended guidance on April 1, 2009. The adoption of the amended guidance did not have a material impact on the Company s financial position, results of operations or liquidity.

In April 2009, the FASB issued new guidance for interim disclosures about the fair value of financial instruments, which requires disclosures about the fair value of financial instruments for interim reporting periods as well as in annual financial statements. Previously, these disclosures were only required in the Company s annual consolidated financial statements. The new guidance is effective for interim reporting periods ending after June 15, 2009, with early adoption permitted under specified circumstances. The Company adopted the new guidance beginning with the interim reporting period ended June 30, 2009 see Note 10 for the required disclosures. The adoption of the new guidance did not have an impact on the Company s financial position, results of operations or liquidity.

In May 2009, the FASB issued new guidance on subsequent events. The new guidance defines subsequent events as events or transactions that occur after the balance sheet date, but before financial statements are issued or are available to be issued, and establishes general standards of accounting for and disclosure of subsequent events. The new guidance also includes a required disclosure of the date through which an entity has evaluated subsequent events, which for public entities is the date upon which the financial statements are issued. The new guidance is effective for interim or annual financial periods ending after June 15, 2009, and shall be applied prospectively. The Company adopted the new guidance beginning with the interim period ended June 30, 2009. The Company evaluated subsequent events through February 8, 2010, which is the date the Company s consolidated financial statements for the interim period ended December 31, 2009 were issued (see Note 18 for the required disclosures). The adoption of the new guidance did not have an impact on the Company s financial position, results of operations or liquidity.

In August 2009, the FASB issued ASU No. 2009-05, Fair Value Measurements and Disclosures (Topic 820): Measuring Liabilities at Fair Value (ASU 2009-05). ASU 2009-05 provides additional guidance on how companies should measure liabilities at fair value, including alternative valuation methods and a hierarchy for their use. Specifically, when valuing a liability, an entity must use a quoted price in an active market for an identical liability if available. Otherwise, an entity may use one or more prescribed techniques, which in all instances should maximize the use of observable inputs and minimize unobservable inputs. Additionally, ASU 2009-05 clarifies that an entity is not required to include a separate adjustment relating to the existence of a restriction that prevents the transfer of a liability when measuring a liability at fair value. ASU 2009-05 is effective for the first reporting period (including interim periods) beginning after issuance. The Company adopted the additional guidance on measuring liabilities at fair value on October 1, 2009. The adoption of the new guidance did not affect the Company s liabilities subject to measurement at fair value, and did not impact the Company s financial position, results of operations or liquidity.

(b) Accounting pronouncements not yet adopted

In October 2009, the FASB issued ASU No. 2009-13, Revenue Recognition (Topic 605): Multiple-Deliverable Revenue Arrangements a consensus of the FASB Emerging Issues Task Force (ASU 2009-13) regarding the allocation of revenue in arrangements containing multiple deliverables. Specifically, ASU 2009-13 modifies existing GAAP by providing new guidance concerning (1) the determination of whether an arrangement involving multiple deliverables contains more than one unit of accounting, and (2) the manner in which arrangement consideration should be allocated to such deliverables. The guidance requires the use of an entity s best estimate of the selling price of a deliverable if vendor specific objective evidence or third-party evidence of the selling price cannot be determined. Additionally, ASU 2009-13 eliminates the use of the residual method of allocating consideration when vendor specific objective evidence or third-party evidence of the selling price is known for some, but not all, of the delivered items in a multiple element arrangement. Finally, ASU 2009-13 requires expanded qualitative and quantitative disclosures in the financial statements. ASU 2009-13 is effective for fiscal years beginning on or after June 15, 2010, with early adoption permitted. Upon adoption, the guidance may be applied either prospectively from the beginning of the fiscal year for new or materially modified arrangements, or it may be applied retrospectively. The Company currently has contracts in place that contain multiple deliverables, principally product supply agreements for gases and container rental. The Company treats the deliverables in these arrangements under current GAAP as separate units of accounting with selling prices derived from Company specific or third-party evidence, and the new guidance is not expected to significantly modify the accounting for these types of arrangements. The Company is continuing to evaluate the effects that ASU 2009-13 may

have on its consolidated financial statements.

8

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

In December 2009, the FASB issued ASU No. 2009-16, Transfers and Servicing (Topic 860): Accounting for Transfers of Financial Assets (ASU 2009 -16) which codifies a previous accounting standard and amends existing GAAP to establish new standards that will change how companies account for transfers of financial assets. Significant changes include (1) elimination of the qualifying special purpose entity (QSPE) concept, (2) new requirements for determining whether transfers of portions of financial assets are eligible for sale accounting, (3) clarification of the derecognition criteria for a transfer to qualify as a sale, (4) changes in recognition of gains or losses on transfers accounted for as sales, and (5) extensive new disclosures. ASU 2009-16 is effective for transfers of financial assets occurring in fiscal years beginning after November 15, 2009, and in interim periods within those fiscal years, with early adoption prohibited. The disclosure provisions of ASU 2009-16 shall be applied to transfers that occur both before and after the effective date of ASU 2009-16. The Company is currently evaluating the effects that ASU 2009-16 may have on its consolidated financial statements. The Company currently participates in a securitization agreement (the Agreement), which expires in March 2010, with three commercial banks to which it sells qualifying trade receivables on a revolving basis (see Note 4). The transaction has been accounted for as a sale under existing GAAP. Should the Company renew the Agreement or enter into a similar trade receivables securitization upon the expiration of the Agreement, the transaction will be evaluated under the provisions of ASU 2009-16. If applicable, this change in accounting treatment could significantly impact the Company s consolidated balance sheet, as the trade receivables sold under the securitization agreement and the related short-term borrowings may be recognized. However, the Company s debt covenants would not be impacted by the potential balance sheet recognition of the short-term borrowings, as borrowings under the Agreement are already factored into the debt covenant calculations.

In December 2009, the FASB issued ASU No. 2009-17, Consolidations (Topic 810): *Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities* (ASU 2009-17), which codifies a previous accounting standard. ASU 2009-17 establishes new standards that will change the consolidation model for variable interest entities (VIEs). Significant changes as a result of ASU 2009-17 include (1) changes in considerations as to whether an entity is a VIE, (2) a qualitative rather than quantitative assessment to identify the primary beneficiary of a VIE, (3) an ongoing rather than event-driven assessment of the VIE s primary beneficiary, and (4) the elimination of the QSPE scope exception. ASU 2009-17 is effective for fiscal years, and interim periods within those fiscal years, beginning after November 15, 2009 with early adoption prohibited. Upon adoption, if a VIE is required to be consolidated, an election must be made as to whether to retrospectively apply the guidance of ASU 2009-17 or record a cumulative-effect adjustment to retained earnings on the date of adoption. Additionally, if a VIE is deconsolidated upon the adoption of ASU 2009-17, a separate cumulative-effect adjustment to retained earnings will be recognized. The Company is currently evaluating the effects, if any, that ASU 2009-17 may have on its consolidated financial statements.

In January 2010, the FASB issued ASU No. 2010-06, Fair Value Measurements and Disclosures (Topic 820): *Improving Disclosures about Fair Value Measurements* (ASU 2010-06), which amends the disclosure guidance with respect to fair value measurements. Specifically, the new guidance requires disclosure of amounts transferred in and out of Levels 1 and 2 fair value measurements, a reconciliation presented on a gross basis rather than a net basis of activity in Level 3 fair value measurements, greater disaggregation of the assets and liabilities for which fair value measurements are presented and more robust disclosure of the valuation techniques and inputs used to measure Level 2 and 3 fair value measurements. ASU 2010-06 is effective for interim and annual reporting periods beginning after December 15, 2009, with the exception of the new guidance around the Level 3 activity reconciliations, which is effective for fiscal years beginning after December 15, 2010. The Company is currently evaluating the impact that ASU 2010-06 may have on its fair value measurement disclosures, but the new guidance will not impact the Company s financial position, results of operations or liquidity.

(3) ACQUISITIONS

Acquisitions occurring in fiscal 2010 were recorded using the acquisition method of accounting in accordance with the revised guidance on business combinations, which the Company adopted on April 1, 2009. The results of acquired companies operations have been included in the Company's consolidated financial statements since the effective date of each respective acquisition.

Purchase Price Allocation

During the nine months ended December 31, 2009, the Company purchased four businesses. The largest of these businesses was the November 2009 acquisition of Tri-Tech, a Florida-based industrial gas and welding supply distributor with 16 locations throughout Florida, Georgia and

South Carolina and historical annual revenues of approximately \$31 million. A total of \$70.4 million in cash was paid for these businesses, and an additional \$4.7 million was paid for the settlement of holdback liabilities associated with

9

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

certain prior year acquisitions. Transaction and other integration costs incurred for the nine months ended December 31, 2009 amounted to approximately \$300 thousand, which are included in selling, distribution and administrative expense in the Company s consolidated statement of earnings. These businesses had aggregate historical annual revenues of approximately \$43 million. Net sales from the acquired businesses that are included in the Company s results for the nine months ended December 31, 2009 are approximately \$4.8 million. The Company acquired these businesses in order to expand its geographic coverage and strengthen its national network of branch-store locations.

The following table summarizes the fair values of the assets acquired and liabilities assumed related to fiscal 2010 acquisitions, as well as adjustments related to certain prior year acquisitions. Purchase price allocations are based on third-party valuations and management s estimates. The purchase price allocations related to fiscal 2010 acquisitions continue to be based on provisional fair values and are subject to revision as the Company finalizes appraisals and other analyses. Final determination of the fair values may result in further adjustments to the values presented below.

(In thousands)	Distribution Business Segment	All Other Operations Business Segment	Total
Current assets, net	\$ 5,221	\$ 543	\$ 5,764
Property and equipment	21,981	113	22,094
Goodwill	38,046	1,016	39,062
Other intangible assets	14,201	1,035	15,236
Current liabilities	(2,375)	(1,096)	(3,471)
Long-term liabilities	(3,462)	(156)	(3,618)
Total cash consideration	\$ 73,612	\$ 1,455	\$ 75,067

The fair value of the assets acquired in fiscal 2010 includes trade receivables of \$3.8 million with gross contractual amounts receivable of \$4.0 million, of which \$200 thousand is expected to be uncollectible. The fiscal 2010 acquisitions resulted in the recognition of \$30.9 million of goodwill, which is deductible for income tax purposes. Goodwill is attributable to the workforce of the acquired business, expected synergies to be realized with the acquisition, and the expansion of geographical coverage that will facilitate the sale of industrial, medical, and specialty gases and related supplies. See Note 6 for further information on goodwill, including its allocation by segment.

Linde Purchase Accounting Integration Accruals

In connection with the June 2007 Linde Packaged Gas acquisition, the Company recorded accruals associated with one-time severance benefits for acquired employees who were involuntarily terminated, facility exit-related costs associated with exiting certain acquired facilities that overlapped with the Company s existing operations, and a multi-employer pension plan withdrawal liability associated with renegotiating certain union contracts. Of the \$3.3 million remaining facility exit accrual at December 31, 2009, \$3.0 million relates to a non-cancellable lease obligation through 2017 for the former Linde corporate headquarters. Other integration accruals of \$4.2 million at December 31, 2009 consist primarily of the multi-employer pension liability, which is expected to be paid during fiscal 2011 and 2012. The table below summarizes the liabilities established through purchase accounting and the related payments made from the acquisition date through March 31, 2009 and during the nine months ended December 31, 2009:

(In thousands)	Severai	nce Accruals	ility Exit ccruals	Integration ecruals	Total Integration Accruals
	\$	5,517	\$ 6,088	\$ 6,434	

Edgar Filing: AIRGAS INC - Form 10-Q

Amounts originally included in purchase accounting, including adjustments				\$ 18,039
Payments	(5,358)	(2,510)	(1,801)	(9,669)
Balance at March 31, 2009	159 (159)	3,578	4,633 (479)	8,370 (947)
Payments	(139)	(309)	(479)	(947)
Balance at December 31, 2009	\$	\$ 3,269	\$ 4,154	\$ 7,423

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

Pro Forma Operating Results

The following table provides unaudited pro forma results of operations for the nine-month periods ended December 31, 2009 and 2008 as if fiscal 2010 and 2009 acquisitions had occurred on April 1, 2008. The pro forma results were prepared from financial information obtained from the sellers of the businesses, as well as information obtained during the due diligence process associated with the acquisitions. The unaudited pro forma results reflect certain adjustments related to the acquisitions, such as increased depreciation and amortization expense resulting from the Company s stepped-up basis in acquired assets and adjustments to reflect the Company s borrowing and tax rates. The pro forma operating results do not include any anticipated synergies related to combining the businesses. Accordingly, such pro forma operating results were prepared for comparative purposes only and do not purport to be indicative of what would have occurred had the acquisitions been made as of April 1, 2008 or of results that may occur in the future.

	Nine Mo	Nine Months Ended				
	Decer	nber 31,				
(In thousands, except per share amounts)	2009	2008				
Net sales	\$ 2,911,543	\$ 3,474,742				
Net earnings	157,821	206,840				
Diluted earnings per share	\$ 1.89	\$ 2.46				

(4) TRADE RECEIVABLES SECURITIZATION

The Company participates in a securitization agreement (the Agreement) with three commercial banks to which it sells qualifying trade receivables on a revolving basis. The maximum size of the facility is \$345 million and the Agreement contains customary events of termination, including standard cross default provisions with respect to outstanding debt. During the nine months ended December 31, 2009, the Company sold approximately \$2.6 billion of trade receivables and remitted to bank conduits, pursuant to a servicing agreement, approximately \$2.6 billion in collections on those receivables. The amount of receivables sold under the Agreement was \$268 million at December 31, 2009 and \$311 million at March 31, 2009.

The transaction has been accounted for as a sale, as control of the receivables has been surrendered. Under the Agreement, trade receivables are sold to bank conduits through a bankruptcy-remote special purpose entity, which is consolidated for financial reporting purposes. The difference between the proceeds from the sale and the carrying value of the receivables is recognized as Discount on securitization of trade receivables in the accompanying Consolidated Statements of Earnings and varies on a monthly basis depending on the amount of receivables sold and market rates. The Company retains a subordinated interest in the receivables sold, which is recorded at the receivables previous carrying value. Accordingly, the Company is exposed to credit risk associated with its retained interest in the receivables. The Company is not exposed to interest rate risk due to the short-term nature of the receivables and their general collectability.

Subordinated retained interests of \$149.6 million, net of an allowance for doubtful accounts of \$26.6 million, and \$147.9 million, net of an allowance for doubtful accounts of \$26.4 million, are included in trade receivables in the accompanying Consolidated Balance Sheets at December 31, 2009 and March 31, 2009, respectively. At December 31, 2009 and March 31, 2009, approximately 7.2% and 6.4%, respectively, of the accounts included in the retained interest were delinquent, as defined under the Agreement. Credit losses for the nine months ended December 31, 2009 and December 31, 2008 were \$11.7 million and \$15.3 million, respectively. On a monthly basis, management calculates the fair value of the retained interest based on management s best estimate of the undiscounted expected future cash collections on the receivables. Changes in the fair value are recognized as bad debt expense. Actual cash collections may differ from these estimates and would directly affect the fair value of the retained interest. In accordance with the servicing agreement, the Company continues to service, administer and collect the trade receivables on behalf of the bank conduits. The servicing fees charged to the bank conduits approximate the costs of collections. Accordingly, the net servicing asset is immaterial. The Company does not provide any financial guarantees of the bank conduits obligations.

The Agreement expires in March 2010. The Company is evaluating the renewal of the current arrangement and continues to evaluate this and other financing alternatives. Other financing alternatives include drawing down on the unused portion of the Company s Credit Facility (see Note 8), which was \$523 million at December 31, 2009, issuing additional senior notes or any combination of renewing a smaller agreement, issuing senior notes and financing under the Company s Credit Facility. The Company anticipates renewing the Agreement or having other financing alternatives in place before the Agreement s expiration in March 2010.

11

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

(5) **INVENTORIES, NET**

Inventories, net, consist of:

(In thousands)	mber 31, 2009	March 31, 2009
Hardgoods	\$ 231,527	\$ 252,293
Gases	115,640	138,152
	\$ 347,167	\$ 390,445

Hardgoods inventories determined using the LIFO inventory method totaled \$34 million at December 31, 2009 and March 31, 2009. The balance of the hardgoods inventories is valued using the FIFO and average-cost inventory methods. If the hardgoods inventories valued under the LIFO method had been valued using the FIFO method, the carrying value of hardgoods inventory would have been \$10.0 million higher at December 31, 2009 and \$10.4 million higher at March 31, 2009. Substantially all of the inventories are finished goods.

(6) GOODWILL AND OTHER INTANGIBLE ASSETS

Goodwill is the excess of cost of an acquired entity over the amounts assigned to assets acquired and liabilities assumed in a purchase business combination. The valuations of assets acquired and liabilities assumed from recent acquisitions are based on preliminary estimates of fair value and are subject to revision as the Company finalizes appraisals and other analyses. Changes in the carrying amount of goodwill for the nine months ended December 31, 2009 were as follows:

(In thousands)	Distribution Business Segment	All Other Operations Business Segment	Total
Balance at March 31, 2009	\$ 879,082	\$ 184,288	\$ 1,063,370
Acquisitions (1)	38.046	1.016	39,062
Other adjustments, including foreign currency translation	2,488	1,169	3,657
Balance at December 31, 2009	\$ 919,616	\$ 186,473	\$ 1,106,089

Test for Goodwill Impairment

The Company is required to perform an assessment of the carrying value of goodwill associated with each of its reporting units at least annually and whenever events or circumstances indicate that it is more likely than not that goodwill may be impaired. The Company elected to perform its annual assessment of the carrying value of goodwill as of October 31 of each year. The annual assessment of the carrying value of goodwill at October 31, 2009 indicated that the Company s goodwill was not impaired.

⁽¹⁾ Includes current acquisitions and adjustments made to prior year acquisitions.

As of October 31, 2009, the Company had 18 reporting units in the Distribution business segment and six reporting units in the All Other Operations business segment. The Company determined the estimated fair value of each of its reporting units as of October 31, 2009 using a discounted cash flow model and compared those values to the carrying value of each of the respective reporting units. Significant assumptions used in the cash flow model include revenue growth rates and profit margins based on specific reporting unit business plans, future capital expenditures, working capital needs, discount rates and perpetual growth rates. At October 31, 2009, the discount rates used in the model were 10% for the Distribution reporting units and 11.0% to 11.5% for the smaller reporting units in the All Other Operations business segment. The perpetual growth rate assumed in the discounted cash flow model was consistent with the long-term rate of growth as measured by the U.S. Gross Domestic Product. In addition to Company specific growth targets, general economic conditions, the long-term economic outlook for the U.S. economy, and market conditions affecting borrowing costs and returns on equity all influence the estimated fair value of each of the Company s reporting units.

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

Although the annual assessment indicated that the Company s goodwill was not impaired, it also denoted that for all but one of the Company s reporting units, the fair value of each of the reporting units exceeded its carrying value by a substantial amount. However, for the Company s refrigerants business, a reporting unit in the All Other Operations business segment, the annual assessment indicated that its fair value exceeded its carrying value by 7%. Therefore, a hypothetical 7% reduction in that reporting unit s fair value would be an indication of impairment and trigger the need to perform a second step to measure the amount of impairment, if any. The forecast cash flows for this reporting unit reflect a changing regulatory environment, which the Company believes will contribute to revenue growth and improved profitability. Should the markets in which this reporting unit operates not respond as anticipated to the changing regulatory environment or an alternative to the reporting unit s products be developed, the forecast cash flows and consequently the reporting unit s estimated fair value would be negatively affected. The amount of goodwill associated with this reporting unit was \$88 million at December 31, 2009.

Other Intangible Assets

Other intangible assets amounted to \$216 million, net of accumulated amortization of \$50 million at December 31, 2009 and \$216 million, net of accumulated amortization of \$39 million at March 31, 2009. These intangible assets primarily consist of acquired customer relationships, which are amortized over 7 to 17 years, and non-competition agreements, which are amortized over the term of the agreements. The determination of the estimated benefit period associated with customer relationships is based on the analysis of historical customer sales attrition information and other customer-related factors at the date of acquisition. There are no expected residual values related to these intangible assets. The Company evaluates the estimated benefit periods and recoverability of its intangible assets when facts and circumstances indicate that the lives may not be appropriate and/or the carrying value of the assets may not be recoverable. If the carrying value of an intangible asset is determined to not be recoverable, it will be written down to its estimated fair value. Fair value is generally estimated based on either appraised value or other valuation techniques. Intangible assets also include trade names with indefinite useful lives valued at \$1.3 million. Estimated future amortization expense by fiscal year is as follows: remainder of 2010 - \$5.8 million; 2011 - \$22.9 million; 2012 - \$21.7 million; 2013 - \$20.8 million; 2014 - \$18.4 million; 2015 \$16.9 million; and \$108.5 million thereafter.

(7) ACCRUED EXPENSES AND OTHER CURRENT LIABILITIES

Accrued expenses and other current liabilities include:

(In thousands)	De	cember 31, 2009	March 31, 2009
Accrued payroll and employee benefits	\$	77,312	\$ 80,630
Business insurance reserves		43,442	44,986
Taxes other than income taxes		15,625	17,098
Cash overdraft		58,653	42,933
Deferred rental revenue		25,105	25,611
Other accrued expenses and current liabilities		58,279	53,306
	\$	278,416	\$ 264,564

With respect to the business insurance reserves above, the Company had corresponding insurance receivables of \$9.9 million at December 31, 2009 and \$9.7 million at March 31, 2009. The insurance receivables represent the balance of probable claim losses in excess of the Company s self-insured retention for which the Company is fully insured.

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

(8) INDEBTEDNESS

Long-term debt consists of:

(In thousands)	December 31, 2009	March 31, 2009
Revolving credit borrowings - U.S.	\$ 495,400	\$ 751,200
Revolving credit borrowings - Multi-currency	31,627	23,712
Revolving credit borrowings - Canadian	11,858	14,660
Term loans	330,000	397,500
Senior notes, net of discount	399,544	
Senior subordinated notes	328,534	550,000
Acquisition and other notes	17,210	24,294
Total long-term debt	1,614,173	1,761,366
Less current portion of long-term debt	(10,103)	(11,058)
Long-term debt, excluding current portion	\$ 1,604,070	\$ 1,750,308

Senior Subordinated Note Redemption

On October 13, 2009, the Company redeemed its \$150 million, 6.25% senior subordinated notes maturing July 15, 2014 (the 2004 Notes) at a price of 103.125% of the principal. A loss on the early extinguishment of debt of approximately \$6.1 million (\$3.8 million after tax) was recognized related to the redemption premium and write-off of unamortized debt issuance costs.

During the three and nine months ended December 31, 2009, the Company repurchased \$13.4 million and \$71.5 million, respectively, of its 7.125% senior subordinated notes maturing October 1, 2018 (the 2008 Notes). The 2008 Notes were repurchased at an average price of 102.5%. Losses on the early extinguishment of the 2008 Notes were \$607 thousand (\$382 thousand after tax) and \$2.6 million (\$1.7 million after tax) for the three- and nine-month periods ended December 31, 2009, respectively, and related to the redemption premiums and write-off of unamortized debt issuance costs.

Senior Credit Facility

The Company maintains a senior credit facility (the Credit Facility) with a syndicate of lenders. At December 31, 2009, the Credit Facility permitted the Company to borrow up to \$991 million under a U.S. dollar revolving credit line, up to \$75 million (U.S. dollar equivalent) under a multi-currency revolving credit line, and up to C\$40 million (U.S. \$38 million) under a Canadian dollar revolving credit line. The Credit Facility also contains a term loan provision through which the Company borrowed \$600 million with scheduled repayment terms. The term loans are repayable in quarterly installments of \$22.5 million through June 30, 2010. The quarterly installments then increase to \$71.2 million from September 30, 2010 to June 30, 2011. Principal payments due over the next twelve months on the term loans are classified as Long-term debt in the Company s Consolidated Balance Sheets based on the Company s ability and intention to refinance the payments with borrowings under its long-term revolving credit facilities. As principal amounts under the term loans are repaid, no additional borrowing capacity is created under the term loan provision. The Credit Facility will mature on July 25, 2011.

As of December 31, 2009, the Company had approximately \$869 million of borrowings under the Credit Facility: \$495 million under the U.S. dollar revolver, \$330 million under the term loans, \$32 million (in U.S. dollars) under the multi-currency revolver and C\$12 million (U.S. \$12

million) under the Canadian dollar revolver. The Company also had outstanding letters of credit of \$42 million issued under the Credit Facility. During the second fiscal quarter, the Company s credit ratings were upgraded resulting in a lowering of the interest rate spreads on the borrowings above. The U.S. dollar revolver borrowings and the term loans bear interest at the London Interbank Offered Rate (LIBOR) plus 50 basis points. The multi-currency revolver bears interest based on a spread of 50 basis points over the Euro currency rate applicable to each foreign currency borrowing. The Canadian dollar borrowings bear interest at the Canadian Bankers Acceptance Rate plus 50 basis points. As of December 31, 2009, the average effective interest rates on the U.S. dollar revolver, the term loans, the multi-currency revolver and the Canadian dollar revolver were 0.72%, 0.75%, 0.99% and 0.99%, respectively.

At December 31, 2009, the financial covenants of the Credit Facility did not restrict the Company s ability to borrow on the unused portion of the Credit Facility. The Credit Facility contains customary events of default, including nonpayment and breach covenants. In the event of default, repayment of borrowings under the Credit Facility may be accelerated. The Company s Credit Facility also contains cross-default provisions whereby a default under the Credit Facility would likely result in defaults under the

14

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

senior subordinated notes discussed below. As of December 31, 2009, approximately \$523 million remained unused under the Company s Credit Facility.

Effective July 31, 2009, the Company was upgraded to an investment grade credit rating and with the October 13, 2009 redemption of the 2004 Notes, the guarantees and collateral under the Credit Facility were released. Prior to the redemption of the 2004 Notes, the Company's domestic subsidiaries, exclusive of the bankruptcy-remote special purpose entity (the domestic subsidiaries), guaranteed the U.S. dollar revolver, term loans, multi-currency revolver and Canadian dollar revolver. The multi-currency revolver and Canadian dollar revolver were also guaranteed by the Company's foreign subsidiaries. The guarantees were full and unconditional and were made on a joint and several basis. The Company had pledged 100% of the stock of its domestic subsidiaries and 65% of the stock of its foreign subsidiaries as surety for its obligations under the Credit Facility. The Credit Facility provided for the release of the guarantees and collateral if the Company attained an investment grade credit rating and a similar release on its 2004 Notes.

Money Market Loans

The Company has an agreement with a financial institution that provides access to short-term advances not to exceed \$35 million. The agreement expires on December 1, 2010, but may be extended subject to renewal provisions contained in the agreement. The advances are generally overnight or for up to seven days. The amount, term and interest rate of an advance are established through mutual agreement with the financial institution when the Company requests such an advance. At December 31, 2009, there were no advances outstanding under the agreement.

Senior Notes

On September 11, 2009, the Company issued \$400 million of senior notes (the 2009 Notes). The 2009 Notes were issued at a discount and mature on September 15, 2014 with an effective yield of 4.527%. The net proceeds from the sale of the 2009 Notes were used to reduce the borrowings under the Company s revolving credit line under the Credit Facility. The 2009 Notes bear interest at a fixed annual rate of 4.5%, payable semi-annually on March 15 and September 15 of each year, commencing March 15, 2010. Additionally, the Company has the option to redeem the 2009 Notes prior to their maturity, in whole or in part, at 100% of the principal plus any accrued but unpaid interest and any applicable make-whole amounts.

Senior Subordinated Notes

At December 31, 2009, the Company had \$328.5 million of its 2008 Notes outstanding with a maturity date of October 1, 2018. The 2008 Notes bear interest at a fixed annual rate of 7.125%, payable semi-annually on October 1 and April 1 of each year. The 2008 Notes have an optional redemption provision, which permits the Company, at its option, to call the 2008 Notes at scheduled dates and prices. The first scheduled optional redemption date is October 1, 2013 at a price of 103.563% of the principal amount.

The 2008 and 2009 Notes contain covenants that could restrict the payment of dividends, the repurchase of common stock, the issuance of preferred stock, and the incurrence of additional indebtedness and liens. With the July 31, 2009 credit rating upgrades and the October 13, 2009 redemption of the 2004 Notes, the subsidiary guarantees on the 2008 and 2009 Notes were released.

Acquisition and Other Notes

The Company s long-term debt also includes acquisition and other notes, principally consisting of notes issued to sellers of businesses acquired, which are repayable in periodic installments. At December 31, 2009, acquisition and other notes totaled \$17 million with an average interest rate of approximately 6% and an average maturity of approximately two years.

Aggregate Long-term Debt Maturities

The aggregate maturities of long-term debt at December 31, 2009 are as follows:

(In thousands)	Del	ot Maturities
December 31, 2010 (1)	\$	10,103
March 31, 2011		73,339
March 31, 2012		800,852
March 31, 2013		727
March 31, 2014		518
Thereafter (1)		729,090
	\$	1,614,629

15

AIRGAS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(UNAUDITED)

(1) The Company has theate the merger agreement and abandon the merger;

that the parties will not have performed in all material respects their obligations contained in the merger agreement before the closing date;

that the representations and warranties made by the parties in the merger agreement will not be true and correct as of the closing of the merger in a manner which results in a material adverse effect on us;

that there is litigation that restrains or prohibits the merger;

required regulatory notification and reporting requirements required under the HSR Act are not completed; and

that a governmental entity will have enacted a law, rule, regulation or order that imposes a restraint that prohibits the merger.

As a result of various risks to the completion of the merger, there can be no assurance that the merger will be completed even if the requisite stockholder approval is obtained.

Merger Financing

The total amount of funds necessary to complete the merger and to pay the related fees and expenses is estimated to be approximately (i) \$ for Media Arts Group and (ii) \$ for Mergerco. Media Arts Group and Mergerco anticipate that the merger consideration will be funded from two primary sources:

Our existing cash, cash equivalents and working capital, net of amounts necessary for our ongoing business needs.

Financing from GE Corporate Financial Services, Inc. (or one of its affiliates) in the form of (i) a two-year term loan in the amount of \$3 million, and (ii) a three-year revolving line of credit facility in the amount of \$22 million.

As of September 30, 2003, our net working capital was \$42.6 million, which included cash and cash equivalents of \$22.6 million. Assuming the merger is completed, we currently anticipate that we will be able to generate sufficient cash together with amounts available under the new credit facility to fund our ongoing operations, including working capital needs, for at least the next 12 months.

37

Mergerco has obtained a commitment letter with GE Corporate Financial Services, Inc. regarding the credit facility and term loan financing. This commitment letter provides for the following principal terms:

For the revolving line of credit facility, advances will be made against a formula based on up to 70% of eligible accounts receivable, plus up to 85% of the value of eligible lithographed sheet and canvas inventory, plus certain additional eligible other inventory, less reserves. The interest rate for the credit facility will be equal to the index rate for 30-day commercial paper plus 4.00%.

For the term loan, the interest rate will be equal to the index rate plus 4.50%.

The revolving line of credit facility has a maturity of three years.

The term loan has a maturity of two years from funding.

This commitment is subject to fulfillment of several conditions to the satisfaction of GE Corporate Financial Services, Inc., including, among others, the following:

contribution by Mr. Kinkade to Media Arts Group of not less than \$1.0 million of additional subordinated debt financing on terms that are acceptable to GE Corporate Financial Services, Inc. (including paid-in-kind interest payments);

the absence of any material adverse change in the business, financial or other condition of Media Arts Group or its business or prospects, the industry in which it operates, or the collateral securing the loans;

the absence of litigation which, if successful, would have a material adverse effect on Media Arts Group;

the absence of litigation challenging the merger which, if successful, could have a material adverse impact on the liens securing the loans or the prospects for repayment of the loans;

the pledge of collateral and other security by Mr. Kinkade and his affiliates, including personal guarantees secured by personal real estate and other assets, the assignment of rights under Mr. Kinkade's license agreement with Media Arts Group and a pledge of their stock in Media Arts Group;

satisfactory completion of any additional due diligence;

completion of definitive agreements and security arrangements satisfactory to GE Corporate Financial Services, Inc.; and

other customary conditions.

As of the date of this proxy statement, the definitive loan documents have not been executed. Execution of the loan documents is anticipated at or around the completion of the merger.

Mergerco and Mr. Kinkade will not be required to complete the merger unless they have received the funding contemplated by the commitment letter from GE Corporate Financial Services, Inc. or otherwise have access to sufficient funds under any other commitment that is acceptable to them to enable Mergerco to make the payments contemplated by the merger agreement. As a result, even if we receive the requisite approvals of our stockholders that are necessary to complete the merger, Mergerco and Mr. Kinkade might not complete the merger if they do not receive the necessary financing to complete the merger.

Mergerco and Mr. Kinkade have agreed to use commercially reasonable efforts to procure the financing necessary to complete the merger. However, we cannot assure you that Mergerco and Mr. Kinkade will be able to obtain such financing and, as noted above, if they fail to do so, we will not be able to complete the merger.

Certain Relationships and Related Transactions

Transactions with Management and Others.

License Agreement with Thomas Kinkade. Effective December 3, 1997, we entered into a license agreement with Mr. Kinkade. Under the license agreement, Mr. Kinkade receives a flat fee of \$25,000 per painting delivered to us for reproduction and he was paid a royalty of 4.5% of the net revenues of Thomas Kinkade branded artwork and products through May 8, 2000 and he is currently paid a royalty of 5.0% of the net revenues of Thomas Kinkade branded artwork and products. Mr. Kinkade is also employed by us as Art Director.

Under the terms of our license agreement with Mr. Kinkade, we have the complete, unencumbered, exclusive and perpetual rights to reproduce, adapt, manufacture, sub-license, publish, market, distribute, sell and display all art-based and non-art-based products and services associated with Mr. Kinkade and his "artwork." The "artwork" includes at least 150 paintings that Mr. Kinkade is required to provide to us between December 1997 and December 2012 (with at least 10 paintings to be delivered during each of the first five years), and all original sketches, drawings, writings, paintings and other works of art created by Mr. Kinkade prior to December 1997 (including "archive" images) and during the term of the license agreement. The license agreement also grants us the right to use the name and likeness of Mr. Kinkade in promoting the sale of our products and development of any brand name associated with Mr. Kinkade. Our creative services department collaborates with Mr. Kinkade, dealers, the sales department and strategic business partners to assist in the creation and development of new products. Proposed new products are tested and pre-marketed prior to product introductions. Mr. Kinkade has the right to approve our products based upon his artwork, as well as promotional materials, business plans and strategic relationships relating to such products or the use of his name or likeness. Mr. Kinkade retains ownership of the original paintings he produces.

The license agreement permits Mr. Kinkade, independently from Media Arts Group, to reproduce up to two pieces annually to be sold in and around the City of Placerville, California. Mr. Kinkade also retained the right to use his name, likeness and certain artwork in association with non-profit organizations. In addition, Mr. Kinkade retained the right to use his name in connection with for-profit ventures with our prior consent, provided that he first offers the opportunity to us. Mr. Kinkade is otherwise subject to a non-compete agreement with us under the license agreement.

The license agreement is terminable by either party after failure by the other party for 90 days to cure a material breach of the agreement. In addition, Mr. Kinkade may terminate the license agreement in the event of our insolvency or upon a change of control of us. A change in control is defined to occur on the date when any person or group (as defined in Rule 13(d)(3) under the Exchange Act) beneficially owns (as defined in such Rule) a number of shares of our common stock in excess of the number of shares then beneficially owned by Mr. Kinkade. The computation excludes stockholders as of December 3, 1997, to the extent of their beneficial holdings of common stock as of such date. Mr. Kinkade cannot invoke this right of termination if it is triggered as a result of Mr. Kinkade's transfer of shares. After December 3, 2012, the perpetual nature of the license agreement may be terminated by Mr. Kinkade if we engage in any material business enterprises unrelated to his work or brand name to which he objects. Upon any termination of the license agreement by Mr. Kinkade, we would be prohibited from selling any products based upon Mr. Kinkade's artwork, other than our then existing product inventory.

We incurred royalties to Mr. Kinkade under the license agreement in the amount of approximately \$2.9 million, \$5.7 million, \$4.7 million, \$8.2 million and \$9.7 million for the nine month period ended September 30, 2003, the fiscal year ended December 31, 2002, the nine-month period ended December 31, 2001 and the fiscal years ended March 31, 2001 and 2000, respectively.

In addition, pursuant to the license agreement, we and Mr. Kinkade entered into a stock option agreement as of December 3, 1997, wherein we granted to Mr. Kinkade a fifteen-year non-statutory

39

option to purchase 600,000 shares of our common stock at \$12.375, the closing price on the date of grant. The option will be cancelled as a result of the merger.

Employment Agreement with Thomas Kinkade. By mutual verbal agreement as of January 1, 1999, Mr. Kinkade and Media Arts Group extended certain terms of Mr. Kinkade's employment agreement dated January 2, 1994 for his services as Art Director (which written employment agreement expired on December 31, 1998). Under the oral agreement, Mr. Kinkade is paid an annual base salary of

\$172,000, and is entitled to receive certain health benefits, vacation time, fringe benefits and reimbursement of certain expenses. In addition, under certain circumstances, we are required to indemnify Mr. Kinkade. The oral agreement is renewable year-to-year.

Certain Business Relationships.

Licensing Agent Agreement with Creative Brands Group. Media Arts Group and Creative Brands Group, Inc. ("CBG") are parties to a contract dated July 17, 2001, as amended as of August 1, 2002, pursuant to which CBG manages and develops licensing arrangements with certain of our business partners. The contract expires July 31, 2006. Under the contract, as amended, CBG receives 24% of licensing revenues received by us from the business partners that CBG manages on behalf of us. CBG is primarily owned by Kenneth E. Raasch, a significant stockholder, co-founder, former Chief Executive Officer and former member of our board of directors. For the year ended December 31, 2002 and the nine-month period ended September 30, 2003, we incurred commissions of approximately \$1.8 million and \$1.9 million to CBG, respectively.

Trade Accounts Receivable. Effective September 30, 2001, we converted \$1.6 million of trade accounts receivable, due from an entity in which a relative of Mr. Kinkade is an investor, into a full recourse three year promissory note. The balance is repayable in equal installments during the three-year period and bears interest of 12% per annum. The terms of this note are consistent with those of other notes we have received from non-related parties in the sale of company-owned Thomas Kinkade stores. In addition, the related party is indebted to us for the sale of company-owned Thomas Kinkade stores in Monterey and Carmel, California as discussed below. The note is in default and we have commenced legal proceedings to recover the full amount of the note.

Sale of Company-Owned Store. Effective September 15, 1999, we sold company-owned Thomas Kinkade Stores located in Monterey and Carmel, California to an entity in which a relative of a Mr. Kinkade is an investor. Consideration for this sale consisted of cash of \$75,000 and an 8.5% promissory note in the amount of \$1.1 million due in August 2006 with semi-annual principal and interest payments. The inventory in the stores at the time of the transfer was sold under a separate transaction at normal wholesale prices with extended payment terms. The terms of this company-owned store sale are consistent with the terms for the sales of other company-owned stores to non-related parties. We commissioned an independent valuation of company-owned stores in Monterey and Carmel, California and the terms of the sale were based upon the independent valuation. At December 31, 2001, we determined that the notes described above were impaired and accordingly wrote them down to estimated realizable value. The note is in default and we have commenced legal proceedings to recover the full amount of the note.

Charitable Contributions to World Vision U.S. From time to time, we make donations to, contribute product to, or otherwise assist, certain charities. From time to time, we engaged in such charitable activities with World Vision U.S. The charitable contributions to World Vision U.S. are not material to our financial condition. Richard Stearns, a member of our board of directors, is President of World Vision U.S.

Sales to Thomas Kinkade Foundation. From time to time, we have sold products to the Thomas Kinkade Foundation at our cost. The Thomas Kinkade Foundation is an organization established by Mr. Kinkade and Nannette Kinkade. Mr. Kinkade is a co-founder of Media Arts Group and a member

40

of our board of directors and the chairman of the board of the Thomas Kinkade Foundation. The sales to The Thomas Kinkade Foundation are not material to our financial condition.

Interests of Certain Persons in the Merger; Potential Conflicts of Interest

The Board of Directors. In considering the recommendations of our board of directors, our stockholders should be aware that certain of our executive officers and directors have interests in the transaction that are different from, or are in addition to, the interests of our stockholders generally. None of the independent members of the board of directors who considered this proposal (in the case of Mr. Halvorson, from and until June 22, 2003, at which time he ceased to be an independent director and recused himself from the board deliberations related to the proposed merger) are current or former officers or employees of Media Arts Group, and none of them have any financial interest in the proposed merger different from our stockholders generally, to evaluate, negotiate and recommend the merger agreement and to evaluate whether the merger is in the best interests of our stockholders (other than Mergerco and the affiliated stockholders). The board of directors acting through the independent directors was aware of the differing interests of other directors and officers and considered them, among other matters, in evaluating and negotiating the merger agreement and the merger.

Fees and Expenses to Independent Directors. Each of the independent directors received \$1,000 for each meeting attended in connection with the merger. As of the date of this proxy statement, Ms. Grzelakowski has received \$17,000, Mr. LaBonte has received

\$19,000, Mr. Potter has received \$19,000, and Mr. Stearns has received \$15,000 in fees, representing an aggregate fee of \$70,000 for their services performed since October 29, 2002 as our independent directors in connection with their consideration of the merger, strategic alternatives to the merger, and related matters. Each member was also reimbursed for his or her out-of-pocket expenses. Additionally, Mr. Halvorson received \$13,000 for the 13 independent director meetings he attended during the period between October 29, 2002 and June 23, 2003, during which time he was an independent director.

Management of Media Arts Group Following the Merger. Eric Halvorson is our President and a director, Herbert D. Montgomery is our Chief Financial Officer, an Executive Vice President and was a director until August 2003. It is expected that Messrs. Halvorson and Montgomery will continue in these capacities after completion of the merger. The employment agreements currently in effect regarding the terms of Messrs. Halvorson and Montgomery employment are expected to continue with the surviving corporation. The independent members of the board of directors will not continue as directors, and will have no other involvement with Media Arts Group, following the merger.

Success Fee to Management. Mr. Kinkade has agreed to pay each of Mr. Thomopoulos, our Chairman and Chief Executive Officer, Mr. Halvorson, our President, and Mr. Montgomery, our Chief Financial Officer, a success fee of \$100,000 upon completion of the merger for their additional responsibilities and services in connection with the merger agreement and the merger.

Management Involvement in Preparation of Projections. Messrs. Halvorson and Montgomery were involved in preparing the projections that were used by Jefferies in rendering their fairness opinion and by the board of directors in considering the fairness of the merger.

Pre-existing Indemnification and Insurance. Our certificate of incorporation and bylaws provide that we will indemnify our directors and executive officers to the fullest extent permitted by Delaware law. We also maintain directors and officers' liability insurance for the benefit of such persons.

Indemnification and Insurance After the Merger. The merger agreement provides that after the effective time, the surviving corporation will fulfill and honor in all respects the obligations of Media Arts Group pursuant to any indemnification agreement currently in effect between us and each person who is or was a director or officer of ours at or prior to the effective time and any indemnification provisions under our current certificate of incorporation or bylaws as each is in effect on the date of the merger agreement. In addition, the merger agreement provides that the surviving corporation will

41

preserve the rights of individuals who were, prior to the effective time, our directors or officers with respect to indemnification and exculpation from liability set forth in any such indemnification agreement or our current certificate of incorporation and bylaws as of the date of the merger agreement for a period of six years from the effective time of the merger.

The merger agreement also provides that the surviving corporation will, up to a maximum cost, either (i) cause to be maintained for six years from the effective time of the merger our current directors' and officers' liability policy with respect to claims arising from facts or events that occurred at or prior to the effective time of the merger, (ii) extend the discovery or reporting period under our current policy for six years from the effective time of the merger with respect to such claims, or (iii) substitute coverage with respect to such claims under a different policy on no less advantageous terms.

Amounts to be Received by Directors and Officers in the Merger. Upon consummation of the merger, our director Mr. Kinkade will not receive any amounts in connection with his ownership of our shares of common stock. Our other directors or officers who are not affiliated stockholders will receive the same merger consideration and option cash-out amount as other stockholders.

Acceleration of Stock Options. Any stock option (whether or not vested) that has not been exercised prior to the merger will be converted into the right to receive cash if the exercise price of the option is less than \$4.00 per share (other than stock options held by Mr. Kinkade). Management employees and directors (other than Mr. Kinkade) hold stock options, many of which have exercise prices below \$4.00 per share and will be accelerated by the merger. As a result of the merger, options to purchase 545,000 shares of our common stock held by our executive officers and directors (other than Mr. Kinkade) will be cashed out. Mr. Kinkade's stock options will be canceled without payment. In the aggregate, our executive officers and directors will receive payment of \$694,487 as a result of their options being cashed out in the merger.

Estimated Fees and Expenses of the Merger

Whether or not the merger is completed, in general, all fees and expenses incurred in connection with the merger will be paid by the party incurring those fees and expenses. Under certain circumstances described in "THE MERGER AGREEMENT Fee and Expense Reimbursement," we will reimburse Mergerco, or Mergerco and Mr. Kinkade will reimburse us under certain circumstances, up to \$1.0 million for reasonable out-of-pocket expenses incurred in connection with the merger. The estimated total fees and expenses to be incurred by us and by Mr. Kinkade and his affiliates in connection with the merger are as follows:

Legal fees	\$
Investment banker fees	
Accounting fees	
Printing, proxy solicitation and mailing expenses	
Total	\$

These expenses will not reduce the merger consideration to be received by our stockholders.

Appraisal Rights

If the merger is consummated, holders of our common stock who follow the procedures set forth below will be entitled to appraisal rights under Section 262 of the Delaware General Corporation Law, a copy of which is attached hereto as Appendix C. Stockholders who perfect their appraisal rights and follow certain procedures in the manner prescribed by the Delaware General Corporate Law will be entitled to receive, in lieu of the \$4.00 merger consideration, a cash payment equal to the "fair value" of their shares of our common stock.

42

If the merger is consummated, stockholders who do not vote "FOR" approval and adoption of the merger agreement and the merger, who hold shares of our common stock of record on the date of making a written demand for appraisal as described below, who continuously hold shares of our common stock through the closing of the merger, and who otherwise comply fully with Section 262 of the Delaware General Corporation Law (referred to as the "Delaware Law") will be entitled to a judicial determination of the fair value of their shares of common stock exclusive of any element of value arising from the accomplishment of the merger in connection with the provisions of Section 262 and to receive from us payment of such fair value in cash together with a fair rate of interest, if any, as determined by such court.

The following discussion is not a complete statement of the law pertaining to appraisal rights under Delaware Law and is qualified in its entirety by the full text of Section 262, which is provided in its entirety as Appendix C to this proxy statement. All references in Section 262 and in this summary to a "stockholder" are to the record holder of the shares of our common stock as to which appraisal rights are asserted.

A person having a beneficial interest in shares of our common stock held of record in the name of another person, such as a broker or nominee, must act promptly to cause the record holder to timely follow the steps required by Delaware Law to perfect appraisal rights.

Under Section 262, where a proposed merger is to be submitted for approval at a meeting of stockholders, as in the case of the special meeting, the corporation, not less than 20 days before the meeting, must notify each of its stockholders entitled to appraisal rights that appraisal rights are available and include in that notice a copy of Section 262. This proxy statement constitutes that notice to the holders of our common stock, and the applicable statutory provisions of Delaware Law are attached to this proxy statement as Appendix C. Any stockholder who wishes to exercise appraisal rights or who wishes to preserve that right should review carefully the following discussion and Appendix C to this proxy statement. Moreover, because of the complexity of the procedures for exercising the right to seek appraisal of the common stock, we believe that stockholders who consider exercising such appraisal rights should seek the advice of counsel, which counsel or other appraisal services will not be paid for by us. Failure to comply with the procedures specified in Section 262 timely and properly will result in the loss of appraisal rights.

Filing written objection. Any holder of our common stock wishing to exercise the right to demand appraisal under Section 262 of the Delaware Law must satisfy each of the following conditions:

As more fully described below, the holder must deliver to us a written demand for appraisal of the holder's shares before the vote on the merger agreement and the merger at the special meeting, which demand must reasonably inform us of the identity of the holder and that the holder intends to demand appraisal of the holder's shares.

The holder must not vote the holder's shares of common stock in favor of the merger agreement and the merger at the special meeting nor consent thereto in writing pursuant to Section 228 of the Delaware Law. A stockholder who submits a proxy and wishes to exercise appraisal rights must vote against the merger agreement and the merger or abstain from voting on the merger agreement and the merger, because a proxy which does not contain voting instructions will, unless revoked, be voted in favor of the merger agreement and the merger.

The holder must continuously hold the shares from the date of making the demand through the effective time of the merger. A stockholder who is the record holder of shares of common stock on the date the written demand for appraisal is made, but who thereafter transfers those shares before the effective time of the merger, will lose any right to appraisal in respect of those shares.

The written demand for appraisal must be in addition to and separate from any proxy or vote. Voting (in person or by proxy) against, abstaining from voting or failing to vote on the proposed

43

merger agreement and the merger will not constitute a written demand for appraisal within the meaning of Section 262.

Only a holder of record of shares of common stock issued and outstanding through the effective time of the merger is entitled to assert appraisal rights for the shares of common stock registered in that holder's name. A demand for appraisal should be executed by or on behalf of the stockholder of record, fully and correctly, as the stockholder's name appears on the applicable stock certificates, should specify the stockholder's name and mailing address, the number of shares of common stock owned and that the stockholder intends to demand appraisal of the stockholder's common stock. If the shares are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of the demand should be made in that capacity. If the shares are owned of record by more than one person, as in a joint tenancy or tenancy in common, the demand should be executed by or on behalf of all owners. An authorized agent, including one or more joint owners, may execute a demand for appraisal on behalf of a stockholder; however, the agent must identify the record owner or owners and expressly disclose the fact that, in executing the demand, the agent is acting as agent for such owner or owners. A record holder such as a broker who holds shares as nominee for several beneficial owners may exercise appraisal rights with respect to the shares held for one or more other beneficial owners while not exercising appraisal rights with respect to the shares held for one or more beneficial owners; in such case, the written demand should set forth the number of shares as to which appraisal is sought, and where no number of shares is expressly mentioned, the demand will be presumed to cover all shares held in the name of the record owner. If a stockholder holds shares of common stock through a broker which in turn holds the shares through a central securities depository nominee such as Cede & Co., a demand for appraisal of such shares must be made by or on behalf of the depository nominee and must identify the depository nominee as record holder.

Beneficial owners who are not record owners and who intend to exercise appraisal rights should instruct the record owner to comply strictly with the statutory requirements with respect to the delivery of written demand for appraisal. A demand for appraisal submitted by a beneficial owner who is not the record owner will not be honored.

Any stockholder who has duly demanded an appraisal in compliance with Section 262 will not, from and after the effective time of the merger, be entitled to vote or consent by written action the shares subject to that demand for any purpose or be entitled to the payment of dividends or other distributions on those shares (except dividends or other distributions payable to holders of record of shares as of a record date before the effective time of the merger).

Any stockholder may withdraw its demand for appraisal and accept \$4.00 per share by delivering to us a written withdrawal of the stockholder's demand for appraisal. However, any such attempt to withdraw made more than 60 days after the effective date of the merger will require written approval of the surviving corporation. No appraisal proceeding in the Delaware Court of Chancery will be dismissed as to any stockholder without the approval of the Court of Chancery, and such approval may be conditioned upon such terms as the Court of Chancery deems just. If the surviving corporation does not approve a stockholder's request to withdraw a demand for appraisal when that approval is required, or if the Court of Chancery does not approve the dismissal of an appraisal proceeding, the stockholder will be entitled to receive only the appraised value determined in any such appraisal proceeding, which value could be more than, the same as or less than \$4.00 per share.

A stockholder who elects to exercise appraisal rights under Section 262 should mail or deliver a written demand to Media Arts Group, 900 Lightpost Way, Morgan Hill, California 95037, Attn: Robert C. Murray, General Counsel.

Notice by Media Arts Group. If the merger agreement and the merger are approved and adopted at the special meeting, then within 10 days after the effective time of the merger, the surviving corporation must send a notice as to the effectiveness of the merger to each of our former stockholders

44

who (1) has made a written demand for appraisal in accordance with Section 262 and (2) has not voted to approve and adopt, nor consented to, the merger agreement and the merger.

Under the merger agreement, we have agreed to give Mergerco prompt notice of any demands for appraisal received by us. Mergerco has the right to participate in all negotiations and proceedings with respect to demands for appraisal. We will not, except with the prior written consent of Mergerco, make any payment with respect to any demands for appraisal, or offer to settle, or settle, any such demands.

Within 120 days after the effective time of the merger, any of our former stockholders who has demanded an appraisal and who has not withdrawn such demand in accordance with Section 262 will be entitled to receive from the surviving corporation, upon written request, a statement setting forth the aggregate number of shares not voted in favor of the merger agreement and the merger and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. The surviving corporation must mail that statement to the stockholder within 10 days of receipt of the request or within 10 days after expiration of the period for delivery of demands for appraisals under Section 262, whichever is later.

Filing a petition for appraisal. Within 120 days after the effective date of the merger, either the surviving corporation or any stockholder who has demanded an appraisal and who has not withdrawn such demand in accordance with the requirements of Section 262 may file a petition with the Court of Chancery demanding a determination of the value of the shares of common stock held by all such stockholders. We are under no obligation, and have no present intent, to file a petition for appraisal, and stockholders seeking to exercise appraisal rights should not assume that the surviving corporation will file such a petition or that it will initiate any negotiations with respect to the fair value of the shares. Accordingly, stockholders who desire to have their shares appraised should initiate any petitions necessary for the perfection of their appraisal rights within the time and the manner prescribed in Section 262. Inasmuch as we have no obligation to file such a petition, the failure of a stockholder to do so within the time specified could nullify the stockholder's previous written demand for appraisal. If, within the 120-day period following the effective time of the merger, no petition shall have been filed as provided above, all rights to appraisal will cease and all dissenting stockholders who owned shares of common stock will become entitled to receive the merger consideration for each share of common stock held, without interest.

A stockholder timely filing a petition for appraisal with the Court of Chancery must deliver a copy to the surviving corporation, which will then be obligated within 20 days to provide the Register in Chancery with a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving corporation. After notice to those stockholders, the Court of Chancery may conduct a hearing on the petition to determine which stockholders have become entitled to appraisal rights. The Court of Chancery may require stockholders who have demanded an appraisal of their shares and who hold stock represented by certificates to submit their certificates to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings. If any stockholder fails to comply with the requirement, the Court of Chancery may dismiss the proceedings as to that stockholder.

Determination of fair value. After determining the stockholders entitled to an appraisal, the Court of Chancery will appraise the shares of common stock owned by the dissenting stockholders, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value.

Stockholders considering seeking appraisal should be aware that the fair value of their shares as determined under Section 262 could be more than, the same as or less than the \$4.00 per share they would receive under the merger agreement if they did not seek appraisal of their shares. Stockholders should also be aware that investment banking opinions are not opinions as to fair value under

45

Section 262. We reserve the right to assert in any appraisal proceedings, that, for purposes of Section 262, the "fair value" of a share of common stock is less than the consideration payable pursuant to the merger agreement.

In determining fair value and, if applicable, a fair rate of interest, the Court of Chancery is to take into account all relevant factors. In Weinberger v. UOP, Inc., the Delaware Supreme Court discussed the factors that could be considered in determining fair value in an appraisal proceeding, stating that "proof of value by any techniques or methods which are generally considered acceptable in the financial community and otherwise admissible in court" should be considered and that "fair price obviously requires consideration of all relevant factors involving the value of a company." The Delaware Supreme Court stated that, in making this determination of fair value, the court must consider "market value, asset value, dividends, earnings prospects, the nature of the enterprise and any other facts which were known or which could be ascertained as of the date of the merger and which throw any light on future prospects of the merged corporation." Furthermore, the court may consider "elements of future value, including the nature of the enterprise, which are known or susceptible of proof as of the date of the merger and not the product of speculation." The Delaware Supreme Court noted that Section 262 provides that fair value is to be determined "exclusive of any element of value arising from the accomplishment or expectation of the merger."

The costs of the action may be determined by the Court of Chancery and taxed upon the parties as the Court of Chancery deems equitable. Upon application of a dissenting stockholder, the Court of Chancery may also order that all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorneys' fees and the fees and expenses of experts, be charged pro rata against the value of all of the shares entitled to appraisal.

Any stockholder wishing to exercise appraisal rights is urged to consult legal counsel before attempting to exercise appraisal rights. Failure to comply strictly with all of the procedures set forth in Section 262 of the Delaware law may result in the loss of a stockholder's statutory appraisal rights.

Regulatory Matters

We do not believe any material regulatory approvals are required to permit completion of the merger from U.S. regulatory authorities, including antitrust authorities.

The merger is subject to the HSR Act, which provides that acquisition transactions meeting the filing threshold may not be consummated unless certain information has been furnished to the Antitrust Division of the Department of Justice and the Federal Trade Commission, and certain waiting period requirements have been satisfied.

46

THE SPECIAL MEETING

General

The enclosed proxy is solicited on behalf of our board of directors for use at a special meeting of stockholders to be held on , 2003, at :00 a.m. Pacific Standard time, at , or at any adjournments or postponements thereof, for the purposes set forth in this proxy statement and in the accompanying notice of special meeting. We mailed this proxy statement and accompanying proxy card on or about , 2003 to all stockholders entitled to vote at the special meeting.

Purpose of the Special Meeting

At the special meeting, our stockholders will be asked to:

consider and vote upon a proposal to approve and adopt the merger agreement and the merger contemplated by the merger agreement.

grant discretionary authority to vote on such other matters as may properly come before the special meeting and to vote in favor of any postponements or adjournments.

We do not expect a vote to be taken on any other matters at the special meeting. However, if any other matters are properly presented at the special meeting for consideration, the holders of the proxies will have discretion to vote on these matters in accordance with their best judgment.

Record Date and Voting Information

Only holders of record of common stock at the close of business on pecial meeting. At the close of business on pecial meeting. At the close of business on period of entitled to vote. A list of our stockholders will be available for review at our executive offices during regular business hours for a period of 10 days before the special meeting. Each holder of record of common stock on the record date will be entitled to one vote for each share held. The presence, in person or by proxy, of the holders of a majority of the outstanding shares of our common stock entitled to vote at the special meeting is necessary to constitute a quorum for the transaction of business at the special meeting.

All votes will be tabulated by the inspector of election appointed for the special meeting, who will separately tabulate affirmative and negative votes, abstentions and broker non-votes. Brokers who hold shares in street name for clients typically have the authority to vote on "routine" proposals when they have not received instructions from beneficial owners. However, absent specific instructions from the beneficial owner of the shares, brokers are not allowed to exercise their voting discretion with respect to the approval and adoption of non-routine matters, such as the merger agreement and the merger; proxies submitted without a vote by the brokers on these matters are referred to as broker non-votes. Abstentions and broker non-votes are counted for purposes of determining whether a quorum exists at the special meeting.

Pursuant to our certificate of incorporation, bylaws and Delaware law, the affirmative vote of the holders of a majority of our outstanding shares of common stock is required to approve and adopt the merger agreement and the merger. For this vote, proxies that reflect abstentions and broker non-votes, as well as proxies that are not returned, will have the same effect as a vote against approval and adoption of the merger agreement and the merger.

As an additional condition, the merger agreement and the merger requires the affirmative vote of the holders of majority of the shares of our common stock that are cast either for or against approval of the merger agreement and the merger (other than shares held by Mergerco, the officers and directors of Media Arts Group and Mergerco, and the affiliated stockholders). For this vote, proxies that reflect abstentions and broker non-votes, as well as proxies that are not returned, will have no effect on approval of the merger agreement and the merger. Because shares held by Mergerco and the

47

affiliated stockholders are excluded from this vote, the affiliated stockholders do not have the ability to assure approval of the merger agreement and the merger.

Voting by Proxy

Stockholders of record can ensure that their shares are voted at the special meeting by completing, dating, signing and returning the enclosed proxy card. A return envelope (which is postage prepaid if mailed in the United Stated) is enclosed for that purpose. Submitting instructions by this method will not affect your right to attend the special meeting and vote in person after withdrawing your proxy.

Revocation of Proxies

Any person giving a proxy pursuant to this solicitation has the power to revoke it at any time before it is voted. It may be revoked by filing with the Secretary of Media Arts Group, at our executive offices located at 900 Lightpost Way, Morgan Hill, California 95037, a written notice of revocation or a duly executed proxy bearing a later date, or it may be revoked by attending the special meeting and voting in person. Attendance at the special meeting will not, by itself, revoke a proxy. Furthermore, if a stockholder's shares are held of record by a broker, bank or other nominee and the stockholder wishes to vote at the meeting, the stockholder must obtain from the record holder a proxy issued in the stockholder's name.

Expenses of Proxy Solicitation

We will bear the entire cost of solicitation of proxies, including preparation, assembly, printing and mailing of this proxy statement, the proxy card and any additional information furnished to stockholders. We have retained and its employees, at an estimated cost of approximately \$ plus reimbursement of expenses, to assist in the solicitation. Copies of solicitation materials will be furnished to banks, brokerage houses, fiduciaries and custodians holding in their names shares of common stock beneficially owned by others to forward to these beneficial owners. We may reimburse persons representing beneficial owners of common stock for their costs of forwarding solicitation materials to such beneficial owners. Original solicitation of proxies by mail may be supplemented by telephone, telegram or personal solicitation by our directors, officers or other regular employees. No additional

compensation will be paid to directors, officers or other regular employees for their services.

Adjournments

Although it is not expected, the special meeting may be adjourned for the purpose of soliciting additional proxies. Any adjournment of the special meeting may be made without notice, other than by an announcement made at the special meeting, by approval of the holders of a majority of the outstanding shares of our common stock present in person or represented by proxy at the special meeting, whether or not a quorum exists. We are soliciting proxies to grant discretionary authority to vote in favor of any postponements or adjournments of the special meeting. In particular, discretionary authority is expected to be exercised if the purpose of the postponement or adjournment is to provide additional time to solicit votes to approve and adopt the merger agreement and the merger.

Exchanging Stock Certificates

Please do not send in stock certificates at this time. In the event the merger is completed, instructions regarding the procedures for exchanging your stock certificates for the \$4.00 per share cash payment will be distributed.

48

THE MERGER AGREEMENT

The following is a summary of the material terms of the merger agreement and the merger. This summary does not purport to be complete and is qualified in its entirety by reference to the appendices to this proxy statement, including Appendix A which sets forth the full text of the merger agreement. Stockholders are urged to read the entire merger agreement.

Effective Time of the Merger

The merger will become effective upon the acceptance for filing of the certificate of merger with the Secretary of State of the State of Delaware, or at such other time or date as we and Mergerco agree to specify in such certificate, which time is referred to as the "effective time."

Conversion of Common Stock

At the effective time, each issued and outstanding share of our common stock will be converted into the right to receive \$4.00 per share in cash, without interest, referred to as the "merger consideration," except for:

shares held by us in treasury, which shall be cancelled without any payment;

shares held by Mergerco or any of the affiliated stockholders, which shall be cancelled without any payment; and

shares held by stockholders seeking appraisal rights in accordance with Delaware Law.

Payment for Shares

Prior to the effective time, Mergerco will irrevocably deposit or cause to be deposited with the paying agent sufficient funds to pay the aggregate merger consideration and required cash payments to holders of certain outstanding stock options and warrants, as described in " Treatment of Stock Options and Warrants." Promptly after the effective time, the surviving corporation will cause the paying agent to mail to each holder of record of shares of our common stock immediately prior to the effective time a form of letter of transmittal and instructions to effect the surrender of their share certificate(s) in exchange for payment of the merger consideration.

Our stockholders (other than Mergerco and the affiliated stockholders) will be entitled to receive \$4.00 in cash for each share of our common stock they hold only upon surrender to the paying agent of a share certificate, together with such letter of transmittal, duly completed in accordance with the instructions thereto. If payment of the merger consideration is to be made to a person whose name is other than that of the person in whose name the share certificate is registered, it will be a condition of payment that (1) the paying agent

receive all documents required to evidence and effect such transfer and (2) the paying agent receives evidence that any applicable stock transfer taxes have been paid. No interest on any merger consideration will be paid or accrued upon the surrender of the share certificates for the benefit of holders of the share certificates.

STOCKHOLDERS SHOULD NOT FORWARD THEIR STOCK CERTIFICATES TO THE PAYING AGENT WITHOUT A LETTER OF TRANSMITTAL AND SHOULD NOT RETURN THEIR STOCK CERTIFICATES WITH THE ENCLOSED PROXY.

Transfer of Shares

After the effective time, there will be no further registration of transfers on our records or by our transfer agent of certificates representing shares of our common stock and any such certificates presented to the surviving corporation for transfer, other than shares held by stockholders seeking appraisal rights, will be cancelled. From and after the effective time, the holders of share certificates

49

will cease to have any rights with respect to these shares except as otherwise provided for in the merger agreement or by applicable law.

Treatment of Stock Options and Warrants

At the effective time, each outstanding option or warrant to purchase our stock (whether or not vested) (other than those held by Mr. Kinkade or his affiliates for which no payment will be made) will be cancelled and converted into the right to receive (net of any applicable withholding taxes) a cash payment equal to the \$4.00 merger consideration per share, minus the per share exercise price of the option or warrant, multiplied by the number of shares of stock that are issuable upon the exercise of the option or warrant.

As of November 7, 2003, outstanding stock options and warrants to purchase 1,087,267 shares of our common stock have exercise prices less than the \$4.00 per share merger consideration amount. The remaining outstanding stock options and warrants have exercise prices that are greater than the merger consideration amount and therefore are unlikely to be exercised.

Representations and Warranties

The merger agreement contains various representations and warranties made by us to Mergerco, including representations and warranties relating to:

our due organization, valid existence and good standing;
our requisite corporate power and authorization to enter into, and the enforceability of, the merger agreement;
the correctness and completeness of our certificate of incorporation and bylaws as provided to Mergerco;
our capitalization;
required consents and approvals;
certain actions by our board of directors;
the inapplicability of certain state takeover laws;

the requisite vote required to effect the merger;

the receipt of a fairness opinion from Jefferies;

the absence of certain finders' fees and brokerage commissions;

the exemption of the transactions contemplated by the merger agreement from Section 16(b) liability under the Exchange Act.

The merger agreement also contains various representations and warranties by Mergerco and Mr. Kinkade to us, including representations and warranties relating to:

Mergerco's due organization, valid existence and good standing;

Mergerco's requisite corporate power and authorization to enter into, and the enforceability of, the merger agreement against Mergerco and Mr. Kinkade;

required consents and approvals;

the absence of certain finders' fees and brokerage commissions;

the receipt of a commitment letter regarding financing;

50

the limited purpose and operation of Mergerco;

the absence of other representations and warranties made by us or our directors, officers, employees, agents and other representatives regarding information provided to Mergerco or Mr. Kinkade.

None of the representations and warranties made by us, Mergerco or Mr. Kinkade in the merger agreement will survive after the completion of the merger.

Conduct of Business Pending the Merger

We are subject to restrictions on our conduct and operations until the merger is completed. In the merger agreement, we have agreed, with limited exceptions, that we will not do any of the following, except as expressly contemplated by the merger agreement or otherwise consented to in writing by Mergerco:

amend or otherwise change our certificate of incorporation or bylaws or other organizational documents;

issue, sell or grant, or authorize the issuance, sale or grant of any shares of our capital stock or any options, warrants, convertible securities or other rights of any kind to acquire any shares of our capital stock, or any other ownership interest, except for the issuance of shares of our common stock issued pursuant to the exercise of options and warrants outstanding on the date of the merger agreement or pursuant to our employee stock purchase plan;

declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of our capital stock, other than dividends and distributions by a subsidiary to us in accordance with applicable law;

reclassify, combine, split, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of our capital stock;

acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the stock or assets of, or by any other manner, any business or any corporation, partnership, joint venture, association or other business organization which would be material to us and our subsidiaries, taken as a whole;

incur any indebtedness for borrowed money or guarantee any such indebtedness of another person, issue or sell any debt securities or warrants or other rights to acquire any debt securities or Media Arts Group or its subsidiaries, guarantee any debt securities of another person, enter into any "keep well" or other agreement to maintain any financial statement condition of another person or enter into any arrangement having the economic effect of any of the foregoing, with limited exceptions;

make any loans, advances or capital contributions to, or investments in, any other person, with limited exceptions;

adopt resolutions providing for or authorizing a liquidation or a dissolution, with limited exceptions;

take, or agree to commit to take, or fail to take any action that would make any of our representations or warranties contained in the merger agreement inaccurate such that a condition to the obligations of Mergerco under the merger agreement will not be satisfied at, or as of any time prior to, the effective time; or

51

enter into, or publicly announce an intention to enter into, any contract, agreement, commitment, plan or arrangement to, or otherwise agree or consent to do any of the foregoing actions.

Limitations on Considering Other Acquisition Proposals

We have agreed that, except as described below, until the effective time or the termination of the merger agreement, neither we, nor any of our subsidiaries, officers, directors, employees, representatives, agents or affiliates will:

directly or indirectly, solicit, initiate, knowingly encourage or facilitate the making of an acquisition proposal (as defined below);

engage in or knowingly encourage in any way negotiations or discussions concerning, or provide any non-public information to, any third party (a person other than Mergerco) relating to an acquisition proposal, or which may reasonably be expected to lead to an acquisition proposal; or

agree to, recommend or endorse any acquisition proposal.

Under the merger agreement, an acquisition proposal means any offer or proposal for:

a transaction or series of related transactions pursuant to which any third party would acquire 25% or more of the outstanding shares of our common stock or voting power, including without limitation a tender offer or an exchange

offer which, if consummated, would result in a third party acquiring 25% or more of the outstanding shares of our common stock or voting power;

a merger or other business combination involving the Company pursuant to which any third party would acquire securities representing 25% or more of the voting power or the outstanding securities of the company surviving the merger or business combination; or

any other transaction pursuant to which any third party would acquire control of assets comprising at least 25% or more the book value of our assets.

Prior to the date on which our stockholders approve and adopt the merger agreement, we may furnish non-public information to or enter into discussions or negotiations with any third party that makes an unsolicited, bona fide acquisition proposal if:

our board of directors determines in good faith, after consultation with its legal counsel, that the failure to furnish such information or to participate in such negotiations or discussions with respect to such an acquisition proposal would be inconsistent with the fiduciary duties of our board of directors;

prior to the disclosure of any non-public information or entering into discussions or negotiations, such third party enters into a confidentiality agreement with us on customary terms and conditions; and

prior to disclosing any non-public information or entering into discussions or negotiations, we notify Mergerco at least two business days in advance and we keep Mergerco reasonably informed of the status and material terms and conditions of such discussions or negotiations.

Our board of directors may withdraw, modify or change its recommendation that our stockholders approve and adopt the merger agreement if we receive a superior proposal and our board of directors determines in good faith, after consultation with its legal counsel, that the failure to withdraw, modify or change its recommendation would be inconsistent with the fiduciary duties of our board of directors.

We may terminate the merger agreement if we receive and accept a superior proposal from a third party, as discussed in "Termination."

52

Under the merger agreement, a superior proposal means an unsolicited, bona fide offer made by a third party to acquire a majority of the outstanding shares of our common stock or substantially all of our assets on terms that our board of directors determines in good faith, after consultation with its financial advisor and outside counsel, to be more favorable to our stockholders (other than Mergerco, Mr. Kinkade and his affiliates) than the transactions contemplated by the merger agreement, taking into account all factors it deems relevant (including whether, in the good faith judgment of our board of directors, such third party is able to finance the transaction and obtain all required regulatory approvals).

Other Covenants Regarding the Merger

In addition to the covenants regarding the conduct of business and limitations on our ability to consider other acquisition proposals, the merger agreement contains other covenants that we, Mergerco and Mr. Kinkade must comply with, including covenants relating to:

the conduct of business of Mergerco pending the merger;

the preparation and filing of this proxy statement and a transaction statement under Section 13(e) of the Exchange Act;

access to our non-public information;

obtaining all necessary approvals and consents to the merger;

our employee stock purchase plan;

any take-over statutes that may become applicable to the merger;

the financing of the merger; and

the maintenance of directors' and officers' liability insurance for our current officers and current members of our board of directors.

Conditions to Completing the Merger

Conditions to each party's obligation. The obligations of Media Arts Group, Mergerco and Mr. Kinkade to complete the merger are subject to the satisfaction or waiver of certain conditions, including the following:

the affirmative vote of the holders of a majority of the outstanding shares of our common stock in favor of the merger agreement and the merger;

the affirmative vote of the holders of a majority the shares that are cast at the special meeting either for or against the merger agreement and the merger (other than shares held by Mergerco, the officers and directors of Media Arts Group and Mergerco, and the affiliated stockholders) in favor of the merger agreement and the merger;

there must not be in effect any law, rule, regulation or order that would make the merger illegal or otherwise prohibit the consummation of the merger;

all consents, approvals and authorizations required to be obtained to consummate the merger must have been obtained, except for such consents, approvals and authorizations the failure of which to obtain could not reasonably be expected to have a material adverse effect on us; and

any waiting period applicable to the merger under the HSR Act must have terminated or expired.

53

Conditions to the obligation of Mergerco. The obligation of Mergerco to complete the merger is subject to the satisfaction or waiver of the following conditions:

our representations and warranties in the merger agreement must be true and correct as of the closing date, except where the failure to be true and correct has not had a material adverse effect on us;

we must have performed and complied in all material respects with all agreements, conditions and covenants required by the merger agreement on or prior to the closing date;

there must not have occurred or come into existence any change, event, occurrence, state of facts or development that has had a material adverse effect on us; and

sufficient funds must be available to Mergerco in order to complete the merger and pay the merger consideration and perform its other obligations under the merger agreement.

Under the merger agreement, a material adverse effect with respect to us is defined as any effect, change, event, circumstance or condition that, individually or in the aggregate has a material adverse effect on our business operations, assets, properties, results of operations, or financial condition, subject to the following exceptions:

changes in general economic conditions, nationally or regionally;

changes in conditions (including as a result of changes in laws) affecting the industries in which we compete, provided that such changes do not disproportionately affect us;

actions taken or omitted by us required pursuant to the merger agreement or upon the request of Mergerco or with the consent of Mergerco after the date of the merger agreement;

the announcement or pendency of the merger agreement and the transactions contemplated by it, including, without limitation, voluntary departure of employees or downgrading of credit ratings;

changes in our public stock price, by itself;

failure by us to meet or exceed any analyst's revenue or earnings projections or forecasts, in any case whether or not published or otherwise publicly available,

failure by us to meet internal revenue or earnings expectations, by itself, but not effects, changes, events, circumstances or conditions other than the failure to meet expectations that would otherwise constitute or be considered in determining whether a material adverse effect has occurred; and

any facts or circumstances known by Mr. Kinkade or any of his affiliates to exist as of the date of the merger agreement.

Conditions to Media Arts Group's obligation. Our obligation to complete the merger is subject to the satisfaction or waiver of the following conditions:

the representations and warranties of Mergerco contained in the merger agreement must be true and correct in all material respects as of the closing date; and

Mergerco must have complied in all material respects with all agreements, covenants and conditions required by the merger agreement on or prior to the closing date.

We are not aware of any material fact, event or circumstance that would prevent any of the conditions to closing discussed above from being satisfied. Media Arts Group, Mergerco and Mr. Kinkade do not anticipate that any of the closing conditions will be waived. However, in the event any of the closing conditions are waived after the special meeting, we do not anticipate that we will

re-solicit proxies. The conditions identified at " Conditions to Completing the Merger Conditions to each party's obligations," above, may not be waived.

Termination

We and Mergerco may agree by mutual written consent to terminate the merger agreement at any time before the effective time. In addition, the merger agreement may be terminated:

by either party, if the merger is not completed on or before January 31, 2003, unless the non-completion is the proximate result of a breach in any material aspect of the obligations under the merger agreement of the party wishing to terminate, or in certain limited circumstances on or before March 1, 2003;

by either party, if a court of competent jurisdiction or an administrative, governmental, or regulatory body or authority has issued a final nonappealable order, decree or ruling, or taken any other action, having the effect of permanently restraining, enjoining or otherwise prohibiting the merger;

by either party, if our stockholders do not approve the merger agreement and the merger, as contemplated by the required voting terms described at " Conditions to Completing the Merger," above, unless the failure to obtain stockholder approval is the proximate result of the failure to perform under the merger agreement by the party wishing to terminate.

by Mergerco, if (i) we have breached our non-solicitation obligations under the merger agreement, or (ii) our board of directors has recommended to our stockholders any other acquisition proposal or have resolved or announced an intention to do so, or (iii) our board of directors has withdrawn or modified in a manner adverse to Mergerco its approval or recommendation of the merger, or (iv) a tender offer or exchange offer for 25% or more of the outstanding shares of our common stock is announced or commenced, and our board of directors recommends acceptance of such tender offer or exchange offer by our stockholders

by Mergerco, if (i) Mergerco is not in material breach of its obligations under the merger agreement and (ii) there has been a material breach by us of any of our representations, warranties or obligations under the merger agreement such that the conditions in the merger agreement will not be satisfied; provided, however, that if such a breach is curable by us and such cure is reasonably likely to be accomplished prior to the applicable date specified in the merger agreement, then, for so long as we continue to exercise commercially reasonable efforts to accomplish such cure, Mergerco may not terminate this merger agreement under this provision;

by us, if (i) we are not in material breach of our obligations under the merger agreement and (ii) there has been a material breach by Mergerco or Mr. Kinkade of any of its or his representations, warranties or obligations under the merger agreement such that the conditions in the merger agreement will not be satisfied; provided, however, that if such a breach is curable by Mergerco or Mr. Kinkade and such cure is reasonably likely to be accomplished prior to the applicable date specified in the merger agreement, then, for so long as Mergerco or Mr. Kinkade continues to exercise commercially reasonable efforts to accomplish such cure, we may not terminate the merger agreement under this provision; or

by us, if prior to approval of the merger by our stockholders and as a result of a superior proposal, our board of directors determines in its good faith judgment after consultation with its legal counsel and financial advisor, that the failure to terminate the merger agreement and

55

accept such superior proposal would be inconsistent with the fiduciary duties of our board of directors and we:

provide notice of the proposed termination to Mergerco; and

give Mergerco five days to make an offer that is at least as favorable to our stockholders as the superior proposal and negotiate in good faith with Mergerco regarding any revised offer.

Subject to limited exceptions, including the survival of any obligations to pay the termination expenses, if the merger agreement is terminated it will become void and will be of no further effect with no liability on the part of us, Mergerco or Mr. Kinkade. However, subject to the provisions in the merger agreement concerning reimbursement of expenses as liquidated damages, no party will be relieved from its obligations with respect to any material breach of the merger agreement.

Fee and Expense Reimbursement

Generally, the merger agreement provides that all fees and expenses will be paid by the party incurring them, except as described below

We have agreed to reimburse Mergerco and Mr. Kinkade for up to \$1.0 million of out-of-pocket fees and expenses actually and reasonably incurred in connection with the merger under circumstances where:

Mergerco terminates the merger agreement as a result of:

our breaching our obligation not to solicit or support acquisition proposals; or

our breaching our representations, warranties or obligations under the merger agreement and failing to cure such breach; or

we terminate the merger agreement because our board of directors has approved a superior proposal.

Mr. Kinkade and Mergerco have agreed to reimburse us for up to \$1.0 million of out-of-pocket fees and expenses actually and reasonably incurred in connection with the merger under circumstances where:

we terminate the merger agreement as a result of Mergerco or Mr. Kinkade breaching their respective representations, warranties or obligations under the merger agreement and failing to cure such breach; or

the closing condition to the merger that sufficient funds be available to Mergerco in order to complete the merger and pay the merger consideration is not satisfied primarily as a result of the failure by Mr. Kinkade to provide the security contemplated by the financing commitment letter or the failure of Mergerco and/or us to issue subordinated debt as contemplated by such commitment letter.

56

INFORMATION RELATING TO MEDIA ARTS GROUP, INC.

General

We are the designer, manufacturer, marketer, licensor, and distributor of fine art reproductions, based upon the original paintings of Thomas Kinkade. We believe that the message of Thomas Kinkade as a "life style" brand has far reaching consumer appeal, in that his

message celebrates a sense of home, family, nature and traditions, and our products help create a positive environment, in which to live and work.

Our core manufactured products include premium canvas, archival and open edition reproductions. Reproductions may be purchased as framed or unframed products. We also license our artwork and trademarks. Licensed products include stationary, calendars, ornaments, gifts, collectables, home accent and home décor products.

We serve a wide consumer base by offering our manufactured and licensed products through a number of channels, including an extensive network of independently owned and operated Thomas Kinkade Signature Galleries, Showcase Galleries and other independent dealers. Our breadth of product offerings allows our dealers to offer products at a wide range of retail price points.

In addition, we have established strategic business relationships with other companies including QVC, Avon, the Bradford Exchange, Hallmark Cards Inc., Barth & Dreyfuss, BloomCraft, Amcal, Harvest House, Bullfinch, Home Interiors, Teleflora, Thomas Nelson publishing and Madacy Entertainment and other key licensing, manufacturing, distribution and publishing partners. Our distribution network has helped to promote Thomas Kinkade as a "life style" brand.

Our marketing strategy is to support our dealers and other business partners with marketing campaigns and special consumer promotions to enhance Thomas Kinkade "life style" brand awareness, generate consumer excitement and increase traffic into authorized dealer locations. During 2002, we shifted our marketing program from an image based strategy to a branded theme based strategy which enables us and our licensing partners to concurrently develop complimentary gift and collectable products based on the Thomas Kinkade brand. During 2002, we completed our restructuring and alignment of sales and marketing to focus our resources exclusively on the Thomas Kinkade "life style" brand.

Products

Our product offerings include framed and unframed limited and open edition reproductions of original artwork produced by Thomas Kinkade, as well as licensed products including books, stationery items and other home accessories and gift products that feature Mr. Kinkade's unique use of light and his peaceful, warm and inspiring themes.

Product Categories. Art reproductions are generally categorized as limited or open edition products. Limited editions are high quality canvas and archival reproductions produced in limited quantities, each of which is accompanied by a certificate of authenticity stating the size of the edition and the number of the print. Open editions are products that are produced in unlimited quantities and sold indefinitely. For the year ended December 31, 2002 and the nine months ended September 30, 2003, limited editions represented approximately 56% and 46% of our revenue respectively, and open editions represented approximately 24% of our net revenue.

Licensed Products. We license our artwork and trademarks to licensees that create, manufacture, market, distribute and sell products that feature the artwork and/or name of Thomas Kinkade. In some instances, we work directly with a licensee to create the product, and in other instances the licensee will create the product independently. In either case, we have approval rights over all licensed products

57

utilizing the artwork or trademarks. Although these products are not manufactured or sold by us, they are included as our products for purposes of this discussion.

Creative Process. The majority of offered products are based on the artwork of Thomas Kinkade, a winner of multiple awards from the National Association of Limited Edition Dealers, including Artist of the Year and Graphic Artist of the Year. Thomas Kinkade paints in his Northern California studio and on location while traveling. Mr. Kinkade is known for his unique use of light and the manner in which his paintings reflect changes in the intensity of the ambient lighting and provide a warm and inviting life style message to the customer.

Media Arts Group and Mr. Kinkade operate under a license agreement dated December 3, 1997. Under the terms of the license agreement, we have the complete, unencumbered, exclusive and perpetual rights to reproduce, adapt, manufacture, sub-license, publish, market, distribute, sell and display all art-based and non-art-based products and services associated with Mr. Kinkade and his "artwork." The "artwork" includes at least 150 paintings that Mr. Kinkade is required to provide to us between December 1997 and December 2012 (with at least 10 paintings to be delivered during each of the first five years), and all original sketches, drawings, writings, paintings and other works of art created by Mr. Kinkade prior to December 1997 (including "archive" images) and during the term of the license agreement. The license agreement also grants us the right to use the name and likeness of Mr. Kinkade in promoting the sale of our

products and development of any brand name associated with Mr. Kinkade. Our creative services department collaborates with Mr. Kinkade, dealers, the sales department and strategic business partners to assist in the creation and development of new products. Proposed new products are tested and pre-marketed prior to product introductions. Mr. Kinkade has the right to approve our products based upon his artwork, as well as promotional materials, business plans and strategic relationships relating to such products or the use of his name or likeness. Mr. Kinkade retains ownership of the original paintings he produces.

Distribution

As of September 30, 2003, our manufactured products are sold at retail by approximately 215 independently owned Thomas Kinkade Signature Galleries, five Media Arts Group owned stores and a network of approximately 3,500 other independent art dealers. We seek to strengthen our existing art-based brands and increase sales by expanding the network of Thomas Kinkade Signature Galleries and other independent galleries, expanding distribution through strategic business relationships, and expanding our popular licensed products.

Thomas Kinkade Signature Galleries. The independently owned and operated Thomas Kinkade Signature Galleries and Media Arts Group owned Thomas Kinkade Stores provide warm and inviting environments that convey Thomas Kinkade's "life style" brand message and display Thomas Kinkade reproductions and other products as they might appear in a customer's home. In 1996, the Signature Gallery program was initiated to develop a network of stores owned and operated by individual entrepreneurs selling only Thomas Kinkade products. We believe that the Signature Gallery program avails itself of regional knowledge of local Signature Gallery owners to strengthen the Thomas Kinkade brand and broaden the dealer network, with limited investment. In 1999, we began to expand Signature Galleries into international markets. At September 30, 2003, there were four Signature Galleries operating in Europe (three in England and one in Scotland) and one Signature Gallery in Toronto, Canada. As of September 30, 2003, 215 Signature Galleries were in operation worldwide, representing approximately \$20 million in revenue for the nine months ended September 30, 2003. We intend to carefully expand the Thomas Kinkade Signature Gallery program into communities not currently served by Signature Galleries.

In order to be granted a license to use the Thomas Kinkade Signature Gallery trademark and carry our manufactured product, prospective gallery owners must satisfy certain financial criteria

58

including minimum start-up capital and net worth requirements, and demonstrate related business experience. Signature Gallery owners agree to display a broad collection of Thomas Kinkade images and have limited use of the Thomas Kinkade name. Through the qualification and selection process, Media Arts Group attempts to select dealers with the highest probability of long term success.

During the fiscal year ended March 31, 2000, we made the strategic decision to limit direct ownership and operation of Media Arts Group owned Thomas Kinkade Stores, in favor of increasing the focus on the Signature Gallery program. As of September 30, 2003, Media Arts Group operated four Media Arts Group owned Thomas Kinkade Stores and one other Media Arts Group owned store.

Other Independent Dealers. In addition to sales to Signature Galleries, our manufactured products are also sold to approximately 3,500 other independent dealers, including independent gift and collectible retailers, art galleries and frame stores located principally in the United States and, to a lesser extent, in Canada. Dealers are separated into five dealer levels, with each level operating under different criteria. Dealer levels range from Open Edition Accounts, which are authorized to purchase only open edition products and can be opened with an initial minimum purchase of \$500, to Showcase Galleries, which are stores-within-stores, that are committed to purchase a minimum of \$30,000 annually in limited edition canvas and paper products. Dealers can be elevated to a higher dealer level by increasing sales of Thomas Kinkade products. Likewise, dealers who do not meet annual revenue targets may be positioned at a lower dealer level. As of September 30, 2003, there were approximately 348 Showcase Galleries, 478 Premier Dealers, 1,137 Authorized and Certified Dealers, and 1,526 Open Edition Accounts.

Distribution of Licensed Products. Products created and manufactured under a license with Media Arts Group are distributed through the licensee's distribution channels, which may include department stores, mass merchandisers, gift shops, bookstores, specialty stores, and direct mail catalogs, as well as our dealer network.

Strategic Business Relationships

We have entered into agreements with leading consumer marketing companies to build "life style" brand awareness, generate additional sales by reaching a larger audience of consumers and leverage the expertise of these companies in sales and marketing, manufacturing and distribution. Thomas Kinkade products are also sold through direct marketing campaigns, through our licensing and manufacturing partners, on QVC and through third party direct mail catalogs. Reproductions, gift prints and other home accessory/decor

products were featured on QVC shows generating sales to QVC of approximately \$6.0 million in the year ended December 31, 2002 and \$1.7 million for the nine months ended September 30, 2003. In fiscal 2002, we also sold open edition products to Home Interiors and Park West Galleries, Inc., generating approximately \$4.8 million and \$5.8 million, respectively. For the nine months ended September 30, 2003, we sold \$3.2 million open edition products to Home Interiors. We intend to continue to develop strategic business relationships with leading consumer marketing companies throughout the world, to provide consumers with greater access to our products.

The investment in advertising and marketing of licensed Thomas Kinkade products and merchandise by licensees enhances the brand's overall reach. In 2003, some of the licensees that have conducted advertising and marketing of Thomas Kinkade products include: Avon, Bradford Exchange, Hallmark Cards Inc., Barth & Dreyfuss, BloomCraft and Amcal. We intend to continue to develop strategic relationships with licensees that will undertake such advertising and marketing campaigns.

Creative Services

During the fourth quarter of 2002, we established the creative services department to direct, manage and oversee all of our product development activities to extend the Thomas Kinkade "life style" brand into new markets. The charter of the creative services unit is to "art direct" new product

59

offerings including concept viability, design, testing, development and marketing. The business unit is responsible for the development of creative architectures to facilitate the development of new products by us, our licensees and manufacturing alliance partners. Creative architectures include style guides, color palettes, and the development of specific product categories.

Marketing Programs. We utilize multi-media marketing programs including print, radio and television to enhance our marketing capabilities to consumers and collectors. In addition, Media Arts Group designs and sells promotional materials, including postcards, catalogs, videotapes, add slicks, wall talkers and other collateral to Signature Galleries and authorized independent dealers, and to our distribution partners. A print advertising allowance program is available to Signature Galleries on a co-operative basis. Signature Galleries and other high level independent dealers may also participate in regional events (including artist appearances) organized by us. In many instances, advertising is also provided by our strategic business partners, such as QVC, Bradford Exchange and others. We develop direct marketing campaigns to reach consumers and collectors through our marketing database.

Sales and Business Development

Our sales and marketing activities include a sales force that sells to and services dealer accounts, and provides training assistance for retail dealer sales personnel, marketing and promotional programs.

During 2002, we completed a reorganization of our sales and customer care organizations to focus attention on existing profitable products and markets, while at the same time improve customer service, assist individual dealers with promotion and create an effective marketing strategy through improved media campaigns which are designed to increase customer traffic into authorized dealers and partner stores.

We intend to expand our product-appropriate distribution concept (*i.e.*, certain products will not be available in all channels). This will serve to better differentiate between our market channels and target specific products for specific markets. Strategically, we will continue to position the Signature Galleries as the flagships of the Thomas Kinkade brand.

Expansion of markets will be pursued by the sales organization, with emphasis on expansion of distribution channels within North America, and, to a limited extent, in Western Europe.

Sales Force. Our sales force includes regional gallery account managers, a customer care team, business development managers, key account managers, and licensing personnel. The sales force is generally compensated on a salary plus incentive basis.

As part of the reorganization of our sales and marketing organizations, during 2002 we closed our Retail Development Team's operations in Monterey, California. The team had been responsible for expansion of Signature Galleries and training of the owners. The team's functions were integrated into our sales organization. Exclusive consultative sales personnel will work with individual Signature Galleries to provide training programs and product support for gallery owners. Our prior training program, Thomas Kinkade University, was closed in 2001.

Thomas Kinkade Collectors' Society. Media Arts Group sponsors and operates a collectors club for consumers of Thomas Kinkade products. As of September 30, 2003, the Thomas Kinkade Collectors' Society had approximately 21,100 members, with an annual membership fee of \$50. Membership in the Society includes quarterly newsletters and current information about Mr. Kinkade's artwork, including upcoming releases and events, as well as the opportunity to purchase Collectors Society only product offerings.

Charitable Activities. As part of our corporate citizen mission, during 2002, we developed and strengthened key strategic relationships with charitable organizations. During 2002, we donated product (mostly canvas and reproductions) with a retail value of \$325,000 to 338 different charities. No one

60

item exceeded \$3,500 in retail value. We also made a cash donation of \$9,000 to the American Heart Association, as part of an employer matching contribution program for the Heart Walkathon.

Manufacturing and Production

Limited edition canvas and archival reproductions are manufactured, assembled, warehoused and shipped from our production facility in Morgan Hill, California. Some open edition reproductions are manufactured, assembled and shipped from our manufacturing alliance partners. Three-dimensional products and gift items are manufactured by third parties under separate manufacturing or licensing arrangements.

Our proprietary manufacturing process for a canvas reproduction begins with an original painting. The painting is first digitally photographed to capture the image for color separation and proofing, and approval by the artist. Our product development team develops a reproduction that best represents the image of the original painting. The reproductions are printed and sent to the manufacturing facility, where they are sent through a double authentication signing process, inspected, transferred to canvas and hand-highlighted by one of our trained highlighters.

The manufacturing cycle takes approximately five days to complete. Finished canvas reproductions are shipped either framed or unframed, in accordance with customer order specifications. Third party vendors supply the frames, paper, canvas, paint and other raw materials and components used in the canvas reproduction production process. We are committed to assuring that products meet rigid quality standards and continually evaluate our manufacturing processes and supplier relations to maintain quality control. Systematic quality control procedures, including spot inspections and regular inspection check stations, are in place at various points in the reproduction process. In order to improve quality control, shorten production time and increase capacity, we may automate certain portions of the production process, invest in packaging, conveyance and MIS equipment, vertically integrate certain manufacturing processes and develop further improvements in the reproduction process or outsource some or all of our processes.

Backlog

Because products are generally shipped within a short period after receipt of an order, we have a limited backlog of unfilled orders. As a result, revenue in any quarter is substantially dependent on orders booked and shipped in that quarter.

Employees

As of September 30, 2003, we employed 286 employees, including 140 in manufacturing and distribution, 95 in sales and marketing and 51 in corporate administration. We believe that labor relations are satisfactory and has never experienced a work stoppage.

Customers

In the nine months ended September 30, 2003, as well as the fiscal year ended December 31, 2002, the transition period ended December 31, 2001, the fiscal year ended March 31, 2001, and the fiscal year ended March 31, 2000, no single customer accounted for more than 10% of revenues.

Properties

Our manufacturing, distribution, sales and marketing, administration and executive offices are in three leased campus facilities in Morgan Hill, California encompassing approximately 343,000 square feet. As of September 30, 2003, Media Arts Group directly operated

retail operations located in five leased sites in the United States ranging from 900 to 1,500 square feet, with an aggregate of

61

approximately 5,000 square feet. We believe that our existing facilities are adequate for our present and future needs.

Legal Proceedings

From time to time, we are involved in various legal proceedings arising from the normal course of our business activities. Included among these various legal proceedings are lawsuits, claims and other proceedings against us and our affiliates by dealers and gallery owners. These dealer and gallery owner matters generally arise out of contractual, dealer and other relationship claims or demands for rescission, damages or equitable relief. Generally, we also have claims against these dealers or gallery owners, primarily for non-payment of trade accounts payable to us incurred by the dealer or gallery owner from the purchase of reproductions and other products. We regularly evaluate the expected outcome of these litigation matters. Any adverse outcome from these matters is currently not expected to have a material adverse impact on the results of operations, cash flows or our financial position, either individually or in the aggregate. However, our evaluation of these pending disputes could change in the future. In addition, although we do not currently expect any adverse outcomes in these litigation matters to have a material adverse impact, due to the inherently uncertain nature of litigation, we may be unable to successfully defend ourselves in these litigation matters, and our results of operations, cash flows or financial position could be materially adversely affected as a result.

On November 6, 2003, a putative class action complaint was filed in the Superior Court of the State of California for the County of Santa Clara, by a plaintiff on behalf of himself and all others similarly situated. The complaint named us and our directors, including Mr. Kinkade, as defendants. As of November 9, 2003, we have not been served with the complaint.

62

SELECTED HISTORICAL FINANCIAL DATA

The selected financial data set forth below as of the fiscal year ended December 31, 2002, the transition period ended December 31, 2001, and the fiscal years ended March 31, 2001, 2000 and 1999 have been derived from our consolidated financial statements, which are incorporated by reference in this proxy statement. The selected consolidated financial data set forth below as of and for the nine months ended September 30, 2003 and 2002 have been derived from our unaudited condensed consolidated financial statements, incorporated by reference in this proxy statement. Such unaudited condensed consolidated financial statements have been prepared by us on a basis consistent with our annual audited consolidated financial statements and, in the opinion of our management, contain all normal recurring adjustments necessary for a fair presentation of the consolidated financial position and the results of operations for the applicable periods. Operating results for the nine months ended September 30, 2003 are not necessarily indicative of the results that may be expected for the fiscal year ending December 31, 2003. We have not provided any pro forma data giving effect to the merger as we do not believe that such information is material to our stockholders in evaluating the merger and the merger agreement. The merger consideration consists solely of cash and, if the merger is consummated, our common stock will cease to be publicly traded. As a result, we do not believe that the changes to Media Arts Group's financial condition resulting from the merger would provide meaningful or relevant information in evaluating the merger and the merger agreement since our stockholders (other than Mr. Kinkade and his affiliates) will not be stockholders of, and will have no interest in, Media Arts Group following the merger.

The information set forth below should be read in conjunction with our consolidated financial statements and notes thereto and Management's discussion and analysis of financial condition and results of operations included in our periodic reports incorporated by reference elsewhere in this proxy statement.

Nine months ended September 30,		Year ended December 31,	Nine months ended December 31,	Year ended March 31,				
2003	2002	2002	2001	2001	2000	1999		

	e	Nine mo		Year ended December 31,		Nine months ended December 31,	Yea	ar ende	ed Marcl	131,
	_			(In thousa	ands,	except per sha	re data)	_		
tatement of Operations Data:	ф	44.540.4	D 54.532	Φ 102.5	7 00 d	00.004	Ф. 122.00		1.41.065	ф. 120.20.
Net revenues	\$				780 \$				141,065	
Gross margin		20,148	34,593	49,5		34,456	76,436		87,918	84,26
Total operating expenses		27,605	39,084	51,8		52,962	62,753		64,805	55,033
Operating income (loss)		(7,457)	(4,491)		246)	(18,506)	•		23,113	29,228
Income (loss) from continuing operation Discontinued operations tax benefit fro closure of subsidiary		(4,257)	(2,546)	(1,.	116)	(12,750) 1,869	8,904	ŀ	14,150	18,352
Net income (loss)	\$	(3,915) 5	\$ (2,546)	\$ (1,1)	116) \$	·	\$ 8,904	\$	14,150	\$ 18,35
Basic and diluted net income (loss) per s Net income (loss) per share from continuing operations:	share:									
Basic	\$	(0.32) 5	\$ (0.19)	\$ (0	.08) \$	(0.97)	\$ 0.68	3 \$	1.09	\$ 1.4
Diluted Net income per share from discontinuoperations:	\$ ued	(0.32) \$	\$ (0.19)	\$ (0	.08) \$	(0.97)	\$ 0.67	7 \$	1.07	\$ 1.34
Basic	\$	0.03			\$	0.15				
Diluted	\$	0.03			\$	0.15				
Net income (loss) per share:	·									
Basic	\$	(0.30) \$	\$ (0.19)	\$ (0	.08) \$	(0.82)	\$ 0.68	3 \$	1.09	\$ 1.42
Diluted	\$	(0.30) \$	(0.19)	\$ (0 63	.08) \$	(0.82)	\$ 0.67	7 \$	1.07	\$ 1.34
_		e months eptember 3	30, I	Year ended December 31,		ne months ended cember 31,			d March	
_	2003	2002	<u> </u>	2002		2001	2001	20	000	1999
				(In thousands,	excep	ot per share da	ta)			
ash flows related to: Operating activities \$	(1,414) \$ 18	,012 \$	23,553	\$	9,461 \$	· · · · · · · · · · · · · · · · · · ·	\$ 1	10,643	\$ 1,849
Investing activities	(423)	(232)	1,520		(13,902)	(6,269)	(1	11,693)	(9,234)
Financing activities \$	`) \$ (1 s of	,680) \$	(2,683) As of	\$	1,453 \$ As of	(42)	\$	233	\$ (2,655)
	Septer	nber 30,	Dece	ember 31,	Dec	ember 31,		As of	March 3	1,
<u>-</u>	20	003		2002		2001	2001		2000	1999
				(In thousands	, exce	pt per share d	ata)			
alance Sheet Data:										
Cash and cash equivalents	3	22,555	\$	24,538 \$		2,148	\$ 5,136	5 \$	5,544	\$ 6,36
Current assets		55,690)	61,361		60,661	70,594	ļ	65,528	56,032
Non-current assets		17,444		20,947		24,829	22,834	ļ	23,252	12,114
Current liabilities		13,094	ļ.	15,806		20,692	13,300)	18,161	14,247
Non-current liabilities		23	3	2,571			4,539)	4,211	2,108

	As Septem		_	As o Decembe			s of nber 31,		A	s of	March 3	1,	
Total stockholders' equity		60,017			63,931		64,798		75,589		66,408		51,791
Book value per common share outstanding Computation of Ratio of Earnings to Fi	\$ xed Charg	4.54 es:	\$		4.84	\$	4.91	\$	5.74	\$	5.05	\$	4.01
		Nine mon Septen				ar ended ember 31,	Nine mor ended December						
		2003		2002		2002	2001		_				
				(In	n thousai	nds)			_				
Pre-tax income (loss) from continuing operations	\$	(7,111)	\$	(4,348)	\$	(1,992)	\$ (18,144	1)				
Fixed changes:									_				
Interest expense		11		123		139		44	1				
Building rentals 33%		1,673		1,111		1,647		2,04	1				
Equipment rentals 33%		300		425		502		403	5				
Total fixed charges(A)	\$	1,984	\$	1,659	\$	2,288	\$	2,890)				
Pre-tax income (loss) from continuing operations plus fixed charges	\$	(5,127)	\$	(2,689)	\$	296	\$ (15,254	4)				
Ratio of earnings to fixed charges			(B)		(B)	0.13			(B)				
Fixed charges deficiency	\$	7,111	\$	4,348			\$	18,14	1				

(A)

Fixed charges consist of interest expense and one third of our rental expenses, which we believe to be representative of interest attributable to rent.

(B)

Due to Media Arts Group's loss for the nine months ended September 30, 2003, the nine months ended September 30, 2002 and the transition period ended December 31, 2001, the ratio coverage was less than 1:1. Additional earnings of \$7,111,000, \$4,347,000 and \$18,144,000 would have been required in each of those periods, respectively, to achieve a coverage ratio of 1:1.

6

DIRECTORS AND EXECUTIVE OFFICERS OF MEDIA ARTS GROUP, INC.

The following information describes the name and principal occupation of each of our directors, his or her position with us and the date he or she first became a director, if applicable. Unless otherwise indicated below, the business address and telephone number of each director is c/o Media Arts Group, Inc., 900 Lightpost Way, Morgan Hill, California 95037; (408) 201-5000.

			Director
Name	Age	Position(s) with Media Arts Group, Inc.	Since

Name	Age	Position(s) with Media Arts Group, Inc.	Director Since
Anthony D. Thomopoulos	65	Chairman of the Board of Directors and Chief Executive Officer	2000
Moe Grzelakowski	49	Member of the Board of Directors, Chairwoman of the Nominating and Governance Committee and Member of the Audit and Compensation Committees	2001
Eric H. Halvorson	54	Member of the Board of Directors and President	2001
Thomas Kinkade	45	Member of the Board of Directors and Art Director	1990
C. Joseph LaBonte	64	Member of the Board of Directors, Chairman of the Compensation Committee and Member of the Nominating and Governance Committee	2001
Donald Potter	58	Member of the Board of Directors, Chairman of the Audit Committee and Member of the Nominating and Governance Committee	2001
Richard E. Stearns	52	Member of the Board of Directors and Member of the Audit, Compensation and Nominating and Governance Committees	2002

There are no family relationships between any of our directors or executive officers.

Mr. Anthony D. Thomopoulos has been a member of our board of directors since July 2000, the Chairman of our board of directors since June 2001 and our Chief Executive Officer since August 2002. Mr. Thomopoulos served as our interim Chief Executive Officer from June 2001 to January 2002. From July 1999 to April 2001, Mr. Thomopoulos was the Vice Chairman of the Board of Directors of OnVANTAGE, Inc., formerly Familyroom.com. From March 1995 to December 1997, Mr. Thomopoulos was the Chief Executive Officer of MTM Entertainment and President of The Family Channel, both subsidiaries of International Family Entertainment, Inc. From April 1991 to February 1995 he was President of Amblin Television, a division of Amblin Entertainment. In 1989 he founded Thomopoulos Productions, an independent motion picture and television production company. In 1986 he was named Chairman of the Board of Directors of United Artist Pictures. Prior thereto, he spent 12 years at ABC in positions including Vice President of Prime Time Programs and President of ABC Broadcast Group.

Mr. Thomas Kinkade co-founded Media Arts Group through a preceding company in 1990 and has been the Art Director and a member of our board of directors since our inception. Mr. Kinkade has provided artwork to us for our productions since our inception. In addition, Mr. Kinkade provides strategic vision for the Thomas Kinkade brand, assisting in product development and communicating

65

our brand message through public appearances and books. Prior to March 1990, Mr. Kinkade was a self-employed artist. Mr. Kinkade is a citizen of the United States of America.

Ms. Moe Grzelakowski has been a member of our board of directors since November 2001. Since January 2001, Ms. Grzelakowski has been Chief Executive Officer of MindGifts. In April 2002 she became President of Oak Hills Country Club. From July 2000 to January 2001, Ms. Grzelakowski was Senior Vice President of Dell Computer. From January 1996 to July 2000, Ms. Grzelakowski was the Senior Vice President and General Manager of strategic marketing for Motorola's Network Solutions. Ms. Grzelakowski's earlier positions at Motorola included Vice President and General Manager of the International Cellular Infrastructure Division, and corporate Vice President and General Manager of Cellular Systems Group. Ms. Grzelakowski also held various managerial and technical positions at AT&T and Bell Laboratories from July 1977 to January 1996. Ms. Grzelakowski holds a bachelors degree in electrical engineering and masters degrees in computer science and management from Northwestern University.

Mr. Eric H. Halvorson has been a member of our board of directors since November 2001 and has served as our President since June 2003. From August 2000 to May 2003, Mr. Halvorson was a Visiting Professor of Business Law and Accounting at Pepperdine University, Malibu, California, where he instructed classes in Legal and Regulatory Environment of Business, Financial Accounting and

Corporate Finance. From January 1995 to August 2000, Mr. Halvorson was Executive Vice President and Chief Operating Officer of Salem Communications Corporation, a radio broadcasting company. He also served as General Counsel of Salem Communications Corporation from 1985 to 2000. From 1976 to 1985 and from 1988 to 1991, Mr. Halvorson practiced law with Godfrey & Kahn, a Milwaukee, Wisconsin based law firm, where he specialized in corporate, banking and securities law with a particular emphasis in mergers and acquisitions. Prior to practicing law, Mr. Halvorson was a Certified Public Accountant with Arthur Andersen & Company in Atlanta. Mr. Halvorson holds a Bachelor of Science degree in accounting from Bob Jones University and a Juris Doctor degree from Duke University School of Law. Mr. Halvorson is currently a director of Salem Communications Corporation and Intuitive Surgical, Inc.

Mr. C. Joseph LaBonte has been a member of our board of directors since November 2001. Mr. LaBonte was the President and Chief Executive Officer of Jenny Craig, Inc. from April 1994 to December 1997 and was a director of Jenny Craig, Inc. from August 1992 to January 1998. From May 1987 to January 1990, Mr. LaBonte was President, Chief Operating Officer and a director of Reebok International Ltd. From 1979 through 1983, Mr. LaBonte was President, Chief Operating Officer and a director of 20th Century Fox Film Corporation. Mr. LaBonte holds a Master of Business Administration degree from Harvard Business School, where he graduated as a Baker Scholar, a Bachelor of Science degree in Industrial Technology and an Associate of Mechanical Engineering degree from Northeastern University. Mr. LaBonte is the Chairman of The Vantage Group, an investment and financial advisory firm, which he founded in 1983. Mr. LaBonte also serves as a director of a privately owned company.

Mr. Donald Potter has been a member of our board of directors since November 2001. Mr. Potter has served as a consultant to senior management of substantial companies since 1973. From 1984 to present, Mr. Potter has served senior management of various domestic and foreign companies while heading Windermere Associates, Inc. From 1973 to 1984, Mr. Potter was with the international consulting firm of McKinsey & Company, where he served as a partner from July 1979 to March 1984. Mr. Potter holds a Master of Business Administration degree from Harvard Business School, where he graduated as a Baker Scholar, an award granted to five percent of the graduating class. Mr. Potter is also a magna cum laude graduate in Arts and Letters from the University of Notre Dame. Mr. Potter currently serves as a director of one private company.

66

Mr. Richard E. Stearns has been a member of our board of directors since January 2002. Mr. Stearns is currently President of World Vision U.S., a position he assumed in June 1998. Prior to that, Mr. Stearns held positions with Lenox Inc. from June 1987 to June 1998. He served as President and Chief Executive Officer of Lenox Inc. from 1995 to 1998. From 1975 to 1986 Mr. Stearns held various executive and management positions with Gillette, Parker Brother Games and the Franklin Mint. Mr. Stearns received a Bachelor's Degree from Cornell University in Neurobiology and a Master's Degree in Business Administration from the Wharton School at the University of Pennsylvania.

Executive Officers

The following information describes the name and principal occupation of each of our executive officers and his or her position with us. Unless otherwise indicated below, the business address and telephone number of each executive officer is c/o Media Arts Group, Inc., 900 Lightpost Way, Morgan Hill, California 95037; (408) 201-5000.

Name	Age	Position
Anthony D. Thomopoulos	65	Chairman of the Board and Chief Executive Officer
Eric H. Halvorson	54	Member of the Board of Directors and President
Herbert D. Montgomery	61	Executive Vice President and Chief Financial Officer
Daniel Byrne	41	Executive Vice President, Sales and Marketing
Robert C. Murray	43	Vice President, General Counsel and Secretary

Mr. Anthony D. Thomopoulos has been a member of our board of directors since July 2000, the Chairman of our board of directors since June 2001 and our Chief Executive Officer since July 2002. Mr. Thomopoulos served as our interim Chief Executive Officer from June 2001 to January 2002. From July 1999 to April 2001, Mr. Thomopoulos was the Vice Chairman of the Board of Directors of OnVANTAGE, Inc., formerly Familyroom.com. From March 1995 to December 1997, Mr. Thomopoulos was the Chief Executive Officer of MTM Entertainment and President of The Family Channel, both subsidiaries of International Family Entertainment, Inc. From April 1991 to February 1995 he was President of Amblin Television, a division of Amblin Entertainment. In 1989 he founded Thomopoulos Productions, an independent motion picture and television production company. In 1986 he was named Chairman of the

Board of Directors of United Artist Pictures. Prior thereto, he spent 12 years at ABC in positions including Vice President of Prime Time Programs and President of ABC Broadcast Group.

Mr. Eric H. Halvorson has been a member of our board of directors since November 2001 and has served as our President since June 2003. From August 2000 to May 2003, Mr. Halvorson was a Visiting Professor of Business Law and Accounting at Pepperdine University, Malibu, California, where he instructed classes in Legal and Regulatory Environment of Business, Financial Accounting and Corporate Finance. From January 1995 to August 2000, Mr. Halvorson was Executive Vice President and Chief Operating Officer of Salem Communications Corporation, a radio broadcasting company. He also served as General Counsel of Salem Communications Corporation from 1985 to 2000. From 1976 to 1985 and from 1988 to 1991, Mr. Halvorson practiced law with Godfrey & Kahn, a Milwaukee, Wisconsin based law firm, where he specialized in corporate, banking and securities law with a particular emphasis in mergers and acquisitions. Prior to practicing law, Mr. Halvorson was a Certified Public Accountant with Arthur Andersen & Company in Atlanta. Mr. Halvorson holds a Bachelor of Science degree in accounting from Bob Jones University and a Juris Doctor degree from Duke University School of Law. Mr. Halvorson is currently a director of Salem Communications Corporation and Intuitive Surgical, Inc.

67

Mr. Herbert D. Montgomery has served as our Executive Vice President and Chief Financial Officer since May 2001. From July 2000 through August 2003 he also served as a member of our board of directors. From November 1999 to May 2001, Mr. Montgomery was Executive Vice President, Chief Financial Officer and Treasurer of Standard Media International. From January 1998 to November 1999, Mr. Montgomery was the Senior Vice President, Chief Financial Officer and Treasurer of Cotelligent, Inc. From June 1994 to January 1998, Mr. Montgomery was Senior Vice President, Chief Financial Officer and Treasurer of Guy F. Atkinson. Mr. Montgomery has specialized in high growth, high technology and turn-around environments. He has taken three companies public and has served as Chief Financial Officer of technology, product and services companies over the last 20 years. In September 2001, Standard Media International filed for bankruptcy. Mr. Montgomery holds a Master of Science degree in Management and a Bachelor of Science degree in Finance from California State University, Northridge. Mr. Montgomery is currently a director of Institute for OneWorld Health.

Daniel P. Byrne has served as our Executive Vice President, Sales and Marketing since July 2003. Prior to re-joining us in July, Mr. Byrne was an associate consultant with Creative Brands Group from July 2001 through June 2003. From August 1999 through September 2000, Mr. Byrne was a member of the Board of Directors of Exclaim Inc. and served as a new business development consultant to Exclaim during that same time period. From January 1992 through July 1999, Mr. Byrne held a variety of executive positions at Media Arts Group including Senior Vice President of Marketing, Managing director of John Hine Ltd UK, and President of Lightpost Publishing. From Sept 1989 to 1992, Mr. Byrne was Director of Product Development for the Bradford Exchange. From December 1984 to July 1987, Mr. Byrne was the Product Manger for the Precious Moments division at Enesco Imports.

Mr. Robert C. Murray has been our Vice President, General Counsel and Secretary since January 2002. Prior to joining us, Mr. Murray was Vice President, General Counsel and Secretary of Affinity Internet, Inc. from February 2000 to December 2001. From October 1997 to February 2000, he was Vice President, General Counsel and Secretary of NewStar Media, Inc. and Dove Entertainment. Prior to that time, Mr. Murray practiced with the law firm of O'Melveny & Myers LLP in Los Angeles, California. He holds a Bachelor degree in Political Science from the University of Washington, a doctorate in Chemistry from the Massachusetts Institute of Technology and a law degree from Stanford University.

68

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table provides information relating to the beneficial ownership of our common stock as of November 7, 2003 by:

each stockholder known by us to own beneficially more than 5% of our common stock;

each of our executive officers named in the section of this proxy statement labeled "DIRECTORS AND EXECUTIVE OFFICERS OF MEDIA ARTS GROUP, INC. Executive Officers";

each of our directors; and

all of our directors and executive officers as a group.

Beneficial ownership is determined based on the rules of the Securities and Exchange Commission. The column captioned "Total Shares and Shares Underlying Exercisable Options Beneficially Owned" includes the number of shares of our common stock subject to options that are currently exercisable or will become exercisable on or before January 6, 2004, 60 days from November 7, 2003. The number of shares subject to options that each beneficial owner has the right to acquire on or before January 6, 2004 is listed separately under the column "Number of Shares Underlying Options." These shares are not deemed exercisable for purposes of computing the beneficial ownership of any other person. Percent of beneficial ownership is based upon 13,224,603 shares of our common stock outstanding as of November 7, 2003. The address for those individuals for which an address is not otherwise provided is c/o Media Arts Group, Inc., 900 Lightpost Way, Morgan Hill, California 95037. Unless otherwise indicated, we believe the stockholders listed below have sole voting or investment power with respect to all shares, subject to applicable community property laws.

Name and Address(1)	Number of Outstanding Shares Beneficially Owned	Number of Shares Underlying Options Exercisable on or before January 6, 2004	Total Shares and Shares Underlying Exercisable Options Beneficially Owned	Percentage of Outstanding Shares Beneficially Owned
Principal Stockholders				
Kenneth E. Raasch(2)	1,991,591		1,991,591	15.06
FMR Corp.(3)	1,322,000		1,322,000	10.00
Dimensional Fund Advisors(4)	797,950		797,950	6.03
Named Executive Officers and Directors				
Anthony D. Thomopoulos	1,000	10,000	11,000	*
Thomas Kinkade(5)	4,267,276	600,000	4,867,276	35.21
Moe Grzelakowski		15,000	15,000	*
Eric H. Halvorson	5,000	48,333	53,333	*
C. Joseph LaBonte		15,000	15,000	*
Herbert D. Montgomery(6)	2,000	91,667	93,667	*
Donald Potter		15,000	15,000	*
Richard E. Stearns	750	15,000	15,750	*
Daniel Byrne	500			*
Robert C. Murray		20,000	20,000	*
All current executive officers and directors as a group (10 persons)	4,276,526	830,000	5,106,026	36.33

Less than 1%

69

(1)
The address for Mr. Raasch is 333 W. Santa Clara Street, Suite 1000, San Jose, CA 95113. The address for FMR Corp. is 82 Devonshire Street, Boston, MA 02109. The address for Dimensional Fund Advisors is 1299 Ocean Avenue, 11th Floor, Santa Monica, CA 90401.

(2) Includes shares held in the Raasch 1993 Revocable Trust.

(3)
Information is derived from Schedule 13G filed by FMR Corp. with the Securities and Exchange Commission on or about June 10, 2002. Pursuant to such filing, FMR Corp. claimed beneficial ownership of the shares and disclaimed the power to vote

and to dispose or direct the disposition of the shares. This number consists of 1,322,000 shares beneficially owned by Fidelity Management & Research Company, a wholly-owned subsidiary of FMR Corp. and a registered investment adviser, as a result of acting as an investment adviser to various registered investment companies. FMR Group is the parent company of various Fidelity funds and related parties. The ownership of one investment company, Fidelity Low Priced Stock Fund, amounted to 1,322,000 shares of the Common Stock outstanding. Edward C. Johnson 3d, Chairman of FMR Corp., and Abigail P. Johnson, a Director of FMR Corp., are also listed on the Schedule 13G as beneficial owners of the 1,322,000 shares.

- (4)
 Information is derived from Schedule 13G (Amendment) filed by Dimensional Fund Advisors Inc. with the Securities and Exchange Commission on or about February 10, 2003. Pursuant to such filing, Dimensional Fund Advisors claimed voting and/or investment power over the shares and disclaimed beneficial ownership of the shares.
- (5) Includes shares held with Mr. Kinkade's wife, as joint tenants, and in the Kinkade Family Trust.
- (6) These shares are held in the Montgomery Family Trust.

70

PURCHASES BY MEDIA ARTS GROUP AND ITS DIRECTORS AND EXECUTIVE OFFICERS

None of Media Arts Group, Media Arts Group's executive officers or directors, any of the executive officers or directors of Mergerco, or Mr. Kinkade has purchased or sold shares of Media Arts Group stock during the past 60 days. Additionally, none of Media Arts Group or Mergerco have purchased shares of Media Arts Group's stock during the past two years.

On July 12, 2001, Mr. Kinkade entered into a Stock Sale Agreement with The Kenneth and Linda Raasch Trust whereby Mr. Kinkade agreed to buy an aggregate of 1,000,000 shares of Media Arts Group stock for \$4.00 per share at two separate closing dates. The agreement required the first 500,000 shares to be purchased on or before July 13, 2001 and the remaining 500,000 shares to be purchased on or before January 21, 2002, for an aggregate purchase price of \$4,000,000.

On February 8, 2002, Mr. Kinkade completed the purchase of the remaining 500,000 shares of Media Arts Group stock pursuant to the agreement.

PRICE RANGE OF COMMON STOCK AND DIVIDENDS

Our common stock has been traded on the New York Stock Exchange since December 7, 1998 under the symbol "MDA." Previously, our common stock had been traded on the NASDAQ National Market since our initial public offering on August 10, 1994 under the symbol "ARTS." The following table sets forth, for the periods indicated, the high and low sales prices per share of our common stock, as reported by the New York Stock Exchange:

	H	ligh	Low
Year ended March 31, 2001			
First Quarter	\$	7.12	\$ 3.62
Second Quarter		4.43	3.25
Third Quarter		5.37	3.18
Fourth Quarter		5.60	4.06
Nine-month period ended December 31, 2001			
Three months ended June 30	\$	4.48	\$ 2.45
Three months ended September 30		3.60	2.00
Three months ended December 31		3.69	1.88

			High		Low	
Year ended December 31, 2002)					
First Quarter	•		\$	3.27	\$	2.70
Second Quarter				4.50		2.35
Third Quarter				4.43		1.74
Fourth Quarter				3.75		1.57
Year ending December 31, 200	13					
First Quarter			\$	3.50	\$	2.48
Second Quarter				2.81		2.25
Third Quarter				2.70		2.02
Fourth Quarter (through	, 2003)					
, 2003 there were approxi	mately	holders of record of	of our cor	nmon stoc	k.	

As of

We have never paid cash dividends on our common stock. Our current intent is to retain earnings, if any, for use in the business and we do not anticipate paying cash dividends in the foreseeable future. Pursuant to the terms of our line-of-credit with Comerica Bank-California, which we renewed in June

71

2003, we are prohibited from paying any dividends or making any other distributions or payments on account for redemption, retirement or purchase of its capital stock without the bank's approval.

OTHER MATTERS

We are not aware of any matters to be presented at the special meeting other than those described in this proxy statement. However, if other matters should come before the special meeting, it is intended that the holders of proxies solicited hereby will vote on such matters in their discretion.

FUTURE STOCKHOLDER PROPOSALS

If the merger is completed, there will be no public participation in any future meetings of our stockholders. However, if the merger is not completed, our stockholders will continue to be entitled to attend and participate in our stockholders' meetings. If the merger is not completed, we will inform our stockholders, by press release or other means determined reasonable, of the date by which stockholder proposals must be received by us for inclusion in the proxy materials relating to our 2004 annual meeting, which proposals must comply with the rules and regulations of the Securities and Exchange Commission then in effect.

INDEPENDENT ACCOUNTANTS

The financial statements as of December 31, 2002 and 2001, and for the year ended December 31, 2002, the nine-months ended December 31, 2001 and the year ended March 31, 2001 incorporated by reference into this proxy statement, have been audited by PricewaterhouseCoopers LLP, independent accountants, as stated in their report included in our Annual Report on Form 10-K for the year ended December 31, 2002, incorporated by reference into this proxy statement.

WHERE STOCKHOLDERS CAN FIND MORE INFORMATION

We are subject to the informational reporting requirements of the Securities Exchange Act of 1934 and, accordingly, file reports, proxy statements and other information with the Securities and Exchange Commission. The reports, proxy statements and other

information can be inspected and copies made at the Securities and Exchange Commission's Public Reference Room at:

Public Reference Room Room 1200 450 Fifth Street, N.W. Washington, D.C. 20549

You may obtain copies of this information by mail from the Public Reference Room of the Securities and Exchange Commission, 450 Fifth Street, N.W., Room 1024, Washington, D.C. 20549, at rates set by the Securities and Exchange Commission. Please call the Securities and Exchange Commission at 1-800-SEC-0330 for further information on the operation of the public reference rooms. Our filings with the Securities and Exchange Commission are also available to the public from commercial document retrieval services and at the Web site maintained by the Securities and Exchange Commission at: http://www.sec.gov.

In addition, our common stock is listed on the New York Stock Exchange and similar information concerning us can be inspected and copied at the offices of the New York Stock Exchange, Inc., 20 Broad Street, New York, New York 10005.

We and Mergerco have filed a Section 13(e) Transaction Statement on Schedule 13E-3 with the Securities and Exchange Commission with respect to the merger. The Schedule 13E-3, including all amendments thereto, contains additional information about us and Mergerco. The Schedule 13E-3, including all amendments and exhibits filed or incorporated by reference as part of the Schedule 13E-3,

72

is available for inspection and copying at our principal executive offices during regular business hours, and may be obtained by mail, without charge, by written request directed to us at the following address:

Media Arts Group, Inc. 900 Lightpost Way Morgan Hill, California 95037 Attn: Investor Relations

This proxy statement is dated , 2003. You should not assume that the information contained in this proxy statement is accurate as of any date other than such date.

You should rely only on the information contained in this proxy statement to vote your shares at the special meeting. No persons have been authorized to provide any information or to make any representation other than those contained in this proxy statement, and if made or given, the information or representations must not be relied upon as having been authorized by us or any other person. We have supplied all information contained in this proxy statement relating to us and Mr. Kinkade has supplied all information contained in this proxy statement relating to Mergerco and the affiliated stockholders. No information on our website shall be deemed to be a part of this proxy statement or soliciting materials in connection with the transactions described herein.

INFORMATION INCORPORATED BY REFERENCE

The Securities and Exchange Commission allows us to incorporate by reference into this proxy statement, which means we may disclose important information by referring you to other documents filed separately with the Securities and Exchange Commission. The information incorporated by reference is deemed to be a part of this proxy statement, except for any information superseded by information contained in, or incorporated by reference into, this proxy statement.

This proxy statement incorporates by reference the documents listed below that we previously filed with the Securities and Exchange Commission. These documents contain important information about us and our business, financial condition and results of operations.

The following documents are incorporated by reference:

Annual Report on Form 10-K for the year ended December 31, 2002;

Quarterly Report on Form 10-Q for the quarter ended March 31, 2003;

Quarterly Report on Form 10-Q for the quarter ended June 30, 2003;

Quarterly Report on Form 10-Q for the quarter ended September 30, 2003; and

Current Reports on Form 8-K filed May 7, 2003, July 30, 2003, November 3, 2003 and November 4, 2003.

We also incorporate by reference each document we file under Sections 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934 after the date of this proxy statement and before the special meeting. Any statement contained in a document incorporated by reference in this proxy statement is deemed to be modified or superseded for all purposes to the extent that a statement contained in this proxy statement modifies or replaces the statement. Any statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute part of this proxy statement. Any references to the Private Securities Litigation Reform Act in our publicly filed documents which are incorporated by reference in this proxy statement or the Schedule 13E-3.

73

We undertake to provide by first class mail, without charge and within one business day of receipt of any written or oral request, to any person to whom a copy of this proxy statement has been delivered, a copy of any or all of the documents referred to above which have been incorporated by reference in this proxy statement, other than exhibits to the documents, unless the exhibits are specifically incorporated by reference therein. Requests for copies should be directed to Investor Relations at Media Arts Group, Inc., 900 Lightpost Way, Morgan Hill, California 95037 or (408) 201-5000. Any requested documents will be sent by first class mail or other equally prompt means within one business day of our receipt of such request.

74

APPENDIX A

AGREEMENT AND PLAN OF MERGER BY AND AMONG MEDIA ARTS GROUP, INC., MAIN STREET ACQUISITION COMPANY, INC. AND THOMAS KINKADE

DATED AS OF OCTOBER 31, 2003

TABLE OF CONTENTS

ARTICLE 1.	THE MERGER	A-1
1.1.	The Merger	A-1
1.2.	The Closing	A-1
1.3.	Effective Time	A-2
1.4.	Effect of the Merger	A-2
ARTICLE 2.	THE SURVIVING CORPORATION	A-2
2.1.	Certificate of Incorporation of the Surviving Corporation	A-2

2.2.	Bylaws of the Surviving Corporation	A-2
2.3.	Directors of the Surviving Corporation	A-2 A-2
2.4.	Officers of the Surviving Corporation	A-2 A-2
	egicent of the surviving corporation	A-2
ARTICLE 3.	EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT	
	CORPORATIONS; CANCELLATION AND CONVERSION OF SECURITIES	A-2
3.1.	Conversion of Company Common Stock	A-2
3.2.	Capital Stock of Mergerco	A-3
3.3.	Cancellation of Treasury Stock and Mergerco Owned Stock	A-3
3.4.	Exchange of Certificates	A-3
3.5.	Stock Options	A-5
3.6.	Warrants	A-5
3.7.	Dissenting Shares	A-6
3.8	Tax Withholding	A-6
3.9	Release of Exchange Fund	A-6
3.10	No Liability Under Abandoned Property Laws	A-7
A DOTTICE E A	DEDDEGENTATIONS AND WARD ANTHES OF THE GOMBANY	
ARTICLE 4. 4.1.	REPRESENTATIONS AND WARRANTIES OF THE COMPANY Organization and Qualification	A-7
4.1.	Authorization and Enforceability	A-7
4.2.		A-7
4.3.	Certificate of Incorporation and Bylaws	A-8
4.4.	Capitalization	A-8
	Consents and Approvals	A-8
4.6.	Company Action	A-9
4.7.	State Takeover Laws	A-9
4.8.	Vote Required	A-9
4.9.	Fairness Opinion	A-9
4.10.	No Finders	A-9
4.11.	Section 16 Matters	A-10
ARTICLE 5.	REPRESENTATIONS AND WARRANTIES OF MERGERCO AND THE PRINCIPAL	
merical 3.	AFFILIATED STOCKHOLDER	A-10
5.1.	Organization and Qualification	A-10
5.2.	Authorization	A-10
5.3.	Consents and Approvals	A-10
5.4.	No Finders	A-11
5.5.	Financing	A-11
5.6.	No Prior Activities	A-11
5.7.	No Other Affiliates	A-11
5.8	No Other Representations	A-11
		71.11

i

ARTICLE 6.	COVENANTS	A-11
6.1.	Conduct of Business of the Company	A-11
6.2.	Conduct of Business of Mergerco	A-12
6.3.	No Solicitation	A-12
6.4.	Company Stockholders Meeting	A-14
6.5.	Proxy Statement and Schedule 13E-3	A-15
6.6.	Access to Information	A-16

6.7.	Approvals and Consents; Cooperation	A-16
6.8.	Company Employee Stock Purchase Plan	A-17
6.9.	Take-over Statutes; Inconsistent Actions	A-17
6.10.	Financing	A-17
6.11.	Further Actions	A-17
6.12.	Officers' and Directors' Indemnification	A-18
6.13.	Notification of Certain Matters	A-18
6.14.	Payment of Fees for Advisors	A-19
ARTICLE 7.	CLOSING CONDITIONS	A-19
7.1.	Conditions to Obligations of Mergerco, the Principal Affiliated Stockholder and the Company	A-19
7.2.	Conditions to Obligations of Mergerco and the Principal Affiliated Stockholder	A-19
7.3.	Conditions to Obligations of the Company	A-20
ARTICLE 8.	TERMINATION AND ABANDONMENT	A-20
8.1.	Termination	A-20
8.2.	Effect of Termination	A-22
8.3.	Expense Reimbursement	A-22
8.4.	No Penalty; Costs of Collection	A-22
ARTICLE 9.	GENERAL PROVISIONS	A-23
9.1.	Non-Survival of Representations, Warranties and Covenants	A-23
9.2.	Amendment and Modification	A-23
9.3.	Waiver	A-23
9.4.	Notices	A-23
9.5.	Specific Performance	A-24
9.6.	Assignment	A-24
9.7.	Governing Law	A-25
9.8.	Knowledge	A-25
9.9.	Interpretation	A-25
9.10.	Publicity	A-25
9.11.	Entire Agreement	A-25
9.12.	Severability	A-25
9.13.	Counterparts	A-26
	ii	

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") is dated as of October 31, 2003, by and among Media Arts Group, Inc., a Delaware corporation (the "Company"), Main Street Acquisition Company, Inc., a Delaware corporation ("Mergerco"), and Thomas Kinkade (the "Principal Affiliated Stockholder").

WHEREAS, Mergerco was incorporated for the sole purpose of entering into the transactions contemplated by this Agreement; and

WHEREAS, at the Effective Time (as defined in Section 1.3), Mergerco will be merged with and into the Company, with the Company as the surviving corporation, on the terms and conditions set forth in this Agreement and the General Corporation Law of the State of Delaware ("DGCL") (the "Merger"); and

WHEREAS, the board of directors of the Company (acting through all four of its non-employee, independent directors, consisting of Moe Grzelakowski, C. Joseph LaBonte, Donald Potter and Richard E. Stearns, excluding Eric Halvorson, Thomas Kinkade and Anthony D. Thomopoulos; such four directors constituting a quorum and referred to in this Agreement for all purposes unless otherwise specified, as the "Board of Directors of the Company") has adopted resolutions declaring the advisability of this Agreement, has determined that this

Agreement, the Merger and the other transactions contemplated by this Agreement are fair to and in the best interests of the Company and its stockholders (other than Mergerco or any of the stockholders of the Company listed on Exhibit A hereto (such stockholders listed on Exhibit A are referred to as the "Affiliated Stockholders")) and has approved and adopted this Agreement, the Merger and the other transactions contemplated by this Agreement; and

WHEREAS, the Board of Directors of Mergerco has adopted resolutions declaring the advisability of this Agreement, has determined that this Agreement, the Merger and the other transactions contemplated by this Agreement are fair to and in the best interests of Mergerco and its sole stockholder and has approved and adopted this Agreement, the Merger and the other transactions contemplated by this Agreement and the sole stockholder of Mergerco has acted by written consent and has approved the Merger and the other transactions contemplated by this Agreement.

NOW, THEREFORE, in consideration of the foregoing, and of the representations, warranties, covenants and agreements contained herein, the parties hereby agree as follows:

ARTICLE 1. THE MERGER

- 1.1. *The Merger*. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL, at the Effective Time (as defined in Section 1.3), Mergerco will be merged with and into the Company, whereupon the separate corporate existence of Mergerco will cease, and the Company will continue as the surviving corporation in the Merger (the "Surviving Corporation") under the laws of the State of Delaware under the name "The Thomas Kinkade Company."
- 1.2. The Closing. The closing of the Merger (the "Closing") will take place at 10:00 a.m., Pacific Standard Time, on a date to be specified by the parties which will be no later than the first Business Day (as defined below) after the satisfaction or waiver (subject to applicable law and if permissible under this Agreement) of the conditions (excluding conditions that, by their nature, cannot be satisfied until the Closing Date (as defined below), but subject to such conditions), set forth in Article 7, unless this Agreement has been theretofore terminated pursuant to its terms or unless another time or date is agreed to in writing by the parties hereto (the actual time and date of the Closing being referred to herein as the "Closing Date"). The Closing will take place by telecopy exchange of signature pages with

A-1

originals to follow by overnight delivery, or in such other manner or at such place as the parties hereto may agree. The Company (acting through the Board of Directors of the Company) will as promptly as practicable notify Mergerco, and Mergerco will as promptly as practicable notify the Company, when the conditions to such party's obligation to effect the Merger contained in Article 7 have been satisfied. For purposes of this Agreement, a "Business Day" will mean any day that is not a Saturday, a Sunday or other day on which the offices of the Secretary of State of the State of Delaware is closed.

- 1.3. Effective Time. At the Closing, the Company and Mergerco will file, or cause to be filed, with the Secretary of State of the State of Delaware, a certificate of merger (the "Certificate of Merger") in accordance with the DGCL, in such form as is required by, and executed in accordance with, the relevant provisions of, the DGCL. The parties will take such other and further actions as may be required by law to make the Merger effective. The Merger will become effective at such time as the Certificate of Merger is accepted for filing by the Secretary of State of the State of Delaware or, if agreed to by the Company and Mergerco, at such later time or date as is set forth in the Certificate of Merger (the "Effective Time").
- 1.4. Effect of the Merger. At and after the Effective Time, the Merger will have the effects set forth in the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the property, rights, privileges, powers and franchises of the Company and Mergerco will vest in the Surviving Corporation, and all debts, liabilities, obligations, restrictions, disabilities and duties of the Company and Mergerco will become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving Corporation.

ARTICLE 2. THE SURVIVING CORPORATION

2.1. Certificate of Incorporation of the Surviving Corporation. At the Effective Time, the Certificate of Incorporation of the Company will be the certificate of incorporation of the Surviving Corporation.

- 2.2. Bylaws of the Surviving Corporation. At the Effective Time, the Bylaws of the Company, as in effect immediately prior to the Effective Time, will be the Bylaws of the Surviving Corporation until thereafter amended in accordance with applicable law, the provisions of the Certificate of Incorporation of the Surviving Corporation and the provisions of such Bylaws.
- 2.3. Directors of the Surviving Corporation. The directors of Mergerco immediately prior to the Effective Time will be the directors of the Surviving Corporation until the earlier of their respective deaths, resignations or removals or until their respective successors are duly elected and qualified, as the case may be.
- 2.4. Officers of the Surviving Corporation. The officers of the Company immediately prior to the Effective Time will be the officers of the Surviving Corporation until the earlier of their respective deaths, resignations or removals or until their respective successors are duly elected or appointed and qualified, as the case may be.

ARTICLE 3. EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS; CANCELLATION AND CONVERSION OF SECURITIES

3.1. Conversion of Company Common Stock. At the Effective Time, subject to the provisions of this Agreement (including without limitation Sections 3.4 and 3.7), automatically by virtue of the Merger and without any further action on the part of Mergerco, the Company or any holder of any share of capital stock of the Company or Mergerco, each share of common stock, par value \$0.01 per share, of the Company (the "Company Common Stock") issued and outstanding immediately prior to

A-2

the Effective Time (other than the Cancelled Shares (as defined in Section 3.3) and the Dissenting Shares (as defined in Section 3.7)) will be converted into the right to receive in cash, without interest, an amount equal to \$4.00 (the "Merger Consideration"). At the Effective Time, all such shares of Company Common Stock so converted into Merger Consideration will no longer be outstanding and will automatically be cancelled and retired and will cease to exist, and each holder of a certificate representing any such shares of Company Common Stock will cease to have any rights with respect thereto, except the right to receive the Merger Consideration upon surrender of such certificates in accordance with Section 3.4.

- 3.2. Capital Stock of Mergerco. At the Effective Time, automatically by virtue of the Merger and without any further action on the part of Mergerco, the Company or any holder of any share of capital stock of the Company or Mergerco, each share of common stock, par value \$0.01 per share, of Mergerco issued and outstanding immediately prior to the Effective Time will be converted into and become a number of fully paid and non-assessable shares of common stock, par value \$0.01 per share, of the Surviving Corporation equal to the number of whole and fractional shares of Mergerco common stock to be so converted. Such shares of common stock will constitute all of the issued and outstanding shares of capital stock of the Surviving Corporation at the Effective Time.
- 3.3. Cancellation of Treasury Stock and Mergerco Owned Stock. At the Effective Time, automatically by virtue of the Merger and without any further action on the part of Mergerco, the Company or any holder of any share of capital stock of the Company or Mergerco, each share of Company Common Stock issued and held immediately prior to the Effective Time in the Company's treasury and each share of Company Common Stock that is owned by Mergerco or any of the Affiliated Stockholders immediately prior to the Effective Time (the "Cancelled Shares") will automatically be cancelled and retired and will cease to exist, and no consideration will be delivered in exchange therefor.
 - 3.4. Exchange of Certificates.
- (a) Prior to the Effective Time, Mergerco will (i) designate and retain, at the Surviving Corporation's sole cost and expense, a commercial bank or trust company reasonably acceptable to the Company to act as the paying agent (the "Paying Agent") for the benefit of holders of shares of Company Common Stock (other than the Cancelled Shares and the Dissenting Shares) in the Merger and Mergerco will enter into an agreement with the Paying Agent pursuant to which, after the Effective Time, the Paying Agent will distribute the Merger Consideration on a timely basis, and (ii) irrevocably deposit or cause to be deposited with the Paying Agent, for the benefit of holders of shares of Company Common Stock and the holders of Company Options and Company Warrants, cash in an amount necessary to make the payments in respect of the conversion of shares of Company Common Stock and the cash-out of Company Options and Company Warrants pursuant to Sections 3.1, 3.5 and 3.6 (such cash being hereinafter referred to as the "Exchange Fund"). In the event that the funds comprising the Exchange Fund shall be or become insufficient to make the payments in respect of the conversion of shares of Company Common Stock and the cash-out of Company Options and Company Warrants pursuant to Sections 3.1, 3.5 and 3.6, then Mergerco will promptly deposit additional funds with the Paying Agent in an amount that is equal to such deficiency. The Paying Agent

must, pursuant to irrevocable instructions, deliver the cash contemplated to be paid pursuant to Sections 3.1, 3.5 and 3.6 out of the Exchange Fund. Except as contemplated by Section 3.9, the Exchange Fund must not be used for any other purpose. The Paying Agent will invest the Exchange Fund as directed by the Surviving Corporation (so long as such directions do not impair the rights of the holders of Company Common Stock) in direct obligations of, or money market funds substantially all the assets of which are invested in direct obligations of, the United States of America or any agency the obligations of which are backed by the full faith and credit of the United States of America. Any interest and other income resulting from such investments will be paid to the Surviving Corporation,

A-3

and no interest or other income will be paid or accrued on the Merger Consideration to the holders of Company Common Stock, Company Options or Company Warrants.

- (b) As promptly as reasonably practicable after the Effective Time, the Surviving Corporation will cause the Paying Agent to mail to each holder of record of a certificate or certificates (to the extent such certificates have not already been submitted to the Paying Agent) which immediately prior to the Effective Time represented outstanding shares (other than Cancelled Shares and Dissenting Shares) of Company Common Stock (the "Certificates"), at such holder's address of record (i) a letter of transmittal (which will be in customary form and will specify that delivery will be effected, and risk of loss and title to the Certificates will pass, only upon proper delivery of the Certificates to the Paying Agent and will be in such form and have such other provisions as the Surviving Corporation and the Paying Agent will reasonably specify) and (ii) instructions for use in effecting the surrender of the Certificates in exchange for the aggregate Merger Consideration into which the number of shares of Company Common Stock previously represented by such Certificates have been converted into the right to receive pursuant to this Article 3.
- (c) Upon surrender to the Paying Agent of a Certificate for cancellation, together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, and such other documents as may be reasonably required by the Paying Agent pursuant to such instructions, the holder of such Certificate will be entitled to receive in exchange therefor the Merger Consideration payable in respect of each share of Company Common Stock formerly represented by such Certificate pursuant to this Article 3, to be distributed as soon as practicable after the Effective Time (after giving effect to any required tax withholding) in each case without interest, and the Certificate so surrendered will immediately be cancelled. In the event of a transfer of ownership of shares of Company Common Stock which is not registered in the transfer records of the Company, the Merger Consideration may be issued to a transfere if the Certificate representing such shares of Company Common Stock is presented to the Paying Agent, accompanied by all documents required to evidence and effect such transfer and by evidence that any applicable stock transfer taxes have been paid. Until surrendered as contemplated by this Section 3.4, each Certificate will be deemed, at all times from and after the Effective Time and for all purposes, to represent only the right to receive, upon such surrender, the Merger Consideration payable in respect of each share of Company Common Stock formerly represented thereby pursuant to this Article 3.
- (d) Cash paid upon conversion of the shares of Company Common Stock in accordance with the terms hereof will be deemed to have been issued in full satisfaction of all rights pertaining to such shares of Company Common Stock and, from and after the Effective Time, there will be no further registration of transfers on the stock transfer books of the Surviving Corporation of shares of Company Common Stock that were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Surviving Corporation for any reason, they will be cancelled and exchanged in accordance with this Section 3.4 for the Merger Consideration payable in respect thereof pursuant to this Article 3. From and after the Effective Time, holders of Certificates will cease to have any rights as stockholders of the Company, except for the right to receive upon the surrender of such Certificates, in accordance with this Section 3.4, the Merger Consideration payable in respect of each share of Company Common Stock formerly represented thereby pursuant to this Article 3.
- (e) If any Certificate has been lost, stolen or destroyed, upon the delivery to the Paying Agent of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by such person of a bond, in such reasonable amount as the Surviving Corporation may direct, as indemnity against any claim that may be made against it with respect to such Certificate, the Paying Agent will issue in exchange for such lost, stolen or destroyed Certificate the Merger Consideration payable in respect of each share of Company Common Stock represented by such Certificate pursuant to this Article 3.

A-4

3.5. Stock Options.

- (a) Not later than 30 days prior to the Effective Time, the Company will send a notice (the "Option Notice") to all holders of outstanding options to purchase shares of Company Common Stock (other than options held by any of the Affiliated Stockholders) (the "Company Options") heretofore granted under any stock option, restricted stock or stock appreciation rights plan, program or arrangement of the Company or under any stock option or restricted stock award agreement, including, without limitation, the Company's Employee Stock Option Plan, Stock Option Plan for Outside Directors, 1998 Stock Incentive Plan, and 2002 Stock Plan (collectively, the "Company Stock Plans"): (i) specifying that such options will not be assumed in connection with the Merger, and (ii) specifying that any Company Options outstanding as of the Effective Time will thereafter represent only the right to receive the consideration, if any, specified in Section 3.5(c) in accordance with this Agreement.
- (b) The Company will permit each holder of a Company Option who desires to exercise all or any portion of such Company Option following receipt of the Option Notice to exercise such Company Option prior to the Effective Time.
- (c) At the Effective Time, each Company Option outstanding immediately prior to the Effective Time will, by virtue of the Merger and without any action on the part of the holder thereof, be converted into, and represent only, the right to receive (net of applicable withholding taxes) an amount in cash equal to the excess, if any, of (i) the product of the Merger Consideration multiplied by the number of shares of Company Common Stock issuable upon exercise of such Company Option (regardless of whether such Company Option is vested or not) immediately prior to the Effective Time over (ii) the aggregate exercise price of those shares of Company Common Stock issuable upon the exercise of such Company Option immediately prior to the Effective Time. The aggregate amount payable with respect to each such Company Option pursuant to this Section 3.5(c) is referred to as the "Option Cash-Out Amount."
- (d) Promptly following the Effective Time, the Surviving Corporation will cause the Paying Agent to mail to each holder (as of immediately prior to the Effective Time) of a Company Option which was converted into the right to receive the Option Cash-Out Amount pursuant to Section 3.5(c) hereof, (i) a letter of transmittal (which will be in such form and have such other provisions as the Surviving Corporation may reasonably specify), and (ii) instructions for use in receiving the Option Cash-Out Amount payable in respect of such Company Options pursuant to this Section 3.5. Upon the delivery of such letter of transmittal by or on behalf of a holder of a Company Option, duly completed and validly executed in accordance with the instructions thereto, together with the documentation representing the Company Options surrendered thereby, to the Paying Agent, such holder of a Company Option will be entitled to receive the Option Cash-Out Amount payable to it in respect of such Company Option pursuant to Section 3.5.

3.6. Warrants.

(a) Not later than 30 days prior to the Effective Time, the Company will send a notice to the holders of outstanding warrants to purchase shares of Company Common Stock (the "Company Warrants"): (i) specifying that such warrants will not be assumed in connection with the Merger, and (ii) specifying that any Company Warrants outstanding as of the Effective Time will thereafter represent only the right to receive the consideration, if any, specified in this Section 3.6(a) in accordance with this Agreement. At the Effective Time, each Company Warrant outstanding immediately prior to the Effective Time will, by virtue of the Merger and without any action on the part of the holder thereof, be converted into, and represent only, the right to receive (net of applicable withholding taxes), upon delivery thereof to the Company, an amount in cash equal to the excess, if any, of (i) the product of the Merger Consideration multiplied by the number of shares of Company Common Stock issuable upon exercise of such Company Warrant immediately prior to the Effective Time over (ii) the aggregate exercise price of those shares of Company Common Stock issuable upon

A-5

the exercise of such Company Warrant immediately prior to the Effective Time. The aggregate amount payable with respect to each such Company Warrant pursuant to this Section 3.6(a) will hereinafter be referred to as the "Warrant Cash-Out Amount."

- (b) Promptly following the Effective Time, the Surviving Corporation will cause the Paying Agent to mail to each holder (as of immediately prior to the Effective Time) of a Company Warrant which was converted into the right to receive the Warrant Cash-Out Amount pursuant to Section 3.6(a) hereof, (i) a letter of transmittal (which will be in such form and have such other provisions as the Surviving Corporation may reasonably specify), and (ii) instructions for use in receiving the Warrant Cash-Out Amount payable in respect of such Company Warrants pursuant to this Section 3.6. Upon the delivery of such letter of transmittal by or on behalf of a holder of a Company Warrant, duly completed and validly executed in accordance with the instructions thereto, together with the documentation representing the Company Warrant surrendered thereby, to the Paying Agent, such holder of a Company Warrant will be entitled to receive the Warrant Cash-Out Amount payable to it in respect of such Company Warrant pursuant to Section 3.6.
- 3.7. Dissenting Shares. Notwithstanding anything in this Agreement to the contrary, to the extent (if at all) that holders of Company Common Stock are entitled to appraisal rights under Section 262 of the DGCL, shares of Company Common Stock issued and

outstanding immediately prior to the Effective Time and held by a holder who has properly exercised and perfected his or her demand for appraisal rights under Section 262 of the DGCL or any successor provision (the "Dissenting Shares"), will not be converted into the right to receive the Merger Consideration, but the holders of Dissenting Shares will be entitled to receive from the Company such consideration as will be determined pursuant to Section 262 of the DGCL; provided, however, that if any such holder will have failed to perfect or will effectively withdraw or lose his or her right to appraisal and payment under the DGCL, each share of Company Common Stock held by such holder will thereupon be deemed to have been converted as of the Effective Time into the right to receive the Merger Consideration, without any interest thereon, upon the surrender of the Certificate representing such share of Company Common Stock pursuant to Section 3.4, and such shares will not be deemed to be Dissenting Shares. The Company will give Mergerco (i) prompt notice of any written notices or demands for appraisal of Company Common Stock received by the Company and (ii) the opportunity to participate and direct all negotiations and proceedings with respect to any such demands or notices. The Company will not, without the prior written consent of Mergerco, make any payment with respect to, or settle, offer to settle, or otherwise negotiate any such demands.

- 3.8 Tax Withholding. The Paying Agent or, at any time after 12 months following the Effective Time, the Surviving Corporation will be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Article 3 to any holder of shares of Company Common Stock or any holder of Company Options or Company Warrants such amounts as it is required to deduct and withhold from such payment under the Internal Revenue Code of 1986, as amended (the "Code"), or any provision of state, local or foreign tax law. To the extent that amounts are so withheld by the Paying Agent or the Surviving Corporation, as applicable, such withheld amounts will be treated for all purposes of this Agreement as having been paid to the holder of the shares of Company Common Stock or any holder of Company Options or Company Warrants in respect of which such deduction and withholding was made.
- 3.9 Release of Exchange Fund. To the extent permitted by applicable law, any portion of the Exchange Fund (plus any interest and other income received by the Paying Agent in respect of such funds) which remains undistributed to the holders of shares of Company Common Stock, and the holders of Company Options or Company Warrants, 12 months after the Effective Time will be delivered to the Surviving Corporation, upon demand, and any holders of shares of Company Common Stock, and any holders of Company Options or Company Warrants, who have not theretofore complied with this Section 3.4 must thereafter look, as general creditors, only to the Surviving Corporation for

A-6

the Merger Consideration, the Option Cash-Out Amount or Warrant Cash-Out Amount payable in respect thereof without interest, respectively, pursuant to this Article 3. Any portion of the Exchange Fund remaining unclaimed by holders of shares of Company Common Stock or holders of Company Options or Company Warrants three years after the Effective Time (or such earlier date, as is immediately prior to such time as such amounts would otherwise escheat to or become property of any government entity) will, to the extent permitted by applicable law, become the property of the Surviving Corporation, free and clear of any claims or interest of any person previously entitled thereto.

3.10 No Liability Under Abandoned Property Laws. Notwithstanding any other provision in this Article 3, neither the Paying Agent nor the Surviving Corporation will be liable to any holder of shares of Company Common Stock, or any holder of Company Options or Company Warrants, for any cash delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.

ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Mergerco as of the date hereof as follows:

4.1. Organization and Qualification. The Company and its two subsidiaries (the Company's only two subsidiaries being Thomas Kinkade Stores, Inc., a California corporation, and Media Arts Licensing Company, a California corporation, each referred to herein as a "Company Subsidiary") is a corporation duly organized, validly existing, and in good standing under the laws of its respective jurisdiction of incorporation and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. The Company and each Company Subsidiary is duly qualified and in good standing to do business in each jurisdiction in which the property owned, leased, or operated by it or the nature of the business conducted by it makes such qualification necessary and where the failure to qualify would have a Company Material Adverse Effect (as defined below). For all purposes of and under this Agreement, "Company Material Adverse Effect" means any effect, change, event, circumstance or condition that, individually or in the aggregate with all similar effects, changes, events, circumstances or conditions, has a material adverse effect on the business operations, assets, properties, results of operations, or financial condition of the Company and the Company Subsidiaries, with the

Company and the Company Subsidiaries considered as a whole; provided, however, that any effect, change, event, circumstance or condition that results from or arises out of one or more of the following will not be considered in determining whether a "Company Material Adverse Effect" has occurred or would be reasonably likely to occur: (a) changes in general economic conditions nationally or regionally, (b) changes in conditions (including as a result of changes in laws) affecting the industries in which the Company and the Company Subsidiaries compete, provided that such changes do not disproportionately affect the Company and the Company Subsidiaries, considered as a whole, in any material respect, (c) actions taken or omitted by the Company or any of the Company Subsidiaries required pursuant to this Agreement or upon the request of Mergerco or with the consent of Mergerco after the date of this Agreement, (d) the announcement or pendency of this Agreement and the transactions contemplated hereby, including, without limitation, voluntary departure of employees or downgrading of credit ratings, (e) changes in the Company's stock price, by itself, (f) failure by the Company to meet or exceed any analyst's revenue or earnings projections or forecasts, in any such case whether or not published or otherwise publicly available, (g) failure by the Company to meet internal revenue or earnings expectations, by itself, but not effects, changes, events, circumstances or conditions other than such failure to meet expectations that would otherwise constitute or be considered in determining whether a Company Material Adverse Effect has occurred and (h) any facts or circumstances known by any of the Affiliated Stockholders to exist as of the date hereof.

4.2. Authorization and Enforceability. The Company has the requisite corporate power and authority to execute and deliver this Agreement and, subject to obtaining the necessary approval of this

A-7

Agreement and the Merger by its stockholders under applicable law, the requisite corporate power and authority to consummate the transactions contemplated hereby. The execution and delivery of this Agreement by the Company and the consummation of the transactions contemplated hereby have been duly and validly authorized and approved by the Board of Directors of the Company, subject to obtaining the approval of the Company's stockholders as contemplated by Section 7.1(a), no other corporate action on the part of the Company or any Company Subsidiary is necessary to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement by Mergerco, constitutes the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and rules of law governing specific performance, injunctive relief, or other equitable remedies.

- 4.3. *Certificate of Incorporation and Bylaws*. The Company has heretofore furnished or made available to Mergerco a complete and correct copy of the Certificate of Incorporation and the Bylaws of the Company and the Certificate of Incorporation, Bylaws or equivalent organizational documents of each Company Subsidiary, each as in full force and effect as of the date hereof.
- 4.4. Capitalization. The authorized capital stock of the Company consists of 80,000,000 shares of Company Common Stock and 1,000,000 shares of preferred stock, par value \$0.01 per share ("Company Preferred Stock"). As of the date hereof, (a) 13,226,832 shares of Company Common Stock are issued and outstanding, all of which are validly issued, fully paid and nonassessable, (b) 313,843 shares of Company Common Stock are held in the treasury of the Company, (c) no shares of Company Common Stock are held by the Company Subsidiaries, (d) 1,804,433 shares of Company Common Stock are issuable upon the exercise of outstanding Company Options and Company Warrants, and (e) 88,138 shares of Company Common Stock are reserved for future issuance pursuant to the Company's Employee Stock Purchase Plan ("Company ESPP"). As of the date hereof, no shares of Company Preferred Stock are issued and outstanding. All outstanding shares of capital stock of the Company Subsidiaries are owned by the Company, directly or indirectly. Except as set forth in this Section 4.4, there are no options, warrants or other rights, agreements, arrangements or commitments of any character relating to the issued or unissued capital stock of the Company or any Company Subsidiary or obligating the Company or any Company Subsidiary. Each outstanding share of capital stock of each Company Subsidiary is duly authorized, validly issued, fully paid and nonassessable.
- 4.5. Consents and Approvals. Except as set forth in the disclosure schedule delivered by the Company to Mergerco, dated as of the date hereof, and except for (i) compliance with any applicable requirements of the Securities Act of 1933, as amended and the rules and regulations promulgated thereunder (the "Securities Act"), the Securities Exchange Act of 1934, as amended and the rules regulations promulgated thereunder (the "Exchange Act"), state takeover or securities laws, and the rules of any stock exchange or any other listing organization that are applicable to the Company, (ii) approval by the Company's stockholders, (iii) the filing and recordation of the Certificate of Merger as required by the DGCL, (iv) compliance with Section 262 of the DGCL regarding appraisal rights of the Company's stockholders and (v) compliance with any applicable requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), the execution and delivery of this Agreement by the Company and the consummation of the transactions contemplated hereby will not: (a) violate any provision of the Certificate of Incorporation or Bylaws of the Company or any Company Subsidiary; (b) violate any statute, rule, regulation, order, or decree of any federal, state, local, or foreign body or authority by which the Company or any Company Subsidiary or any of their respective properties or assets are bound; (c) require any filing with or

permit, consent, or approval of any federal, state, local, or foreign administrative, governmental or regulatory body or authority (a

A-8

"Governmental Entity"); or (d) result in any violation or breach of, or constitute (with or without due notice or lapse of time or both) a default under, result in the loss of any material benefit under, or give rise to any right of termination, cancellation, increased payments, or acceleration under, or result in the creation of any lien, charge, security interest, pledge or encumbrance of any kind or nature (any of the foregoing being a "Lien") on any of the properties or assets of the Company or any Company Subsidiary under, any of the terms, conditions, or provisions of any note, bond, mortgage, indenture, license, franchise, permit, authorization, agreement, or other instrument or obligation to which the Company or any Company Subsidiary is a party, or by which it or any of its properties or assets may be bound, except, (x) in the cases of clauses (b) or (c), where such violation, failure to make any such filing or failure to obtain such permit, consent or approval, could not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, and (y) in the case of clause (d), for any such violations, breaches, defaults, or other occurrences that would not have, individually or in the aggregate, a Company Material Adverse Effect.

- 4.6. Company Action. The Board of Directors of the Company, at a meeting duly called and held on October 31, 2003, has (i) declared the advisability of this Agreement, (ii) determined that this Agreement, the Merger and the other transactions contemplated by this Agreement are fair to and in the best interests of the Company, and its stockholders (other than Mergerco and the Affiliated Stockholders), (iii) approved and adopted this Agreement, the Merger and the other transactions contemplated by this Agreement and (iv) recommended that the Company's stockholders (other than Mergerco and the Affiliated Stockholders) approve and adopt this Agreement, the Merger and the other transactions contemplated by this Agreement.
- 4.7. State Takeover Laws. The approval of this Agreement and the transactions contemplated hereby by the Board of Directors of the Company are sufficient so that neither the restrictions on "business combinations" set forth in Section 203(a) of DGCL nor the provisions of any other "fair price," "moratorium," "control share acquisition," or other similar anti-takeover statute or regulation nor the provisions of any applicable anti-takeover provisions in the Certificate of Incorporation or Bylaws of the Company will apply to this Agreement or any of the transactions contemplated by this Agreement.
- 4.8. *Vote Required*. Except for the vote required to satisfy the condition set forth in Section 7.1(a), and except for any vote required under applicable law, the current Certificate of Incorporation of the Company or the rules of any stock exchange or other listing organization that are applicable to the Company, the affirmative vote of the holders of a majority of the shares of Company Common Stock outstanding and entitled to vote thereon on the record date established by the Board of Directors of the Company in accordance with the Bylaws of the Company, applicable law and this Agreement is the only vote of the holders of any class or series of capital stock of the Company necessary to adopt this Agreement and approve the Merger.
- 4.9. Fairness Opinion. The Board of Directors of the Company has received a written opinion from Jefferies & Company, Inc., financial advisor to the Company, dated as of the date hereof, to the effect that, subject to the qualifications and limitations stated therein, the Merger Consideration to be received by the holders of shares of the Company Common Stock in the Merger (other than Mergerco or any of the Affiliated Stockholders) as provided herein is fair to such holders from a financial point of view.
- 4.10. *No Finders*. Except for the engagement letter between the Company and Jefferies & Company, Inc., dated December 5, 2002, and amended on April 14, 2003, copies of which have been provided to Mergerco prior to the date of this Agreement, no act of the Company or any Company Subsidiary has given or will give rise to any claim against any of the parties hereto for a brokerage commission, finder's fee, or other like payment in connection with the transactions contemplated herein.

A-9

4.11. Section 16 Matters. Prior to the Effective Time, Mergerco and the Company shall take all such steps as may be required and permitted to cause the transactions contemplated by this Agreement, including any dispositions of shares of Company Common Stock (including derivative securities with respect to shares of Company Common Stock) by each individual who is or will be subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Company to be exempt under Rule 16b-3 promulgated under the Exchange Act.

ARTICLE 5. REPRESENTATIONS AND WARRANTIES OF MERGERCO AND

THE PRINCIPAL AFFILIATED STOCKHOLDER

Mergerco and the Principal Affiliated Stockholder hereby represent and warrant to the Company as of the date hereof as follows:

- 5.1. Organization and Qualification. Mergerco is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware and has all requisite corporate power and authority to own, lease, and operate its properties and to carry on its business as now being conducted. Mergerco is duly qualified and in good standing to do business in each jurisdiction in which the property owned, leased, or operated by it or the nature of the business conducted by it makes such qualification necessary.
- 5.2. Authorization. Mergerco has the requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement by Mergerco and the consummation of the transactions contemplated hereby have been duly and validly authorized and approved by the Board of Directors of Mergerco and by the sole stockholder of Mergerco, and no other corporate proceedings on the part of Mergerco are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Mergerco and the Principal Affiliated Stockholder and constitutes the valid and binding obligation of Mergerco and the Principal Affiliated Stockholder, enforceable against each of them in accordance with its terms, subject to laws of general application relating to bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and rules of law governing specific performance, injunctive relief, or other equitable remedies.
- 5.3. Consents and Approvals. Except for (i) compliance with any applicable requirements of the Securities Act, the Exchange Act and state takeover and securities laws, (ii) the filing and recordation of the Certificate of Merger as required by the DGCL and the rules of any stock exchange or other listing organization that may be applicable to the Company, (iii) compliance with Section 262 of the DGCL regarding appraisal rights of the Company's stockholders, and (iv) compliance with any applicable requirements of the HSR Act, the execution and delivery of this Agreement by Mergerco and the Principal Affiliated Stockholder and the consummation of the transactions contemplated hereby will not: (a) violate any provision of the Articles or Certificate of Incorporation or Bylaws of Mergerco; (b) violate any statute, rule, regulation, order, or decree of any federal, state, local or foreign body or authority by which Mergerco or the Principal Affiliated Stockholder or any of their respective properties or assets are bound; (c) require any filing with or permit, consent, or approval of any Governmental Entity; or (d) require any consent under or result in a violation or breach of, or constitute (with or without due notice or lapse of time or both) a default (or give rise to any right of termination, cancellation or acceleration of the performance required thereunder) under any of the terms, conditions or provisions of any contract, agreement, instrument or any obligation to which Mergerco or the Principal Affiliated Stockholder is a party or by which either of them or any of their respective properties or assets is bound.

A-10

- 5.4. *No Finders.* No act of Mergerco or the Principal Affiliated Stockholder has given or will give rise to any claim against any of the parties hereto for a brokerage commission, finder's fee, or other like payment in connection with the transactions contemplated herein.
- 5.5. Financing. Mergerco has received, and previously provided to the Board of Directors of the Company, a fully executed commitment letter from GE Corporate Financial Services, Inc. (the "Lender"), dated as of October 24, 2003, as amended through October 29, 2003, providing for a portion of the funds necessary to consummate the Merger and pay the aggregate Merger Consideration, Option Cash-Out Amounts and Warrant Cash-Out Amounts payable pursuant to Article 3, and describing the terms and conditions upon which the Lender will arrange and provide such financing (the "Commitment Letter"). Mergerco has accepted the Commitment Letter and paid all fees then due thereunder as of the date of this Agreement. The Commitment Letter is in full force and effect on the date hereof and has not been withdrawn, amended or modified in any respect, and there is no breach or default existing (or which with notice or lapse of time or both may exist) thereunder. There are no facts or circumstances known to Mergerco or the Principal Affiliated Stockholder or any of their respective affiliates that any of them believes is likely to (i) prevent the conditions to the financing described in the Commitment Letter from being satisfied, or (ii) prevent Mergerco from receiving financing pursuant to the terms of the Commitment Letter. The aggregate proceeds of the financing contemplated by the Commitment Letter, together with the Company's existing cash resources as reflected in the unaudited consolidated balance sheet of the Company as of September 30, 2003, are sufficient to pay the aggregate Merger Consideration, Option Cash-Out Amounts and Warrant Cash-Out Amounts pursuant to Article 3, and to pay the anticipated fees and expenses of Mergerco related to the Merger (the "Required Cash Amount").
- 5.6. No Prior Activities. Mergerco was organized solely to effect the transactions contemplated by this Agreement. Since its founding, except for its obligations incurred in connection with the transactions contemplated by this Agreement (including the financing contemplated by the Commitment Letter), Mergerco has not engaged in any business or other activity of any kind or character whatsoever, entered into or become subject to any contract, agreement, arrangement or other understanding (whether written or oral), or otherwise incurred or become subject to any liability or other obligations to any person that would adversely affect Mergerco's ability to perform its obligations under this Agreement and the transactions contemplated thereby.

- 5.7. No Other Affiliates. Exhibit A hereto includes a complete and accurate list of each affiliate of the Principal Affiliated Stockholder and the aggregate number of shares of Company Common Stock held of record or beneficially by each of them.
- 5.8 No Other Representations. Mergerco and the Principal Affiliated Stockholder acknowledge that, other than as set forth in this Agreement, neither the Company nor any of its directors, officers, employees, agents or other representatives (other than the Affiliated Stockholders) makes any representations or warranties either express or implied, as to the accurateness or completeness of any information provided or made available to Mergerco or any of its directors, officers, employees, agents or other representatives in connection with this Agreement or the transactions contemplated hereby.

ARTICLE 6. COVENANTS

- 6.1. *Conduct of Business of the Company*. Except as contemplated by this Agreement or as set forth in Section 6.1 of the Company Disclosure Schedule, during the period from the date of this Agreement to the Effective Time or the earlier termination of this Agreement, neither the Company nor any Company Subsidiary will, without the prior written consent of Mergerco:
 - (a) amend or otherwise change its Certificate of Incorporation or Bylaws or other organizational documents;

A-11

- (b) issue, sell or grant, or authorize the issuance, sale or grant of any shares of capital stock of the Company or any Company Subsidiary, or any options, warrants, convertible securities or other rights of any kind to acquire any shares of such capital stock, or any other ownership interest (including, without limitation, any phantom interest), of the Company or any Company Subsidiary (except for the issuance of shares of Company Common Stock pursuant to the exercise of Company Options and Company Warrants outstanding on the date of this Agreement or pursuant to the Company ESPP);
- (c) declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock, other than dividends and distributions by a Company Subsidiary to its parent in accordance with applicable law;
 - (d) reclassify, combine, split, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock;
- (e) acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the stock or assets of, or by any other manner, any business or any corporation, partnership, joint venture, association or other business organization which would be material to the Company and the Company Subsidiaries, taken as a whole;
- (f) incur any indebtedness for borrowed money or guarantee any such indebtedness of another person, issue or sell any debt securities or warrants or other rights to acquire any debt securities of the Company or any of the Company Subsidiaries, guarantee any debt securities of another person, enter into any "keep well" or other agreement to maintain any financial statement condition of another person or enter into any arrangement having the economic effect of any of the foregoing, except for borrowings under current credit facilities, loans and for lease obligations, in each case incurred in the ordinary course of business consistent with past practice;
- (g) make any loans, advances or capital contributions to, or investments in, any other person, other than to the Company or any Company Subsidiary made in the ordinary course of business consistent with past practices;
- (h) adopt resolutions providing for or authorizing a liquidation or a dissolution, except as part of an Acquisition Proposal (as defined below); or
- (i) take, or agree to commit to take, or fail to take any action that would make any representation or warranty of the Company contained herein inaccurate such that the conditions in Section 7.2(a) will not be satisfied at, or as of any time prior to, the Effective Time; or
- (j) enter into, or publicly announce an intention to enter into, any contract, agreement, commitment, plan or arrangement to, or otherwise agree or consent to do any of the foregoing actions set forth in this Section 6.1.

6.2. Conduct of Business of Mergerco. From the date of this Agreement to the Effective Time, Mergerco will not take, or agree to commit to take, or fail to take any action that would make any representation or warranty of Mergerco contained herein inaccurate such that the conditions in Section 7.3(a) will not be satisfied at, or as of any time prior to, the Effective Time.

6.3. No Solicitation.

(a) Subject to the provisions of Section 6.3(d), the Company will not, and will cause the Company Subsidiaries not to, and will not authorize or knowingly permit any of its respective officers, directors, employees, financial advisors, counsel, representatives and agents (collectively, "Representatives") to, (i) directly or indirectly, solicit, initiate or knowingly encourage or facilitate the making of an Acquisition Proposal (as defined below), (ii) engage in or knowingly encourage in any way negotiations or discussions concerning, or provide any non-public information to, any Third Party

A-12

(as defined below) relating to an Acquisition Proposal, or which may reasonably be expected to lead to an Acquisition Proposal, or (iii) agree to, recommend or endorse any Acquisition Proposal; provided, however, that nothing contained in this Section 6.3 or in any other provision of this Agreement will prohibit the Company or the Board of Directors of the Company from taking and disclosing to the Company's stockholders a position under Rules 14d-9 or 14e-2(a) promulgated under the Exchange Act or from making any disclosure to the Company's stockholders in order to fulfill, in the good faith judgment of the Board of Directors of the Company after consultation with its legal counsel, its fiduciary duties to such stockholders under applicable law.

- (b) As used in this Agreement, the term "Acquisition Proposal" means any offer or proposal for (i) a transaction or series of related transactions pursuant to which any person (or "group" of persons as such term is defined under Section 13(d) of the Exchange Act) other than Mergerco (a "Third Party") would acquire 25% or more of the outstanding shares of Company Common Stock or voting power (or of securities or rights convertible into or exercisable for such shares of Company Common Stock or voting power), including without limitation a tender offer or an exchange offer which, if consummated, would result in a Third Party acquiring 25% or more of the outstanding shares of Company Common Stock or voting power (or of securities or rights convertible into or exercisable for such shares of Company Common Stock or voting power), (ii) a merger or other business combination involving the Company pursuant to which any Third Party would acquire securities representing 25% or more of the voting power of the outstanding securities of the company surviving the merger or business combination, or (iii) any other transaction pursuant to which any Third Party would acquire control of assets (including for this purpose the outstanding equity securities of any Company Subsidiary) comprising at least 25% or more of the book value of the assets of the Company.
- (c) As used in this Agreement, a "Superior Proposal" means any unsolicited, bona fide offer made by a Third Party to acquire a majority of the outstanding shares of Company Common Stock beneficially owned by stockholders of the Company or to acquire substantially all of the assets of the Company on terms that the Board of Directors of the Company has in good faith determined, after consultation with its financial advisors and outside counsel, to be more favorable to the Company's stockholders (other than Mergerco or any of the Affiliated Stockholders) than the Merger, taking into account all factors it deems relevant (including whether, in the good faith judgment of the Board of Directors of the Company, such Third Party is able to finance the transaction and obtain all required regulatory approvals).
- (d) Notwithstanding the provisions of Section 6.3(a), nothing in this Agreement will prohibit the Board of Directors of the Company from, prior to the date on which the Company's stockholders adopt this Agreement in accordance with the DGCL, furnishing nonpublic information to, or entering into discussions or negotiations with, any Third Party that makes a bona fide Acquisition Proposal that was not solicited in violation of Section 6.3(a), if, and only to the extent that, (i) the Board of Directors of the Company determines in good faith after consultation with its legal counsel that the failure to furnish such information or to participate in such negotiations or discussions with respect to such Acquisition Proposal would be inconsistent with the fiduciary duties of the Board of Directors of the Company to the Company's stockholders under applicable law; (ii) prior to furnishing nonpublic information to, or entering into substantive discussions and negotiations with, such Third Party after the date hereof, the Company (A) provides two Business Days' prior written notice to Mergerco to the effect that it intends to furnish information to, or enter into discussions or negotiations with, such Third Party, and naming and identifying the Third Party making the Acquisition Proposal, and (B) receives from such Third Party an executed confidentiality agreement with terms and conditions (except the "standstill" provisions) that are no less favorable to the Company, in the aggregate, than the terms and conditions of the confidentiality agreements between the Company and the Affiliated Stockholders; and (iii) the Company keeps Mergerco informed on a reasonably current basis of the

A-13

status and the material terms and conditions of such Acquisition Proposal and the status of any discussions or negotiations with such third party related thereto.

- (e) Upon execution of this Agreement, the Company will immediately terminate all discussions with Third Parties concerning any Acquisition Proposal and will request that such Third Parties promptly return any confidential information furnished by the Company in connection with any Acquisition Proposal. The Company will not waive any provision of any confidentiality, standstill or similar agreement entered into with any Third Party regarding any Acquisition Proposal and prior to the Closing will enforce all such agreements in accordance with their terms. Notwithstanding the foregoing, such discussions may be reinstated or waivers may be provided if the conditions of this Section 6.3(d) are otherwise satisfied.
- (f) Nothing contained in this Section 6.3 will (i) permit the Company to terminate this Agreement (except as specifically provided in Article 8 hereof), or (ii) permit the Company to enter into any agreement providing for an Acquisition Proposal (other than the confidentiality agreement as provided and in the circumstances and under the conditions set forth above) for as long as this Agreement remains in effect.
 - 6.4. Company Stockholders Meeting.
- (a) The Company will take all action necessary in accordance with the DGCL and the Company's Certificate of Incorporation and Bylaws to cause a meeting of its stockholders (the "Company Stockholders' Meeting") to be duly called and held to consider and vote upon the approval and adoption of this Agreement and the Merger, and the Company will use its commercially reasonable efforts to hold the Company Stockholders' Meeting as soon as practicable after the date of this Agreement. Subject to the provisions of Section 6.4(b), the Board of Directors of the Company, will recommend such approval and adoption of this Agreement and the Merger by the Company's stockholders as provided herein and will use its commercially reasonable efforts to solicit such approval, including, without limitation, timely mailing the Proxy Statement, unless the Board of Directors of the Company determines in good faith, after consultation with its legal counsel, that the inclusion of such recommendation or solicitation therein would be inconsistent with the fiduciary duties of the Board of Directors of the Company to the Company's stockholders under applicable law.
- (b) The Board of Directors of the Company will not withdraw, modify or change in a manner adverse to Mergerco, its recommendation to the Company's stockholders unless the Board of Directors of the Company has received a Superior Proposal and the Board of Directors of the Company determines in good faith, after consultation with its legal counsel, that the failure to withdraw, modify or change such recommendation would be inconsistent with the fiduciary duties of the Board of Directors of the Company to the Company's stockholders under applicable law. Subject to compliance with the terms of Section 6.3, any withdrawal, change or modification of the recommendation of the Board of Directors of the Company in accordance with the previous sentence will not constitute a breach of the Company's representations, warranties, covenants or agreements contained in this Agreement. Unless this Agreement is previously terminated in accordance with Article 8, the Company will submit this Agreement to its stockholders at the Company Stockholders' Meeting in accordance with Section 6.4(a) even if the Board of Directors of the Company has withdrawn, modified or changed its recommendation of this Agreement or the transactions contemplated by this Agreement and, except as required by applicable law, will not postpone or adjourn such meeting or the vote by the Company's stockholders upon this Agreement and the Merger to another date without Mergerco's approval (which consent shall not be unreasonably withheld, delayed or conditioned).
- (c) Nothing contained in this Section 6.4 or in any other provision of this Agreement will prohibit the Company or the Board of Directors of the Company from taking and disclosing to the Company's stockholders prior to the date of the Company Stockholder's Meeting, a position under Rules 14d-9 or 14e-2(a) promulgated under the Exchange Act or from making any disclosure to the Company's

A-14

stockholders in order to fulfill, in good faith judgment of the Board of Directors of the Company its fiduciary duties to such stockholders under applicable law.

- 6.5. Proxy Statement and Schedule 13E-3.
- (a) As promptly as practicable after the execution of this Agreement, the Company and Mergerco will cooperate to prepare a joint Rule 13E-3 Transaction Statement relating to the transactions contemplated hereby (together with any amendments thereto, the "Schedule 13E-3") and the Company will prepare and file with the SEC a proxy statement (together with any amendments thereto, the "Proxy Statement") for use in connection with the solicitation of proxies for the Company Stockholders' Meeting. Both Mergerco and the

Company will cause the Schedule 13E-3 to comply as to form in all material respects with the applicable provisions of the Exchange Act. The Company will cause the Proxy Statement to comply as to form in all material respects with the applicable provisions of the Exchange Act, the rules and regulations of any stock exchange or other listing organization that are applicable thereto and the DGCL. Both the Company and Mergerco will furnish to each other all information concerning the Company or Mergerco each may reasonably request in connection with such actions and the preparation of the Schedule 13E-3 and the Proxy Statement. Mergerco will be given a reasonable opportunity to review and comment on all filings by the Company with the SEC in connection with the transactions contemplated hereby, including the Proxy Statement and the Schedule 13E-3 and any amendments or supplements thereto, and all mailings to the Company's stockholders in connection with the transaction contemplated by this Agreement. The Company will be given a reasonable opportunity to review and comment on all filings by Mergerco with the SEC in connection with the transactions contemplated hereby, including the Schedule 13E-3 and any amendment or supplement thereto. The Company, with the cooperation of Mergerco, will use its commercially reasonable efforts to cause the Proxy Statement to be mailed to each of the Company's stockholders as promptly as practicable after the compliance with SEC filing requirements and satisfactory resolution of all SEC comments thereon, if any. The Company will also promptly as practicable file, and, if required, mail to the Company's stockholders, any amendment to the Proxy Statement which may become necessary after the date the Proxy Statement is first mailed to the Company's stockholders. The Company and Mergerco will also promptly as practicable file any amendment to the Schedule 13E-3 which may become necessary after the date the Schedule 13E-3 is first filed with the SEC.

- (b) No amendment or supplement to the Proxy Statement or the Schedule 13E-3 may be made by the Company without the prior approval of Mergerco, which approval will not be unreasonably withheld, conditioned or delayed. No amendment or supplement to the Schedule 13E-3 may be made by Mergerco without the prior approval of the Company, which approval will not be unreasonably withheld, conditioned or delayed. The Company will advise Mergerco promptly after it receives notice thereof of any request by the SEC or any stock exchange or other listing organization for amendment of the Proxy Statement or the Schedule 13E-3 or comments thereon and responses thereto or requests by the SEC for additional information.
- (c) Subject to the provisions of Sections 6.3 and 6.4, the Proxy Statement will include the recommendation of the Board of Directors of the Company to the stockholders of the Company that they vote in favor of the adoption of this Agreement and the Merger and that the Board of Directors of the Company has determined that the Merger is fair to, and in the best interests of, the stockholders of the Company (other than the Affiliated Stockholders).
- (d) The information supplied by the Company included in the Proxy Statement and the Schedule 13E-3 will not, at (i) the time the Proxy Statement and the Schedule 13E-3 is filed with the SEC; (ii) the time the Proxy Statement (or any amendment thereof or supplement thereto) is first mailed to the stockholders of the Company, (iii) the time of the Company Stockholders' Meeting and (iv) the Effective Time, contain any untrue statement of a material fact or fail to state any material fact

A-15

required to be stated therein, in light of the circumstances under which they were made, or necessary in order to make the statements therein not misleading, except that no representation or warranty is made by the Company with respect to statements made in or incorporated by reference therein based on information supplied by or on behalf of Mergerco or the Affiliated Stockholders specifically for inclusion or incorporation by reference therein. If at any time prior to the Effective Time any event or circumstances relating to the Company or any of the Company Subsidiaries, or their respective officers and directors (other than Thomas Kinkade or any officers or directors affiliated with Thomas Kinkade), should be discovered by the Company that should be set forth in an amendment or a supplement to the Proxy Statement or the Schedule 13E-3, the Company will promptly inform Mergerco.

- (e) The information supplied by or on behalf of Mergerco included in the Schedule 13E-3 or for inclusion in the Proxy Statement will not, at (i) the time the Proxy Statement and the Schedule 13E-3 is filed with the SEC, (ii) the time the Proxy Statement (or any amendment thereof or supplement thereto) is first mailed to the stockholders of the Company, (iii) the time of the Company Stockholders' Meeting and (iv) the Effective Time, contain any untrue statement of a material fact or fail to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading. If at any time prior to the Effective Time any event or circumstance relating to Mergerco or its officers and directors should be discovered by Mergerco that should be set forth in an amendment or a supplement to the Schedule 13E-3 or the Proxy Statement, Mergerco will promptly inform the Company.
- 6.6. Access to Information. Except as required (i) pursuant to any confidentiality agreement or similar agreement or arrangement to which the Company or any of the Company Subsidiaries is a party (in which case the Company will use all commercially reasonable efforts to provide acceptable alternative arrangements, not in violation of such agreement or arrangement, for disclosure to Mergerco or its advisors), (ii) in order to preserve any attorney-client or other legal privilege or (iii) pursuant to applicable law, from the date hereof through the Effective Time, the Company will afford to Mergerco and to Mergerco's accountants, officers, directors, employees, counsel, and other representatives reasonable access, during normal business hours and upon reasonable prior notice to all of its properties, books,

data, contracts, commitments, and records, and will furnish promptly to Mergerco all information concerning the Company's and the Company Subsidiaries' businesses, prospects, properties, liabilities, results of operations, financial condition, officers, employees, consultants, distributors, customers, suppliers, and others having dealings with the Company as Mergerco may reasonably request.

6.7. Approvals and Consents; Cooperation. Upon the terms and subject to the conditions set forth in this Agreement, each of the parties agrees to cooperate with each other and to use commercially reasonable efforts to promptly take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Merger and the other transactions contemplated by this Agreement, including, without limitation, (i) the obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Entities and the making of all necessary registrations, submissions of information, applications and filings (including filings with Governmental Entities) and the taking of all reasonable steps as may be necessary to obtain an approval or waiver from, or to avoid an action or proceeding by, any Governmental Entity, (ii) the obtaining and maintenance of all necessary consents, approvals, permits, authorizations and other confirmations or waivers from third parties, (iii) the defending of any lawsuits or other legal proceedings, whether judicial or administrative, investigating or challenging this Agreement or the consummation of any of the transactions contemplated by this Agreement, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Entity vacated or reversed and (iv) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this

A-16

Agreement. Prior to the Effective Time, Mergerco intends to change its name to "The Thomas Kinkade Company." The parties hereto agree to permit such change and to cooperate with Mergerco to obtain any necessary consents and complete any necessary filings required to effect such name change; provided, however, that in the event this Agreement is terminated prior to consummation of the Merger, Mergerco agrees to change its name and to refrain from using the name in any way that may conflict with the business of the Company.

- 6.8. Company Employee Stock Purchase Plan. The rights of participants in the Company ESPP with respect to any offering then underway under the Company ESPP shall be determined by treating the last business day prior to the Effective Time as the last day of such Offering Period (as defined in the Company ESPP) and by making such other pro rata adjustments as may be necessary to reflect the shortened offering but otherwise treating such shortened Offering Period as a fully effective and completed Offering Period for all purposes under the Company ESPP. Outstanding rights to purchase shares of Company Common Stock shall automatically be exercised, and each share of Company Common Stock purchased pursuant to such exercise shall by virtue of the Merger, and without any action on the part of the holder thereof, be converted into the right to receive the Merger Consideration, without issuance of certificates representing issued and outstanding shares of Company Common Stock to participants under the Company ESPP. Immediately prior the Effective Time, the Company ESPP shall be terminated.
- 6.9. Take-over Statutes; Inconsistent Actions. If any "fair price," "moratorium," "control share," "business combination," "stockholder protection" or similar or other anti-takeover statute or regulation enacted under any state or federal law becomes applicable to the Merger or any of the other transactions contemplated hereby, the Company and the Board of Directors of the Company will grant such approvals and take all such actions as are within its authority and are reasonable so that the Merger and the other transactions contemplated hereby may be consummated as promptly as practicable on the terms contemplated hereby and otherwise use commercially reasonable efforts to eliminate the effects of such statute or regulation on the Merger and the transactions contemplated hereby. The Company has not and, during the term of this Agreement will not, adopt, effect or implement any "stockholders' rights plan," "poison pill" or similar arrangement.
- 6.10. Financing. Each of Mergerco and the Principal Affiliated Stockholder will use their respective commercially reasonable efforts to consummate the financing contemplated by the Commitment Letter or such other financing as Mergerco and the Principal Affiliated Stockholder may deem necessary and appropriate in order to finance the transactions contemplated hereby. The Company will use its commercially reasonable efforts to assist and cooperate with Mergerco and the Principal Affiliated Stockholder to satisfy on or before the Closing Date all of the conditions to closing the financing contemplated by the Commitment Letter. Mergerco and the Principal Affiliated Stockholder shall keep the Company reasonably apprised of the status of their efforts to consummate the financing contemplated by the Commitment Letter or such other financing as Mergerco may deem necessary or appropriate in order to finance the transactions contemplated hereby. In the event that either Mergerco or the Principal Affiliated Stockholder is informed, orally or in writing, that any or all of the lenders that are parties to the Commitment Letter (or any commitment or other agreement contemplating additional or alternative financings) do not intend to consummate the financing contemplated by the Commitment Letter (or other commitment or agreement), are terminating or withdrawing the Commitment Letter (or other agreement or commitment) or are amending or otherwise modifying the Commitment Letter (or other commitment or agreement), Mergerco and the Principal Affiliated Stockholder shall promptly inform the Company thereof and provide the Company with any written material or other communication related thereto.

6.11. *Further Actions*. Subject to the terms and conditions herein provided and without being required to waive any conditions herein (whether absolute, discretionary, or otherwise), each of the parties hereto agrees to use commercially reasonable efforts to take, or cause to be taken, all action,

A-17

and to do, or cause to be done, all things necessary, proper, or advisable to consummate and make effective the transactions contemplated by this Agreement. In case at any time after the Effective Time any further action is necessary or desirable to carry out the purposes of this Agreement, the proper officers and directors of each party to this Agreement will take all such necessary action.

6.12. Officers' and Directors' Indemnification.

- (a) The Company and the Surviving Corporation agree that all rights to indemnification and all limitations on liability existing in favor of any individual who was on or at any time prior to the Effective Time, a director, officer, employee or agent of the Company (an "Indemnified Person") in respect of acts or omissions of such Indemnified Person on or prior to the Effective Time, as provided in the Certificate of Incorporation or Bylaws of the Company or any agreement between an Indemnified Person and the Company in effect as of the date of this Agreement, shall survive the Merger and shall be observed by the Surviving Corporation to the fullest extent available under Delaware law for a period of six years from the Effective Time. The Surviving Corporation will, at its selection, either: (i) cause to be maintained in effect the Company's current directors' and officers' liability insurance policy with respect to claims arising from facts or events that occurred at or prior to the Effective Time; (ii) extend the discovery or reporting period under the Company's current policy for six years from the Effective Time to maintain in effect directors' and officers' liability insurance with respect to claims arising from facts or events that occurred at or prior to the Effective Time for those persons who are currently covered by the Company's directors' and officers' liability insurance policy on terms no less favorable than the terms of such current insurance policy; or (iii) substitute coverage under other policies providing coverage on terms and conditions that are no less advantageous to such persons than the Company's current insurance with respect to claims arising from facts or events that occurred at or prior to the Effective Time; provided, however, that in no event will the Surviving Corporation be required to expend for any such coverage an amount per year in excess of 300% of the annual premium currently paid by the Company for such insurance or replacement insurance or to expend for an extended period reporting endorsement a total amount in excess of 300% of the annualized cost of the Company's current policy.
- (b) In the event that the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other person or entity and is not the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties and assets to any person or entity, then, and in each such case, proper provision will be made so that the successors and assigns of the Surviving Corporation assume the obligations set forth in this Section 6.12.
- (c) The provisions of this Section 6.12 are intended to be for the benefit of, and will be enforceable by, each Indemnified Person and his or her heirs and representatives and are in addition to, and not in substitution for, any other rights to indemnification or contribution that any such person may have by contract or otherwise.
- 6.13. Notification of Certain Matters. The Company will give prompt written notice to Mergerco, and Mergerco and the Principal Affiliated Stockholder will give prompt written notice to the Company, of (i) the occurrence, or nonoccurrence, of any event the occurrence, or nonoccurrence, of which would be likely to cause any representation or warranty contained herein to be untrue or inaccurate such that the conditions in Section 7.2(a) or Section 7.3(a), respectively, would fail to be satisfied on the Closing Date and (ii) any material failure of the Company, or Mergerco or the Principal Affiliated Stockholder, as the case may be, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder such that the conditions in Section 7.2(b) or Section 7.3(b), respectively, would fail to be satisfied on the Closing Date; provided, however, that the delivery of any notice pursuant to this Section 6.13 will not limit or otherwise affect the remedies available hereunder to the party receiving such notice.

A-18

6.14. *Payment of Fees for Advisors*. At the Closing, the Company will pay in full all amounts owed to the legal and financial advisors of the Board of Directors of the Company and the Company, provided, in the case of payments to financial advisors, that such amounts are provided for pursuant to the terms and conditions of the agreements entered into with such financial advisors.

ARTICLE 7. CLOSING CONDITIONS

- 7.1. Conditions to Obligations of Mergerco, the Principal Affiliated Stockholder and the Company. The respective obligations of each party hereto to consummate the Merger and to consummate the other transactions contemplated by this Agreement to occur on the Closing Date will be subject to the satisfaction at or prior to the Closing Date of the following conditions:
- (a) This Agreement, the Merger and the transactions contemplated hereby will have been approved by (i) the affirmative vote of the holders of a majority of the outstanding shares of Company Common Stock in accordance with the Company's Certificate of Incorporation, Bylaws and the DGCL, and (ii) the affirmative vote of the holders of a majority of the shares of Company Common Stock that are cast either for or against approval of this Agreement (excluding any shares of Company Common Stock held by Mergerco or any of the Affiliated Stockholders or any officers or directors of Mergerco or the Company).
- (b) No Governmental Entity or court of competent jurisdiction will have enacted, issued, promulgated, enforced or entered any law, rule, regulation or order which is then in effect and has the effect of making the Merger illegal or otherwise prohibiting consummation of the Merger.
- (c) All consents, approvals, authorizations legally required to be obtained from any Governmental Entity to consummate the Merger will have been obtained and will be final and in full force and effect as of the Closing, except for such consents, approvals and authorizations the failure of which to obtain could not reasonably be expected to have a Company Material Adverse Effect.
- (d) Any waiting period (and any extension thereof) applicable to the Merger under the HSR Act will have been terminated or will have expired.
- 7.2. Conditions to Obligations of Mergerco and the Principal Affiliated Stockholder. The obligation of Mergerco and the Principal Affiliated Stockholder to consummate the Merger will be subject to the fulfillment or waiver by Mergerco and the Principal Affiliated Stockholder at or prior to the Closing of the following additional conditions:
- (a) Each representation and warranty of the Company contained in this Agreement that is qualified by "Company Material Adverse Effect" or reference to "material" or "in all material respects" or like variations shall be true and correct as of the date hereof and the Closing Date as though such representations and warranties were made on the Closing Date (except those representations and warranties that address matters only as of a particular date will remain true and correct as of such date), and (ii) each representation and warranty of the Company contained in this Agreement that is not qualified by "Company Material Adverse Effect" or reference to "material" or "in all material respects" or like variations shall be true and correct as of the date hereof and the Closing Date as though such representations and warranties were made on the Closing Date (except those representations and warranties that address matters only as of a particular date will remain true and correct as of such date), except in the case of the foregoing clause (ii) for any inaccuracies that have not had, individually or in the aggregate, a Company Material Adverse Effect as of the Closing Date. At the Closing, the Company shall deliver a Certificate signed by the Chief Executive Officer and the Chief Financial Officer of the Company to the foregoing effect.

A-19

- (b) The Company has performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
- (c) Since the date of this Agreement, there will not have occurred or come into existence any change, event, occurrence, state of facts or development that has had, individually or in the aggregate, a Company Material Adverse Effect.
- (d) Mergerco shall have received the funding contemplated by the Commitment Letter or Mergerco will otherwise have immediate access to sufficient funds under any other commitment acceptable to Mergerco to enable performance of the obligations of Mergerco under this Agreement.
- 7.3. *Conditions to Obligations of the Company*. The obligation of the Company to consummate the Merger will be subject to the fulfillment or waiver by the Company at or prior to the Closing of the following additional conditions:
- (a) Each representation and warranty of Mergerco contained in this Agreement is true and correct in all material respects on the date of this Agreement and shall be true as of the Closing Date as though such representations and warranties were made on such date (except

those representations and warranties that address matters only as of a particular date will remain true and correct as of such date).

(b) Mergerco has performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement to be performed or complied with by Mergerco on or prior to the Closing Date.

ARTICLE 8. TERMINATION AND ABANDONMENT

- 8.1. *Termination*. This Agreement may be terminated and the Merger and the other transactions contemplated hereby may be abandoned at any time prior to the Effective Time, whether before or after approval by the stockholders of the Company as provided in Section 7.1(a), only:
 - (a) by mutual written consent duly authorized by the respective Boards of Directors of Mergerco and the Company;
- (b) by either Mergerco or the Company if the Merger has not been consummated on or before January 31, 2004 (the "End Date"); provided, however, that the party wishing to terminate under this Section 8.1(b) has not breached in any material respect its obligations under this Agreement in any manner that was the proximate cause of, or resulted in, the failure to consummate the Merger by such date; provided, further, however, that in the event that the term of the Commitment Letter is extended or Mergerco and the Principal Affiliated Stockholder procure alternative financing arrangements in order to finance the transactions contemplated hereby, the End Date shall be automatically (and without any further action on the part of the Company, Mergerco or the Principal Affiliated Stockholder) extended to the earlier to occur of (i) a date that is commensurate with the expiration date of the Commitment Letter, as so extended, or the expiration date of such alternative financing commitment, as applicable (it being understood and hereby agreed that neither Mergerco nor the Principal Affiliated Stockholder shall have any obligation under this Agreement to so extend the expiration date of the Commitment Letter or otherwise procure alternative financing arrangements to finance the transactions contemplated hereby) or (ii) March 1, 2004;
- (c) by either Mergerco or the Company if a court of competent jurisdiction or other Governmental Entity has issued a final nonappealable order, decree, or ruling, or taken any other action, having the effect of permanently restraining, enjoining, or otherwise prohibiting the Merger;

A-20

- (d) by either Mergerco or the Company if, at the Company Stockholders' Meeting, the vote of the Company's stockholders for approval and adoption of this Agreement and the Merger contemplated by Section 7.1(a) is not obtained, except that the right to terminate this Agreement under this Section 8.1(d) will not be available to any party whose failure to perform any obligation under this Agreement has been the proximate cause of, or resulted in, the failure to obtain the requisite vote of the stockholders of the Company;
- (e) by Mergerco if (i) the Company has breached its obligations under Section 6.3 in any material respect, or (ii) the Board of Directors of the Company has recommended to the stockholders of the Company any Acquisition Proposal or will have resolved or announced an intention to do so, or (iii) the Board of Directors of the Company withdrawn or modified in a manner adverse to Mergerco its approval or recommendation of the Merger (provided, however, that in the event the Company's Board of Directors shall take and disclose to the Company's stockholders a position under Rules 14d-9 or 14e-2(a) promulgated under the Exchange Act, or otherwise make disclosure to the Company's stockholders in order to fulfill, in the good faith judgment of the Company's Board of Directors, its fiduciary duties to such stockholders under applicable law, such action by itself shall not be deemed to constitute a withdrawal or modification of its approval or recommendation of the Merger), or (iv) a tender offer or exchange offer for 25% or more of the outstanding shares of Company Common Stock is announced or commenced and the Board of Directors of the Company recommends acceptance of such tender offer or exchange offer by the Company's stockholders;
- (f) by Mergerco is not in material breach of its obligations under this Agreement and (ii) there has been a material breach by the Company of any of its representations, warranties or obligations under this Agreement such that the conditions in Section 7.2 hereof will not be satisfied as of the Closing Date; provided, however, that if such a breach is curable by the Company and such cure is reasonably likely to be accomplished prior to the End Date, then, for so long as the Company continues to exercise commercially reasonable efforts to cure such breach, Mergerco may not terminate this Agreement under this Section 8.1(f);
- (g) by the Company if (i) the Company is not in material breach of its obligations under this Agreement and (ii) there has been a material breach by Mergerco or the Principal Affiliated Stockholder of any of its or his representations, warranties or obligations under this

Agreement such that the conditions in Section 7.3 hereof will not be satisfied as of the Closing Date; provided, however, that if such a breach is curable by Mergerco or the Principal Affiliated Stockholder and such cure is reasonably likely to be accomplished prior to the End Date, then, for so long as Mergerco or the Principal Affiliated Stockholder continues to exercise commercially reasonable efforts to cure such breach, the Company may not terminate this Agreement under this Section 8.1(g);

(h) by the Company if, prior to approval of the Merger by the Company's stockholders as contemplated in Section 7.1(a) and as a result of a Superior Proposal, the Board of Directors of the Company determines, in its good faith judgment after consultation with its legal counsel and financial advisor, that the failure to terminate this Agreement and accept such Superior Proposal would be inconsistent with the fiduciary duties of the Board of Directors of the Company to the Company's stockholders under applicable law; provided, however, that before the Company may terminate this Agreement pursuant to this Section 8.1(h), the Company must give notice to Mergerco of the proposed termination under this Section 8.1(h) and Mergerco will have the right, in its sole discretion, to offer to amend this Agreement to make an offer that is at least as favorable to the stockholders of the Company as the Superior Proposal and the Company will negotiate in good faith with Mergerco with respect to such proposed amendment; provided, further, that if Mergerco and the Company are unable to reach an agreement with respect to the Mergerco's proposed amendment within five Business Days after Mergerco's receipt of such notice, the Company may terminate this Agreement pursuant to this Section 8.1(h);

A-21

- 8.2. Effect of Termination. The party desiring to terminate this Agreement will give written notice of such termination to the other party. Except for any willful or intentional breach of this Agreement by any party hereto (which breach and liability therefor will not be affected by the termination of this Agreement or the payment of any Reimbursable Expenses (as defined in Section 8.3 hereof)), if this Agreement is terminated pursuant to Section 8.1 hereof, this Agreement will become void and of no effect with no liability on the part of any party hereto; provided, however, that notwithstanding such termination the agreements contained in Sections 8.2, 8.3, 8.4 and Article 9 hereof will survive the termination hereof.
 - 8.3. Expense Reimbursement.
- (a) Whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby will be paid by the party incurring such costs and expenses, subject to Section 8.3(b) and Section 8.3(d). The Company shall pay all costs and expenses in connection with printing and mailing the Proxy Statement and soliciting proxies and of obtaining any consents of third parties (other than fees and expenses paid pursuant to the HSR Act). Mergerco shall pay the filing fee in connection with the filings required under the HSR Act.
- (b) The Company agrees to reimburse Mergerco, in immediately available funds by wire transfer to an account designated by Mergerco, an amount equal to Mergerco's and the Principal Affiliated Stockholder's out-of-pocket costs and expenses (which are reasonably documented) that are reasonably incurred in connection with this Agreement, the Merger and the transactions contemplated hereby (including without limitation, all reasonable legal, accounting, printing and financial advisory fees, fees and expenses payable to any financing sources, filing fees and expenses in connection with filings pursuant to the HSR Act, and other expenses) up to an amount not to exceed \$1,000,000 in the aggregate (collectively, the "Reimbursable Expenses") if:
 - (i) this Agreement is terminated by Mergerco pursuant to Section 8.1(e) or 8.1(f) hereof; or
 - (ii) this Agreement is terminated by the Company pursuant to Section 8.1(h) hereof.
- (c) The Company will pay the Reimbursable Expenses required to be paid pursuant to Section 8.3(b) (if all conditions thereto have been satisfied) (i) on the date of termination of this Agreement if by the Company or (ii) not later than five Business Days after termination of this Agreement if by Mergerco.
- (d) Mergerco and the Principal Affiliated Stockholder agree to reimburse the Company, in immediately available funds by wire transfer to an account designated by the Company, an amount equal to the Company's Reimbursable Expenses if (i) this Agreement is terminated by the Company pursuant to Section 8.1(g) hereof, or (ii) the condition set forth in Section 7.2(d) shall have failed to be satisfied primarily as a result of either (A) the failure by the Principal Affiliated Stockholder to provide the security contemplated by the Commitment Letter or (B) the failure of the Company and/or Mergerco to issue the subordinated debt as contemplated by the Commitment Letter. Mergerco and the Principal Affiliated Stockholder will pay the Reimbursable Expenses (up to an amount not to exceed \$1,000,000 in the aggregate) required to be paid pursuant to this Section 8.3(d) (if all conditions thereto have been satisfied) not later than five Business Days after termination of this Agreement by the Company pursuant to Section 8.1(g).

8.4. No Penalty; Costs of Collection. The Company, Mergerco and the Principal Affiliated Stockholder acknowledge that the agreements contained in Section 8.3(b) and Section 8.3(d) are an integral part of the transactions contemplated by this Agreement and Reimbursable Expenses constitute liquidated damages and not a penalty, and that, without these agreements, Mergerco and the Principal Affiliated Stockholder on the one hand, and the Company on the other hand, would not enter into this Agreement. If the Company or Mergerco and the Principal Affiliated Stockholder, as applicable, fails to pay promptly any amounts due pursuant to Section 8.3, the Company or Mergerco and the Principal

A-22

Affiliated Stockholder, as applicable, will also pay Mergerco's and the Principal Affiliated Stockholder's, on the one hand, or the Company's, on the other hand, as applicable, reasonable costs and expenses (including legal fees and expenses) in connection with any action, including the filing of any lawsuit or other legal action, taken to collect payment, together with interest on the amount of the unpaid Reimbursable Expenses under Section 8.3, accruing from its due date, at an interest rate per annum equal to two percentage points in excess of the prime commercial lending rate quoted by *The Wall Street Journal*. Any change in the interest rate hereunder resulting from a change in such prime rate will be effective at the beginning of the date of such change in such prime rate.

ARTICLE 9. GENERAL PROVISIONS

- 9.1. *Non-Survival of Representations, Warranties and Covenants*. The representations, warranties, covenants and agreements contained in this Agreement and in any certificate delivered pursuant to this Agreement by any person will terminate at the Effective Time or upon the termination of this Agreement pursuant to Section 8.1, as the case may be, except that this Section 9.1 will not limit any covenant or agreement of the parties which by its terms contemplates performance after the Effective Time or after termination of this Agreement, including those contained in Sections 6.12, 8.2, 8.3, 8.4 and Article 9.
- 9.2. Amendment and Modification. This Agreement may be amended by the parties hereto by action taken by or on behalf of the Board of Directors of the Company and the Board of Directors of Mergerco at any time prior to the Effective Time; provided, however, that, after the adoption of this Agreement by the stockholders of the Company as provided herein, no amendment may be made which would reduce the amount or change the type of consideration to be received by the stockholders of the Company pursuant to the Merger or otherwise adversely affect the rights of the Company's stockholders. This Agreement may not be amended except by an instrument in writing signed by the parties hereto.
- 9.3. *Waiver*. Except as otherwise provided herein, at any time prior to the Effective Time, any party hereto may (i) extend the time for the performance of any obligation or other act of any other party hereto, (ii) waive any inaccuracy in the representations and warranties contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any agreement or condition contained herein. Any such extension or waiver will be valid if set forth in an instrument in writing signed by the party or parties to be bound thereby.
- 9.4. *Notices*. All notices and other communications hereunder will be in writing and will be deemed given (i) on the date of delivery if delivered personally, or upon confirmation of receipt if delivered by telecopy, facsimile or email, (ii) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service, or (iii) on the fifth Business Day following the date of mailing if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder must be delivered as set forth below, or pursuant to instructions as may be designated in writing by the party to receive such notice:
 - (a) If to the Company, to:

Media Arts Group, Inc. 900 Lightpost Way Morgan Hill, CA 95037 Attention: Robert C. Murray Facsimile: (408) 201-5092

A-23

with a copy to:

The Board of Directors of Media Arts Group, Inc. 900 Lightpost Way Morgan Hill, CA 95037 Attention: Donald V. Potter Facsimile: (925) 377-2022

and:

Wilson Sonsini Goodrich & Rosati Professional Corporation One Market, Spear Tower Suite 3300 San Francisco, CA 94105 Attention: Michael S. Ringler, Esq.

Facsimile: (415) 947-2099

and:

Wilson Sonsini Goodrich & Rosati Professional Corporation 650 Page Mill Road Palo Alto, CA 94304 Attention: Chris Fennell, Esq.

Facsimile: (650) 493-6811

(b) if to Mergerco, to:

Main Street Acquisition Company, Inc. 900 Lightpost Way Morgan Hill, CA 95037 Attention: Eric H. Halvorson Facsimile: (408) 201-5192

with a copy to:

Gibson, Dunn & Crutcher LLP 4 Park Plaza Irvine, CA 92614

Attention: Thomas D. Magill, Esq.

Facsimile: (949) 451-4220

- 9.5. Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, the parties further agree that each party will be entitled to an injunction or restraining order to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other right or remedy to which such party may be entitled under this Agreement, at law or in equity.
- 9.6. Assignment. This Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests, or obligations hereunder will be assigned by any of the parties hereto without the prior written consent of the other parties, nor is this Agreement intended to confer upon any other person except the parties hereto any rights or remedies hereunder, except that

A-24

from and after the Effective Time, Article 3 and Section 6.13 of this Agreement will inure to the benefit of the persons identified therein who shall be third party beneficiaries of the provisions set forth therein, with full right and authority to enforce such provisions.

- 9.7. *Governing Law*. This Agreement will be construed in accordance with and governed by the law of the State of Delaware (without giving effect to choice of law principles thereof).
- 9.8. *Knowledge*. As used in this Agreement or the instruments, certificates or other documents required hereunder, the term "knowledge" of an entity will mean knowledge actually possessed by any director or executive officer of such entity.
- 9.9. *Interpretation.* The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement will refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section references are to this Agreement unless otherwise specified. Whenever the words "include," "includes" or "including" are used in this Agreement, they will be deemed to be followed by the words "without limitation." The table of contents, article and section headings contained in this Agreement are inserted for reference purposes only and will not affect the meaning or interpretation of this Agreement. This Agreement will be construed without regard to any presumption or other rule requiring the resolution of any ambiguity regarding the interpretation or construction hereof against the party causing this Agreement to be drafted.
- 9.10. *Publicity*. Upon execution of this Agreement by Mergerco, and the Company, the parties will jointly issue a press release, as agreed upon by them. The parties intend that all future statements or communications to the public or press regarding this Agreement or the Merger will be mutually agreed upon by them and neither party will, without such mutual agreement or the prior consent of the other, issue any statement or communication to the public or to the press regarding this Agreement, or any of the terms, conditions, or other matters with respect to this Agreement, except as required by law or the rules of any stock exchange or other listing organization that may be applicable and then only (i) to the extent required by applicable law or the rules of any stock exchange or other listing organization that may be applicable; and (ii) following prior notice to the other party and an opportunity for the other party to discuss with the disclosing party (which notice will include a copy of the proposed statement or communication to be issued to the press or public). The foregoing will not restrict Mergerco's or the Company's communications with their respective employees or customers in the ordinary course of business. In addition, nothing contained in this Section 9.10 or elsewhere in this Agreement shall be deemed to restrict the rights of the Company or the Board of Directors of the Company under Sections 6.3, 6.4 and 6.5 hereof.
- 9.11. Entire Agreement. This Agreement, including the exhibits and schedules hereto and the Company Disclosure Schedule referred to herein and the confidentiality agreements executed by and between the Affiliated Stockholders and the Company, embody the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein and supersede all prior agreements and the understandings between the parties with respect to such subject matter. No discussions regarding or exchange of drafts or comments in connection with the transactions contemplated herein will constitute an agreement among the parties hereto. Any agreement among the parties will exist only when the parties have fully executed and delivered this Agreement.
- 9.12. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms and provisions of this Agreement will nevertheless remain in full force and effect so long as the economics or legal substance of the transactions contemplated hereby are not affected in any manner materially adverse to any party. Upon determination that any term or other provision hereof is invalid, illegal or incapable of being enforced, the parties hereto will negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible to the fullest extent permitted by applicable law in an

A-25

acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.

9.13. *Counterparts*. This Agreement may be executed in two or more counterparts, each of which will be deemed an original, and all of which together will constitute one and the same instrument.

[Signature page follows]

A-26

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

MEDIA ARTS GROUP, INC.

/s/ ANTHONY D. THOMOPOULOS

By:

Name: A. D. Thomopoulos Title: Chairman/CEO

MAIN STREET ACQUISITION COMPANY, INC.

/s/ THOMAS KINKADE

By:

Name: Thomas Kinkade Title: President

/s/ THOMAS KINKADE

THOMAS KINKADE

A-27

EXHIBIT A to Merger Agreement

AFFILIATED STOCKHOLDERS

Stockholder ¹	Number of Shares of Company Common Stock ¹
Thomas Kinkade	10,000
Thomas Kinkade and Nanette Kinkade, as joint tenants	3,257,276
Kinkade Family Trust	1,000,000
Total	4,267,276

Prior to the Effective Time, it is anticipated that shares of Company Common Stock will be contributed by the other Affiliated Stockholders to a limited liability company to be formed by the other Affiliated Stockholders for the purpose of holding Company Common Stock to contribute to Mergerco immediately prior to the Effective Time.

A-28

APPENDIX B

[Jefferies & Company, Inc. Letterhead]

Corporate Finance

October 31, 2003

The Board of Directors Media Arts Group, Inc. 900 Lightpost Way Morgan Hill, California 95037

Members of the Board of Directors:

We understand that Media Arts Group, Inc., a Delaware corporation (the "Company"), and Main Street Acquisition Company, Inc., a Delaware corporation ("MergerCo") and Thomas Kinkade, an individual and sole stockholder of MergerCo, have entered into an Agreement and Plan of Merger, dated October 31, 2003 (the "Merger Agreement"), which provides for, among other things, the merger of MergerCo with and into the Company (the "Merger"). Pursuant to the Merger, we understand that each issued and outstanding share of common stock, par value \$0.01 per share, of the Company (the "Common Stock"), other than shares of Common Stock (i) owned by MergerCo or any of the Affiliated Stockholders (as that term is defined in the Merger Agreement), (ii) held in treasury or (iii) as to which dissenters' demands for appraisal rights have been properly exercised and perfected, will be converted into the right to receive \$4.00 in cash, without interest (the "Merger Consideration"). The terms and conditions of the Merger are more fully set forth in the Merger Agreement.

You have asked for our opinion as investment bankers as to whether the Merger Consideration to be received by the holders of shares of Common Stock pursuant to the Merger Agreement is fair from a financial point of view to such holders (other than MergerCo and the Affiliated Stockholders).

Jefferies & Company, Inc. ("Jefferies"), as part of its investment banking business, is regularly engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements, financial restructurings and other financial services. We are currently acting as financial advisor to the Company in connection with the transactions contemplated by the Merger Agreement and have received and will receive fees for our services, a portion of which will be payable upon delivery of this opinion. The payment of a significant portion of our fees is contingent on consummation of the transactions contemplated by the Merger Agreement. The Company has agreed to indemnify Jefferies against certain

B-1

The Board of Directors Media Arts Group, Inc. October 31, 2003 Page 2

liabilities, including liabilities arising out of or in connection with the services rendered and to be rendered by Jefferies under such engagement. We and our affiliates may own securities of the Company and/or its subsidiaries and affiliates and may maintain a market in the securities of the Company and/or its subsidiaries and affiliates and may publish research reports regarding such securities. In the ordinary course of our business, we and our affiliates may trade or hold such securities for our own account and for the accounts of our customers and, accordingly, may at any time hold long or short positions in those securities.

In conducting our analysis and arriving at the opinion expressed herein, we have, among other things, (i) reviewed the Merger Agreement (including all schedules and exhibits thereto); (ii) reviewed certain financial and other information about the Company that was publicly available; (iii) reviewed information furnished to us by the Company's management, including certain internal financial analyses, budgets, reports, projections, models and other information; (iv) held discussions with various members of senior management of the Company concerning historical and current operations, financial conditions and prospects, as well as the impact on the Company and its prospects of the economy and the Company's industry, including the effect of the current economic environment; (v) discussed and reviewed with various members of senior management of the Company and the Board of Directors a number of potential scenarios regarding the relationship between Thomas Kinkade and the Company, (vi) reviewed the valuations of publicly traded companies which we deemed comparable to the Company; (vii) reviewed the acquisition valuation multiples of publicly traded companies which we deemed comparable to the Company; (viii) reviewed the premiums paid in selected cash-only and cash-and-stock transactions; (ix) made inquiries regarding and discussed the Merger and Merger Agreement and other matters related thereto with the Company's legal counsel; and (x) reviewed the Common Stock trading price and volume history for a period which we deemed appropriate. In addition, we have conducted such other quantitative reviews, analyses and inquiries relating to the Company as we considered appropriate in rendering this opinion.

In our review and analysis and in rendering this opinion, we have, with your permission, relied upon, but have not assumed any responsibility to independently investigate or verify, the accuracy, completeness and fair presentation of all financial and other information that was provided to us by the Company or that was publicly available to us (including, without limitation, the information described above and the financial projections and financial models prepared by the Company regarding the estimated future performance of the Company), or that was otherwise reviewed by us. This opinion is expressly conditioned upon such information (whether written or oral) being complete, accurate and fair in all respects.

With respect to the financial projections and financial models provided to and examined by us, we note that projecting future results of any company is inherently subject to uncertainty. You have advised us that the Company in the past has not met or exceeded the

financial projections prepared by the Company's management. You have informed us, however, and we have assumed with your permission, that such projections and models were reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the

B-2

The Board of Directors Media Arts Group, Inc. October 31, 2003 Page 3

management of the Company as to the future performance of the Company. Although such projections and models did not form the principal basis for our opinion, but rather constituted one of many items that we employed, changes to such projections and models could affect the opinion rendered herein.

Accordingly, Jefferies' analyses must be considered as a whole. Considering any portion of such analyses or the factors considered, without considering all analyses and factors, could create a misleading or incomplete view of the process underlying the conclusions expressed herein. We expressly disclaim any undertaking or obligation to advise any person of any change in any fact or matter affecting our opinion of which we become aware after the date hereof.

In our review, we did not obtain any independent evaluation or appraisal of the assets or liabilities of, nor did we conduct a comprehensive physical inspection of any of the assets of the Company, nor have we been furnished with any such evaluations or appraisals for the Company or reports of such physical inspections for the Company, nor do we assume any responsibility to obtain any such evaluations, appraisals or inspections for the Company. Our opinion is based on economic, monetary, regulatory, market and other conditions existing and which can be evaluated as of the date hereof; however, such conditions are subject to rapid and unpredictable change and such changes could affect the conclusions expressed herein. We have made no independent investigation of any legal or accounting matters affecting the Company, and we have assumed the correctness of all legal and accounting advice given to the Company and the Board of Directors, including, without limitation, advice as to the legal, accounting and tax consequences of the terms of, and transactions contemplated by, the Merger Agreement, to the Company and its stockholders.

In rendering this opinion we have also assumed with your permission that: (i) the transactions contemplated by the Merger Agreement will be consummated on the terms described in the Merger Agreement without any waiver of any material terms or conditions; (ii) there is not now, and there will not as a result of the consummation of the transactions contemplated by the Merger Agreement be, any default, or event of default, under any indenture, credit agreement or other material agreement or instrument to which the Company or any of its subsidiaries or affiliates is a party; and (iii) all material assets and liabilities (contingent or otherwise, known or unknown) of the Company are as set forth in its consolidated financial statements provided to us by the Company.

Our advisory services and the opinion expressed herein are provided for the information of the Board of Directors of the Company in connection with its evaluation of the Merger. The Board of Directors requested that we solicit indications of interest from selected third parties regarding the possible acquisition of the Company, and we held preliminary discussions with certain of the parties prior to the date hereof. Our opinion does not address the relative merits of the transactions contemplated by the Merger Agreement as compared to any alternative transactions that might be available to the Company, nor does it address the underlying business decision by the Company to engage in, nor the process undertaken by the Board of Directors to evaluate and approve, the Merger or the terms of the Merger Agreement or the documents

B-3

The Board of Directors Media Arts Group, Inc. October 31, 2003 Page 4

referred to therein. Our opinion does not constitute a recommendation as to how any holder of shares of Common Stock should vote on any matter related to the Merger. Finally, we are not opining as to the market value or the prices at which any of the securities of the Company may trade at any time. Except as provided in our engagement letter with the Company, our opinion may not be used or referred to by the Company, or quoted or disclosed to any person in any matter, without our prior written consent.

Based upon and subject to the foregoing, we are of the opinion as investment bankers that, as of the date hereof, the Merger Consideration to be received by the holders of shares of Common Stock pursuant to the Merger Agreement is fair from a financial point of view to such holders (others than MergerCo and the Affiliated Stockholders).

Very Truly Yours,

/s/ JEFFERIES & COMPANY, INC.

B-4

APPENDIX C

DELAWARE GENERAL CORPORATION LAW SECTION 262 APPRAISAL RIGHTS

§ 262 Appraisal Rights

- (a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to § 228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder's shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word "stockholder" means a holder of record of stock in a stock corporation and also a member of record of a nonstock corporation; the words "stock" and "share" mean and include what is ordinarily meant by those words and also membership or membership interest of a member of a nonstock corporation; and the words "depository receipt" mean a receipt or other instrument issued by a depository representing an interest in one or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.
- (b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to § 251 (other than a merger effected pursuant to § 251(g) of this title), § 252, § 254, § 257, § 258, § 263 or § 264 of this title:
 - (1) Provided, however, that no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the agreement of merger or consolidation, were either (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in subsection (f) of § 251 of this title.
 - (2) Notwithstanding paragraph (1) of this subsection, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§ 251, 252, 254, 257, 258, 263 and 264 of this title to accept for such stock anything except:
 - a. Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;
 - b. Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or held of record by more than 2,000 holders;
 - c. Cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a. and b. of this paragraph; or

C-1

- d. Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a., b. and c. of this paragraph.
- (3) In the event all of the stock of a subsidiary Delaware corporation party to a merger effected under § 253 of this title is not owned by the parent corporation immediately prior to the merger, appraisal rights shall be available for the shares of the subsidiary Delaware corporation.
- (c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as is practicable.
 - (d) Appraisal rights shall be perfected as follows:
 - (1) If a proposed merger or consolidation for which appraisal rights are provided under this section is to be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for such meeting with respect to shares for which appraisal rights are available pursuant to subsection (b) or (c) hereof that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section. Each stockholder electing to demand the appraisal of such stockholder's shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder's shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or
 - (2) If the merger or consolidation was approved pursuant to § 228 or § 253 of this title, then either a constituent corporation before the effective date of the merger or consolidation, or the surviving or resulting corporation within ten days thereafter, shall notify each of the holders of any class or series of stock of such constituent corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder's shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send second notice to all such holders on or within 10 days after such effective date; provided, however, that if

C-2

such second notice is sent more than 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder's shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is

given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

- (e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) hereof and who is otherwise entitled to appraisal rights, may file a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after the effective date of the merger or consolidation, any stockholder shall have the right to withdraw such stockholder's demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder's written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) hereof, whichever is later.
- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by 1 or more publications at least 1 week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with such direction, the Court may dismiss the proceedings as to such stockholder.
- (h) After determining the stockholders entitled to an appraisal, the Court shall appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take

C-3

into account all relevant factors. In determining the fair rate of interest, the Court may consider all relevant factors, including the rate of interest which the surviving or resulting corporation would have had to pay to borrow money during the pendency of the proceeding. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceeding, the Court may, in its discretion, permit discovery or other pretrial proceedings and may proceed to trial upon the appraisal prior to the final determination of the stockholder entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder's certificates of stock to the Register in Chancery, if such is required, may participate fully in all proceedings until it is finally determined that such stockholder is not entitled to appraisal rights under this section.

- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Interest may be simple or compound, as the Court may direct. Payment shall be so made to each such stockholder, in the case of holders of uncertificated stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court's decree may be enforced as other decrees in the Court of Chancery may be enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.
- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney's fees and the fees and expenses of experts, to be charged pro-rata against the value of all the shares entitled to an appraisal.

- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder's demand for an appraisal and an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection (e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just.
- (l) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of authorized and unissued shares of the surviving or resulting corporation.

C-4

PROXY

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS MEDIA ARTS GROUP, INC. 900 LIGHTPOST WAY MORGAN HILL, CALIFORNIA 95037

SPECIAL MEETING OF STOCKHOLDERS

The undersigned hereby appoints Donald Potter and Robert C. Murray, and either of them, as proxies (each with full power of substitution) to vote, as indicated below and in their discretion upon such other matters, not known or determined at the time of solicitation of this proxy, as may properly come before the special meeting, all shares which the undersigned would be entitled to vote at the special meeting of the stockholders of Media Arts Group, Inc. to be held on , 2003 at :00 a.m., Pacific Standard time, at , and at any adjournment or postponement of the special meeting, as indicated below.

THIS PROXY WILL BE VOTED IN THE MANNER YOU DIRECT OR, IF NO CONTRARY DIRECTION IS INDICATED, WILL BE VOTED "FOR": (1) THE PROPOSAL TO APPROVE AND ADOPT THE AGREEMENT AND PLAN OF MERGER SET FORTH IN PROPOSAL ONE AND (2) AS THE PROXIES DEEM ADVISABLE ON SUCH OTHER MATTERS AS MAY PROPERLY COME BEFORE THE SPECIAL MEETING (OR ANY POSTPONEMENTS OR ADJOURNMENTS THEREOF).

(Continued and to be signed on the reverse side)

- Detach here from proxy voting card									
IF NO DIRECTION IS MADE, THIS PROXY WILL BE VOTED "FOR" ITEM 1 AND ITEM 2.						Please mar your votes indicated i this examp	as n		
The Board of Directors of Media Arts Group, Inc. recommends a vote FOR the proposals regarding:									
	FOR	AGAINST	ABSTAIN		FOR	AGAINST	ABSTAIN		
1. A proposal to approve and adopt the Agreement and Plan of Merger, dated October 31, 2003, by and among Media Arts Group, Inc., Main Street Acquisition Company, Inc. and Thomas Kinkade and approve the merger contemplated		o	o	2. In the discretion of the proxies, to vote upon such other matters as may properly come before the special meeting, including any postponements or adjournments of the special meeting.	0	O	o		
thereby.				This proxy is solicited on behalf of our board of directors. This proxy also delegates discretionary authority with respect to any other business which may properly come before the special meeting or which may properly come before any adjournment or postponement of the special meeting and matter incident to the conduct of the special meeting. No proxies marked AGAINST the proposal to adopt the merger agreement will be voted on a motion to adjourn or postpone the special meeting.					
				The undersigned hereby acknowledges receipt of the no meeting and proxy statement, each dated , 20			special		
				PLEASE SIGN AND DATE THIS PR	ROXY B	ELOW.			
Signature			Signature		Ε	D ate			
NOTE: Please sign exactly as your refull title.	ame appe	ars above. Whe	n signing as atte	orney, executor, administrator, guardian o	or corpor	ate official, ple	ase give		
		Dot	ash have from	nuovu voting and					
		- Deta	acii nere irom]	proxy voting card					

QuickLinks

TABLE OF CONTENTS

APPENDICES

SUMMARY TERM SHEET

QUESTIONS AND ANSWERS ABOUT THE MERGER AND THE SPECIAL MEETING

<u>CAUTIONARY STATEMENT CONCERNING FORWARD-LOOKING INFORMATION</u>

SPECIAL FACTORS

THE SPECIAL MEETING

THE MERGER AGREEMENT

INFORMATION RELATING TO MEDIA ARTS GROUP, INC.

SELECTED HISTORICAL FINANCIAL DATA

DIRECTORS AND EXECUTIVE OFFICERS OF MEDIA ARTS GROUP, INC.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

PURCHASES BY MEDIA ARTS GROUP AND ITS DIRECTORS AND EXECUTIVE OFFICERS

PRICE RANGE OF COMMON STOCK AND DIVIDENDS

OTHER MATTERS

FUTURE STOCKHOLDER PROPOSALS

INDEPENDENT ACCOUNTANTS

WHERE STOCKHOLDERS CAN FIND MORE INFORMATION

INFORMATION INCORPORATED BY REFERENCE

AGREEMENT AND PLAN OF MERGER BY AND AMONG MEDIA ARTS GROUP, INC., MAIN STREET ACQUISITION

COMPANY, INC. AND THOMAS KINKADE

TABLE OF CONTENTS

AGREEMENT AND PLAN OF MERGER

ARTICLE 1. THE MERGER

ARTICLE 2. THE SURVIVING CORPORATION

ARTICLE 3. EFFECT OF THE MERGER ON THE CAPITAL STOCK OF THE CONSTITUENT CORPORATIONS;

CANCELLATION AND CONVERSION OF SECURITIES

ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

ARTICLE 5. REPRESENTATIONS AND WARRANTIES OF MERGERCO AND THE PRINCIPAL AFFILIATED STOCKHOLDER

ARTICLE 6. COVENANTS

ARTICLE 7. CLOSING CONDITIONS

ARTICLE 8. TERMINATION AND ABANDONMENT

ARTICLE 9. GENERAL PROVISIONS

AFFILIATED STOCKHOLDERS

DELAWARE GENERAL CORPORATION LAW SECTION 262 APPRAISAL RIGHTS

PROXY